

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

CANCER CHEMOTHERAPY:

USE OF LOW DENSITY LIPOPROTEINS

AS TARGETING VEHICLES FOR TREATMENT

© SHARYN A. HYNDS B.Sc.

DEPARTMENT OF PATHOLOGICAL BIOCHEMISTRY

UNIVERSITY OF GLASGOW

THESIS SUBMITTED FOR THE DEGREE OF DOCTOR OF PHILOSOPHY IN THE FACULTY OF MEDICINE, UNIVERSITY OF GLASGOW, SCOTLAND

JULY 1986

ProQuest Number: 10991749

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10991749

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

CONTENTS

			PAGE
			NUMBER
	TITLE		1
	CONTE	NTS	2
	LIST	OF TABLES	7
	LIST	OF FIGURES	8
	DECLA	RATION	11
	ACKNO	WLEDGMENTS	12
	ABBRE	VIATIONS	13
	SUMMA	RY	14
	OVERV	IEW	15
	PLAN	OF WORK	16
SECTION	1 -	INTRODUCTION	18
·	1.1	Lipoproteins - Structure and Function	18
	1.1.1	Chylomicrons	18
	2	VLDL	18
	3	IDL	19
	4	LDL	19
	5	HDL	19
	6	Electrophoretic Properties	19
	7	Apolipoproteins	.19
	1.2	Lipoprotein Metabolism	20
	1.2.1	Exogenous Lipid Transport	20
	2	Endogenous Lipid Transport	22
	З	The LDL Receptor Pathway	23
	4	The LDL Receptor	24
	5	Specificity of LDL Receptors in vitro	26
	6	Hepatic Lipoprotein Receptors	27
	7	The LDL Receptor Pathway <u>in vivo</u>	28
	8	Sites of LDL Catabolism <u>in vivo</u>	30
	1.3	LDL and Cancer	32
	1.3.1	Hypocholesterolemia and cancer	32
	2	Plasma IDI Cholesteral	32

			PAGE
			NUMBER
	1.3.3	LDL Receptor Activity in Neoplastic	
		Cells <u>in vitro</u>	33
	1.4	Current Approaches and Problems in	
		Cancer Chemotherapy	35
	1.4.1	Pharmacokinetics	36
	2	Drug Administration	37
	3	Distribution, Metabolism and Excretion	38
	4	Cell Kinetics	39
	5	Drug Resistance	40
	6	Drug Targeting	41
;	7	Drug-Antibody Complexes	42
•	8	Drug-Macromolecule Complexes	42
	9	Microspheres	43
	10	Structural and Functional Advantages of LDL	
		as a Vehicle for Cytotoxic Drugs	44
	11	In vitro Studies With Drug-LDL Complexes	45
	12	LDL as a Cytotoxic Drug Vehicle <u>in vivo</u>	48
	1.5	Chlorambucil, Adriamycin and Daunomycin	51
	1.5.1	Chlorambucil	51
	2	Chemical Action of Chlorambucil	51
	З	Chlorambucil - Therapeutic Uses and Toxicity	53
	4	Resistance to Alkylating Agents	54
	5	Pharmacokinetics of Chlorambucil	54
	6	Anthracyclines	55
	7	Cytotoxic Action of Anthracyclines	55
	8	Metabolism of Anthracyclines	56
	9	Therapeutic Uses and Toxicity of Anthracycli	nes
	10	Resistance to Anthracyclines	57
	11	Pharmacokinetics of Anthracyclines	57

٠

З

		PAGE
		NUMBER
SECTION 2 -	MATERIALS, METHODS AND RESULTS	59
2.1	Materials and Methods I	59
	,	
2.1.1	Preparation of LDL	62
2	Preparation of HDL and	
	d>1.21 Plasma Fraction	62
З	Radioiodination of LDL	63
4	Synthesis and Iodination of Cholesteryl	
	Linoleoyl Ether	64
5	Incorporation of Labelled Cholesteryl	
	Linoleoyl Ether into LDL	64
	A) Preparation of cholate micelles	65 65
	C) Transfer of cholesteryl ether from	
	HDL to LDL	66 66
	D) Purification of labelled LDL	00
6	Transfer of Cholesteryl Ether From LDL to	66
	Other Lipoproteins	00
7	Cellobiose-tyramine-LDL : Synthesis	67
	and Labelling	67
8	Chemical Modification of LDL	69
9	Protein Determination	69
10	Sterilisation of samples for Injection	.70
11	Collection of Blood Samples	70
12	Induction of Tumours	71 71
	B) CFLP Mice/Parotid Adenomata	71
:	C) Sprague-Dawley Rats/Colonic Tumours	71 71
	D) Wistar Rats/Walker256 Tumours	71
13	Animals and Diets	72
14	Measurement of LDL Receptor Activity <u>in vivo</u>	/3
2.2	Results I	75
2.2.1	Measurement of Relative LDL Receptor Activity	
	in Animal Tumour Models Using Native and 1,2-	
	Cyclohexanedione-Modified LDL	75
	A) MAC-13 Tumour in NMRI Mice	75
	B) Parotid Adenomata in CFLP Mice	76 76
	D) Walker256 Tumours in Rats	77

.

	PAGE			
	NUMBER			
Measurement of LDL Catabolism Using				
Cumulative Non-degradable Markers	78			
A) Mac-13 Tumours B) Colonic Tumours C) Pancreatic Tumours	78 79 79			
Feeding Experiments	80			
 A) Cholesterol Feeding B) Bile Salt Supplementation C) Hydrocortisone Therapy D) Combination of Bile Salts and 	80 81 81			
Hydrocortisone	81			
Methods II	83			
Synthesis of Chlorambucil Cholesteryl				
Ester(CBE)	83			
Incorporation of CBE into LDL by Exchange				
From Cholate/Lipid Micelles	84			
A) "Reconstituted" HDL	84			
B) Exchange of CBE from Cholate Micelles Directly to LDL	85			
Degradation of CBE During Incubation with LDL	, 86			
Transfer of CBE from Synthetic Microemulsions				
to LDL	86			
Incorporation of Anthracyclines into LDL	87			

2.3	.1	Synthesis of Chlorambucil Cholesteryl	
		Ester(CBE)	83
	2	Incorporation of CBE into LDL by Exchange	
		From Cholate/Lipid Micelles	84
		A) "Reconstituted" HDL	84
		B) Exchange of CBE from Cholate Micelles Directly to LDL	85
	З	Degradation of CBE During Incubation with LDL	86
	4	Transfer of CBE from Synthetic Microemulsions	
		to LDL	86
	5	<pre>Incorporation of Anthracyclines into LDL A) Daunomycin B) Cholate Method C) DMSO Method D) PBS Method E) Dialysis F) Glycine Buffer G) Surfactants</pre>	87 87 88 88 88 88 88 88 88
	6	HPLC Assays A) Adriamycin B) Daunomycin C) Chlorambucil and CBE	89 90 91 91
	7	LDL Protein Determination in Solutions	
		Containing Anthracyclines	92
	8	Plasma Clearance of Radioactive LDL	93
	9	Cell Culture Media	93
•	10	Detachment of Cells from Culture Flasks	93
	11	Preparation of Lipoprotein-Deficient Serum	94
	12	Microtitre Drug Sensitivity Assay	94
	13	L-DAN Clonogenic Stem Cell Assay and	
		Spheroid Model System	96

2.2.2

З

2.3

		PAGE
		NUMBER
2.3.1	4 The Subrenal Capsular Assay	98
1	5 Measurement of Plasma White	
	Blood Cell Density	99
1	6 Histological Analysis of Tumour Sections	99
1	7 LDL Receptor Activity Assay <u>in vitro</u>	99
2.4	Results II	101
2.4.1	Chlorambucil Cholesteryl Ester-LDL	· 101
2	Comparison of CBE Concentration Measured by	
	Specific Activity and HPLC	102
З	Degradation of CBE During Incubation	
	Procedures	103
4	Transfer of CBE from Synthetic	
	Microemulsions to LDL	104
5	Degradation of CBE in LDL During Storage	105
6	Adriamycin-LDL	105
7	Daunomycin-LDL	107
8	In vitro Testing of Cytotoxicity of	
	Drug-LDL Complexes	107
9	In vivo Testing of Drug-LDL Complexes in	
	the SRC Assay	109
<u>SECTION 3</u> -	DISCUSSION AND CONCLUSIONS	111
3.1	LDL Receptor Activity in vivo	111
2	Evidence for LDL Uptake in Tumours Using	114
	Cumulative Markers of LDL Catabolism	
З	Drug Incorporation into LDL	116
4	Daunomycin and Adriamycin	118
5	Chlorambucil	120
6	In vitro Testing of Drug-LDL Complexes	123
7	In vivo Testing of Drug-LDL Complexes -	
	the Subrenal Capsular Assay	127
8	Conclusions	132
8	the Subrenal Capsular Assay Conclusions	127 132

SECTION 4 -REFERENCES

- 1. Physicochemical properties of human plasma lipoproteins
- 2. Human plasma apolipoproteins
- 3. Cultured cells which obtain cholesterol via the LDL pathway
- 4. Properties of the LDL receptor
- 5. Receptor-mediated uptake of LDL into the tissues of NMRI/MAC-13 mice
- 6. Effects of cholesterol feeding on plasma cholesterol concentrations in control and MAC-13 tumour-inoculated NMRI mice
- 7. Receptor-mediated LDL uptake into tissues of CFLP/viral adenoma mice
- 8. Receptor-mediated uptake of LDL into tissues of control and colonic tumour-bearing rats
 - 9. Receptor-mediated uptake of LDL into tissues of rats with Walker256 tumours
- 10. Receptor-mediated uptake of CE-LDL by NMRI mouse organs and MAC-13 tumour
- 11. Uptake of CT-LDL by rat organs and colonic tumours
- 12. Uptake of CT-LDL by hamster organs and pancreatic tumours
- 13. Effects of cholesterol feeding on receptor-mediated LDL assimilation by control and MAC-13-inoculated NMRI mice
- 14. Effects of bile acid feeding and hydrocortisone on receptormediated LDL uptake by tissues of control and MAC-13 tumourinoculated NMRI mice
- 15. Effects of bile acid feeding and hydrocortisone on receptormediated LDL uptake by rat tissues and colonic tumours
- 16. Apparent concentration of CBE in LDL at various stages of preparation by the Stoffel method
- 17. CBE content of LDL after exchange of drug from microemulsions containing DMPC or DPPC
- Recoveries of CBE and LDL protein at each stage of the preparation of CBE-LDL
- 19. CBE and LDL protein content of CBE-LDL complexes
- 20. Drug and protein content of adriamycin-LDL complexes prepared by a number of methods
- 21. Drug and protein content of daunomycin-LDL complexes
- 22. Cytotoxic effects of daunomycin-LDL on L-DAN cells in
 - a number of in vitro assays of cell growth inhibition

- 1. Cholesterol Metabolism
- 2. Schematic diagram of lipoprotein structure
- 3. Ultracentrifugal analysis of lipoproteins
- 4. Cholesterol transport in lipoproteins
- 5. The LDL receptor pathway
- 6. Plasma decays of native- and cyclohexanedione modified LDL in man
- 7. Diagram of the two-compartment model
- 8. Non-degradable markers of LDL catabolism
- 9. Dose response curve for cytotoxic drugs
- 10. Cell phase and cycle specificity of cytotoxic drugs
- 11. Chemical structure of some alkylating agents and anthracyclines
- 12. The mechanism of action of alkylating agents
- 13. The zonal preparation of LDL
- 14. Elution profile of human plasma following zonal preparation of LDL
- 15. Protocol for preparation of radioiodinated cholesteryl ether-LDL
- 16. Activity of plasma cholesteryl ester transfer protein in vitro
- 17. Chemical modification of LDL by 1,2-cyclohexanedione
- 18. Calculation of LDL receptor activity
- 19. Receptor-mediated organ uptake of LDL in NMRI/MAC-13 mice
- 20. Plasma decays of native- and CHD- LDL in control and tumourbearing NMRI mice
- 21. Receptor-mediated organ uptake of LDL in CFLP/adenoma mice
- 22. Receptor-mediated organ uptake of LDL in rats with colonic tumours
- 23. Receptor mediated organ uptake of LDL in rats with Walker256 tumour implants
- 24. Receptor-mediated uptake of cholesteryl ether-LDL by NMRI mouse tissues and MAC-13 tumour
- 25. Uptake of cellobiose tyramine LDL by rat tissues and colonic tumours
- 26. Uptake of cellobiose tyramine by hamster tissues and pancreatic tumours
- 27. Modulation of hepatic LDL receptors
- 28. Proton NMR spectrum of chlorambucil- -cholesteryl ester
- 29. Protocol for the modified Stoffel method of drug incorporation into LDL

- 30. Protocol for preparation of CBE-LDL using drug-rich microemulsions
- 31. Protocol for preparation of daunomycin-LDL
- 32. HPLC profiles of adriamycin and daunomycin
- 33. HPLC profiles of chlorambucil and chlorambucil cholesteryl ester
- 34. Adriamycin standard curve
- 35. Daunomycin standard curve
- 36. Chlorambucil cholesteryl ester standard curve
- 37. Interference of adriamycin in the Lowry protein assay
- 38. U.V./visible absorption of adriamycin in neutral and basicconditions
- 39. Protocols for HeLa and glioma microtitre drug sensitivity assays
- 40. Microtitre assay growth curves for Hela and glioma cells
- 41. Clonogenic assay protocol
- 42. Spheroid experiments protocol
- 43. Subrenal capsular assay protocol
- Scheme for assay of LDL degradation and internalisation of LDLby cells in monolayer culture
- 45. Proportions of CBE in fractions from centrifugal separation of CBE-HDL prepared using native- or "reconstituted"- lipoprotein
- 46. Evaluation of CBE-LDL, prepared by the Stoffel method, in the glioma microtitre assay
- 47. Evaluation of CBE-LDL in the glioma microtitre assay after replacement of tris buffer with phosphate buffer in the preparation procedure
- 48. Plasma decays of native-and CBE- LDL in rabbits
- 49. Degradation of CBE under various conditions
- 50. Degradation of CBE, encapsulated in LDL, during storage at 4 °C
- 51. HPLC profiles of adriamycin/LDL incubation mixtures at pH 7.4 and pH 10
- 52. Bound and free adriamycin in LDL solutions after dialysis of the lipoprotein against concentrated adriamycin solutions
- 53. Incorporation of adriamycin into LDL by dialysis
- 54. Glioma microtitre assay of daunomycin and daunomycin-LDL
- 55. Uptake and degradation of ¹²⁵I-LDL by L-DAN cells via highand low- affinity routes
- 56. Fluorescence micrograph of L-DAN spheroid after exposure to daunomycin-LDL .

LIST OF FIGURES

- 57. Fluorescence micrograph of L-DAN spheroid after exposure to daunomycin
- 58. HeLa microtitre assay of daunomycin and daunomycin-LDL
- 59. HeLa microtitre assay of adriamycin and adriamycin-LDL
- 60. Glioma microtitre assay of chlorambucil and chlorambucilester-LDL
- 61. Subrenal capsular assay of daunomycin and daunomycin-LDL on human primary gastric tumour
- 62. Haematoxylin and eosin stained section of mouse kidney showing lymphocyte infiltration into tumour implant
- 63. Subrenal capsular assay of chlorambucil and chlorambucil ester-LDL on Walker256 tumour
- 64. Haematoxylin and eosin stained sections of mouse kidneys showing areas of necrosis, fibrosis and lymphocytic infiltration
- 65. White blood cell counts in mouse plasma following exposure to drugs in the subrenal capsular assay of chlorambucil and chlorambucil ester-LDL against Walker256 tumour

. (

DECLARATION

The work presented in this thesis was performed solely by the author except for areas of acknowledged collaboration.

SHARYN A. HYNDS

This project was carried out at the Department of Biochemistry, Glasgow Royal Infirmary (G.R.I.) and I extend my thanks to Prof. H.G. Morgan for allowing me access to the excellent research facilities in his department.

I am indebted to my supervisors, Drs. J. Shepherd and C.J. Packard for their expertise and continued encouragement and direction throughout the experimental work and during the preparation of this thesis.

As a Member of an active research group, I have benefited greatly from the advice and practical help of the other members of the group, to whom I am most grateful.

My thanks go also to the staff of the animal unit, in particular to David F.S. McMurdo, who expertly cared for the animals and, when necessary, assisted during the experimental procedures.

During the course of this project I have collaborated with staff of the Department of Medical Oncology at Glasgow University, who were always willing to offer advice, and to them I extend my thanks. Drs. J. Welsh and K.H. Fearon provided MAC-13 and Walker256 tumours respectively. Dr. R.I. Freshney kindly provided the human glioma cell line and supervised the initial microtitre experiments. Dr. D.J. Kerr has collaborated in the testing of daunomycin-LDL complexes in vitro

I must also thank Dr. M.J. Stewart and the staff at Toxicology, G.R.I., for the loan of HPLC equipment and their advice in setting up the drug assays.

I am grateful to Drs. D. Cunningham and M. Soukop, Medical Oncology, G.R.I., the former for his cooperation and advice in assessing drug-LDL complexes in the SRC assay, and the latter for his interest and encouragement during the project.

Finally I thank my parents for their continued support and encouragement and Miss Yvonne H. Kane for typing this thesis.

The fo	ollov	wing abbreviations are used in this thesis :-
ACAT	:	Acyl:Cholesterol Acyl Transferase
ACR	:	Absolute Catabolic Rate
A-LDL	:	Adriamycin-LDL Complex
Apo B	:	Apolipoprotein B
BOP	:	N-nitrosobis(2-oxopropyl)amine
BSS	:	Balanced Salt Solution
CBE	:	Chlorambucil ¹ -<-cholesteryl Ester
CE	:	Cholesteryl linoleoyl ether
CHD	:	1,2-cyclohexanedione
СТ	:	Cellobiose-tyramine
d	:	density
D-LDL	:	Daunomycin-LDL Complex
DMH	:	Dimethylhydrazine
DMPC	:	Dimyristoyl phosphatidyl choline
DMSO	:	Dimethyl sulphoxide
DNA	:	6'-deoxyribonucleic acid
DNR	:	Daunorubicin, daunomycin
DOX	:	Doxorubicin, adriamycin
DPM	:	Disintegrations per minute
DPPC	:	Dipalmitoyl phosphatidyl choline:
EDTA	:	Ethylenediaminetetraacetate
FCR	:	Fractional catabolic rate
FH	:	Familial hypercholesterolemia
GI	:	Gastrointestinal
Glu	:	Glutamate
HDL	:	High Density Lipoprotein
HEPES	:	N-2-Hydroxyethylpiperazine-N'-2-ethanesulphonic acid
HMG Co	AF	Reductase:
		3-Hydroxy 3-methylglutaryl coenzyme A reductase
HPLC	:	High pressure liquid chromatography
ID	:	Dose of drug which inhibits cell growth by x%
TDI.	•	Intermediate density lipoprotein
TV	:	Intravenous
LCAT		Lecithin:Cholesterol Acyl Transferase
LO.IL	•	Lethal dose which kills x% of those who receive it
X	•	
LDL	:	Low density lipoprotein
MTX	:	Methotrexate
NaIOAc	::	Sodium iodoacetate
OD	:	Optical density
p	:	Probability
PBS	:	Phosphate buffered saline
R.E.	:	Reticuloendotnellal
^R f	:	Distance travelled by solute / distance travelled by solvent
RNA	:	Ribonucleic acid
rpm	:	Revolutions per minute
SDS	:	Sodium dodecyl sulphate
SRC	:	Subrenal capsular
SV 40	:	Simian virus 40
TLC	:	Thin layer chromatography
TCA	:	Trichloroacetic acid
Tris	:	Tris(hydroxymethyl)methylamine
WBC	:	White blood cell

The feasibility of using low density lipoprotein(LDL) as a selective drug-targeting vehicle is examined in this thesis. A number of animal tumours are seen to express considerable receptor-mediated uptake of the lipoprotein. The organs likely to suffer during chemotherapy using this carrier, viz. liver and adrenals, are afforded some protection by the administration of bile salts and hydrocortisone respectively. Several methods are employed for incorporating adriamycin, chlorambucil and daunomycin into LDL and the drug-LDL complexes so formed show enhanced cytotoxicity <u>in vitro</u> and <u>in vivo</u>. It is likely that LDL will be a useful targeting vehicle for anticancer agents if the level of drug incorporation can be increased. Despite recent advances in the design of cytotoxic agents, drug toxicity remains a major problem in the treatment of neoplastic disease. Consequently, there is much interest in methods by which the fraction of the drug dose reaching the tumour may be increased. One such method involves packaging the drugs into macromolecular vehicles, chosen for their ability to seek out tumour cells and deposit their contents only at those sites. In theory, this would markedly increase the efficacy of cytotoxic therapy while minimising any harmful side effects.

Over the years, numerous vehicles have been examined. Drugs have been attached to, for example, DNA¹, proteins ^{2,3}, hormones⁴ and monoclonal antibodies⁵ and they have been encapsulated in liposomes⁶ and liposome-antibody complexes⁷. <u>In vitro</u> experiments with liposomes were encouraging⁶. However, when drug-laden liposomes were administered <u>in vivo</u>, the cells of the reticulo-endothelial system rapidly removed the complexes from the circulation⁸⁻¹¹. The work reported in this thesis investigates the potential of a "natural liposome" - Low Density Lipoprotein (LDL) - as a vehicle for cytotoxic drugs.

In 1978 Ho et al observed 3-100 times the normal rate of degradation of low density lipoprotein in human leukemic cells <u>in vitro</u> and in 1981 Gal et al¹³ reported that a number of human gynaecological cancer cell lines grown in culture exhibited a higher uptake of LDL than did the equivalent non-neoplastic tissue. Other workers¹⁴⁻²³ have observed similar trends in several types of human and animal cancer cells. Recently Vitols et al²⁴ concluded that mononuclear cells in patients

with acute leukemia were capable of metabolising about 35% of the total plasma LDL-cholesterol pool per day.

All cells require cholesterol and they may obtain this in two ways: by endogenous synthesis or by assimilation of cholesterol-containing particles, lipoproteins, from their environment (fig. 1). The most important lipoprotein in this regard is LDL, the major cholesterolcarrying particle in human plasma²⁶. LDL consists of an apolar core of cholesteryl esters and triglycerides surrounded by a phospholipid monolayer containing free cholesterol and apoprotein B²⁷. Cells may take up this lipoprotein by two distinct pathways, one of which involves specific high-affinity receptors on cell membranes²⁵. Following binding to these receptors, located in coated pits²⁸ on the cell surface, an autoregulated sequence of events occurs which culminates in lysosomal degradation of the lipoprotein and subsequent release of the cholesterol for use in the cell²⁵.

The LDL particle possesses several characteristics which could be exploited if the particle is used as a drug vehicle:

- Neoplastic cells readily internalise and degrade LDL by the high-affinity receptor pathway;
- The LDL core has the capacity to store a substantial amount of drug;
- 3. Drug sequestration in the core space provides protection from serum enzymes and water; and
- The entry of drugs into cells via the LDL pathway may circumvent drug resistance mechanisms²³.



Until recently the feasibility of using LDL as a vehicle for cytotoxic drug <u>in vivo</u> had not been investigated to any great extent. There were no reports of LDL assimilation by solid tumours <u>in vivo</u>, although several groups have documented high LDL uptake by tumour cells <u>in vitro</u>¹²⁻²³. A brief outline of the plan of work for this project is given below:

- (a) The quantitative significance of the LDL-receptor pathway was assessed in several animal models and attempts were made to protect the organs likely to suffer during chemotherapy using LDL as a vehicle.
- (b) A number of cytotoxic drugs were chosen for loading into LDL. Several methods of incorporation were employed and the drugs were assayed by HPLC.
- (c) The efficacy of the drug-LDL complexes was tested <u>in vitro</u> using human cancer cell lines and in mouse models in vivo.

SECTION 1

INTRODUCTION

:

.

Cholesterol is an obligatory constituent of all mammalian cells. It is required for membrane synthesis and is also a precursor in the metabolic pathways of bile acid²⁹, steroid hormone^{30,31} and vitamin D. Since it is insoluble in aqueous solution, cholesterol is carried in complexes with other lipids and protein in plasma^{32,33}. These complexes are known as lipoproteins and are composed of apoproteins, cholesterol, cholesteryl esters, phospholipids and triglycerides. All lipoprotein particles share a common structure in which apolar neutral lipids are enveloped in a phospholipid membrane containing free cholesterol and apoprotein $B^{27,34}$. Figure 2 shows a schematic model of a lipoprotein. The lipoproteins are not a single species but a heterogeneous mixture of particles of variable lipid and protein composition^{27,34}. Several classes of lipoproteins may be identified by virtue of differences in size, density (fig.3) and other physical properties.

The five major classes of lipoproteins from human plasma, as defined by density, are:-

1.1.1. CHYLOMICRONS - d<0.95 Kg/L

These particles are high in triglyceride and are secreted into the lymphatic system by cells of the small intestine during the ingestion of dietary fat ^{33,35}.

1.1.2 <u>VERY LOW DENSITY LIPOPROTEINS (VLDL) d 0.95 - 1.006 Kg/L</u> The liver and small intestine secrete these lipoproteins into the plasma. Like chylomicrons, their main function is to transport triglyceride^{33,35}.





Figure 2: Schematic diagram of lipoprotein structure.

Figure 3:

ULTRACENTRIFUGAL ANALYSIS OF LIPOPROTEINS



1.1.3. INTERMEDIATE DENSITY LIPOPROTEINS (IDL) d 1.006 - 1.019 Kg/L IDL are produced in the plasma as intermediates in the metabolism of VLDL³⁶. They may be removed from the circulation by the liver or converted to LDL in the plasma^{33,36}.

1.1.4 LOW DENSITY LIPOPROTEINS (LDL) d 1.019 - 1.063 Kg/L

These are the major cholesterol-carrying lipoproteins of plasma. Twothirds of the circulating cholesterol is contained in LDL²⁶.

1.1.5 HIGH DENSITY LIPOPROTEINS (HDL) d 1.063 - 1.210 Kg/L

HDL are the smallest of the lipoprotein particles and their main lipid component is cholesteryl ester^{33,37}. They are involved in the transport of endogenous cholesterol back to the liver from extrahepatic sites.

1.1.6 ELECTROPHORETIC PROPERTIES

Lipoproteins may also be classified according to their electrophoretic mobility relative to the major plasma globulins³⁵. Thus chylomicrons remain at the origin, HDL migrate as the -1-globulin fraction and are termed -1-lipoproteins, VLDL, as -2, pre- β -lipoproteins, and LDL, β -lipoproteins.

The major properties and compositions of lipoproteins are summarised in table 1.

1.1.7 APOLIPOPROTEINS

Thirteen apolipoproteins have been isolated from human plasma³⁹ and the functions and primary sequences of many have been determined. They are commonly classified by the system of Alaupovic³⁹ into structurally and functionally related groups. Their importance in regulating

TABLE 1:	PHYSICOCHEMIC	AL PROPERTIES 0	15 HUMAN PLASN	IN LLPUPHOUTEINS			
	•	•				•	
PROPERTIES		CHYLOMI CRONS	VLDL	IDL	TDT	TOH	
DENSITY (Kg/)		4 0.95	0.95-1.006	1.006-1.019	1.019-1.063	1.063-1.21	
ELECTROPHORE'	FIC MOBILITY	Origin	pre- β	Б	ත.	Ŷ	
یں +		>400	20-400	12-20	0-12	I	
PARTICLE DIA	METER (nm)	120-1100	30-90	25–30	21–25	7-10	
COMPOSITION	(mean % dry w	eight)					
MAJOR LIPID	CONSTITUENTS						
TRIGLYCERIDE		88	58	30	თ	വ	
DHOSPHOLIPID		10	20	22	21	27	
ESTERIFIED C	HOLESTEROL	4	15	22	38	14	
FREE CHOLEST	EROL	N	Ø	ω	б	വ	
PROTEIN CONS'	TITUENT	CV.	ω	1.8	23	48	
MAJOR APOLIP	OPROTEINS	C,B,A	C,B,E	B,C,E	В	Α, С, Ε	
* Flotatio	on rate in Sved	10 ⁻	-13/2/32/2				

.

38,338 ì ;

.

"cm/s/dyne/g) in NaCl solution of density 1.063 g/mL at 26°C Flotation rate in Svedberg units (10⁻⁻

lipoprotein metabolism arises from their ability to interact with key enzymes and cell membrane receptors. The cardinal properties of apoproteins are shown in table 2.

1.2 LIPOPROTEIN METABOLISM

Two classes of lipids with distinct physical properties are carried by the lipoprotein transport system: the apolar lipids (cholesteryl esters and triglycerides); and the amphipathic lipids (cholesterol and phospholipids). These are organised into pseudomicellar structures in which the apolar lipids are sequestered from the bulk aqueous phase in the hydrophobic interior of lipoprotein particles (fig.2) Prior to their use in the cell the apolar lipids must be metabolised to yield free cholesterol and fatty acids⁴¹. The majority of triglycerides are removed by adipose tissue or muscle where their fatty acids are stored or oxidised to provide energy. Cholesteryl esters are transported to all cells of the body where they yield sterol for inclusion in membrane structures and as precursors of steroid hormones ^{30,31} and bile acids²⁹.

Cholesterol present in the circulation may originate from dietary intake or from endogenous synthesis in the liver and extrahepatic cells^{42,43}. The two "types" of cholesterol are initially transported in different pathways - the "exogenous" and "endogenous" routes.

1.2.1. EXOGENOUS LIPID TRANSPORT

Dietary lipids are absorbed by cells of the small intestine and secreted into the lymphatic system as large triglyceride-rich chylomicrons^{33,35}. The nascent chylomicrons possess apo B and

TABLE 2: <u>HUMAN PLASMA APOLIPOPROTEINS</u>⁴⁰

APOLIPOPROTEIN	DENSITY CLASS	MOLECULAR WEIGHT	PROPOSED FUNCTION
AI	HDL	28,000	Cofactor for LCAT
AII	HDL	17,000	Cofactor for Hepatic Lipase
AIV	CHYLOMICRONS	45,000	?
B-100	VLDL, IDL, LDL	550,000	Structural protein in VLDL, IDL & LDL. Receptor interaction in LDL pathway.
B-48	CHYLOMICRONS, REMNANTS	210,000	Structural protein of chylomicrons
CI	VLDL, HDL	6,500	?
CII	VLDL, HDL	10,000	Cofactor for Lipoprotein Lipase
CIII	VLDL, HDL	10,000	?
D	HDL ₃	20,000	? Cholesteryl ester exchange protein interaction
^E 2 ^{-E} 4	VLDL, LDL, HDL	35,000	Receptor interaction in chylomicron remnant pathway
F	HDL	30,000	?
G	VHDL	75,000	?
Н	CHYLOMICRONS & VLDL	50,000	?

apo AI and AII on their surface and obtain further apoproteins (apo C and apo E) from HDL upon entering the bloodstream. In the course of their metabolism 90% of the chylomicron apo C returns to HDL^{44} while apo B, an essential structural component of chylomicrons, remains with the particle throughout its lifetime in plasma⁴⁵. Chylomicrons are too large to pass through the endothelial barrier 46 and enter Their triglyceride is hydrolysed in the interstitial fluid. plasma compartment, yielding free fatty acids immediately adjacent to the cells which use them. This is accomplished by the action of lipoprotein lipase (E.C. 3.1.1.34) an enzyme which is attached to the luminal surface of capillary beds in muscle and adipose tissue. The lipase is activated by apo CII⁴⁷ on the chylomicron surface and the products of the reaction pass into adjacent adipocytes or muscle cells. Having lost much of its core, the chylomicron now has an excess of surface phospholipid which is shed and assimilated by HDL. The triglyceride-depleted particle, known as chylomicron remnant, is composed mainly of cholesteryl esters, with apo B and apo E as the major apoproteins. This lipoprotein is removed rapidly from the circulation by a receptor mechanism which recognises apo E on the particle surface 48,49. Following binding the remnants are subject to endocytosis and transferred to lysosomes where they are degraded. The average half-life of chylomicrons in the circulation is only a few minutes ^{50,51} and consequently, in normal individuals, they are abundant only in post-prandial plasma. In this way dietary cholesterol finds its way to the liver, the major organ involved in regulating cholesterol homeostasis. Dietary cholesterol in excess of immediate metabolic requirements is secreted in bile, both as free cholesterol and in the oxidised form of bile acids 29,32,50,51. Much of the former material is reabsorbed in the gut, transferred to chylomicrons and eventually returned to the liver, thus completing an enterohepatic

circulation, (fig. 4). Each time the cycle is completed a portion of the cholesterol is lost from the gut. Typically, humans acquire about 1 gram of cholesterol each day, 800 mg from endogenous synthesis and the remainder from dietary sources⁵¹. This is balanced by the faecal loss of neutral and acidic steroids.

1.2.2. ENDOGENOUS LIPID TRANSPORT

Fatty acids and carbohydrates are converted to triglyceride in the liver and this lipid is incorporated into lipoproteins prior to its secretion into the circulation. Similarly, cholesterol and cholesteryl esters are packaged in lipoproteins for delivery to extrahepatic cells. As noted above, some of the cholesterol is obtained from the diet and the remainder is produced endogenously^{27,52} from acetate under the regulation of 3-Hydroxy-3-Methyl-Glutaryl Coenzyme A Reductase (HMG CoA Reductase, E.C. 1.1.1.34) the rate-limiting enzyme in cholesterol biosynthesis⁵³.

VLDL are the major lipoproteins synthesised in the liver and when released into the bloodstream they, like chylomicrons, bind to lipoprotein lipase and, under its action, lose much of their tryglyceride. Apo CII is again required for activation of the enzyme⁴⁷ and is mainly acquired by transfer from plasma HDL⁴⁴. The apo B of VLDL remains an integral component of the particle throughout the delipidation cascade to IDL and LDL⁴⁵. The apo B of chylomicrons (termed apo B-48)⁵⁴ synthesised in the small intestine is known to be different from the VLDL apo B (apo B-100)⁵⁴ of hepatic origin. VLDL catabolism is much slower than that of chylomicrons, (half-life 1-3 hours³⁶) due possibly to their lower efficiency of binding to lipoprotein lipase⁵¹. Again excess surface phospholipid and cholesterol is transferred to HDL. The particles produced by the action of lipoprotein lipase are the triglyceride-deficient and much smaller intermediate density



.

Sec. 4

lipoproteins (IDL)^{36,55} which, like VLDL have Apo B as the major apoprotein, with trace amounts of apo C and apo E^{45} . At this stage, there is some removal of the IDL from the plasma by the liver 56. The remainder is metabolised to LDL, which are smaller particles, rich in cholesteryl ester, and containing apo B as the only apoprotein. The site of this second stage of conversion is uncertain but it is thought to occur in the liver. The acquisition of too much cholesterol by extrahepatic cells necessitates its export into the plasma. This is thought to be achieved by transfer of the sterol from the cell surface to HDL⁵⁷ and within the lipoprotein it is esterified by the action of Lecithin : Cholesterol Acyl Transferase (LCAT). The cholesteryl esters so formed are eventually transferred to VLDL, IDL and LDL via the agency of lipid transfer proteins 57,58. Several genetic disorders result in elevation of one or more of the lipoprotein fractions. Such defects contribute to about 20% of all myocardial infarctions occurring in those under the age of 60⁵⁹.

1.2.3. THE LDL RECEPTOR PATHWAY

In man, LDL carries more than two-thirds of the total plasma cholesterol²⁶. Plasma LDL has a half-life of the order of 2-3 days and delivers cholesterol to all cells of the body⁶⁰. Cells may obtain the cholesterol they require by endogenous synthesis of the sterol from acetate units or from LDL. When available, LDL is usually the preferred source. The lipoprotein is assimilated by high-affinity and low-affinity routes. On binding to specific high-affinity receptors in coated pits on the cell surface, LDL is rapidly taken into the cell by endocytosis⁶¹ (fig.5). The endocytic vesicle so formed migrates to the interior of the cell and fuses with lysosomes, exposing LDL to hydrolytic enzymes which release cholesterol, fatty acids, monoglycerides and amino acids. Cholesterol taken up in this way



Figure 5:

The LDL receptor pathway in cultured human fibroblasts²⁵. (ACAT - Acyl : Cholesterol Acyl Transferase HMG CoA Reductase - 3-Hydroxy 3-Methyl-

Glutaryl Coenzyme A Reductase)

regulates several intracellular events which collectively mediate cholesterol homeostasis within the cell, i.e. the activity of HMG CoA Reductase, the rate-limiting enzyme in cholesterol biosynthesis, is reduced $^{62,\,63}$ and Acyl : Cholesterol Acyl Transferase (ACAT) activity is increased⁶⁴ so that LDL cholesterol may be esterified for storage as cytoplasmic lipid droplets. A further consequence of receptor-dependent LDL uptake by the cell is the suppression of receptor synthesis, which prevents over-accumulation of cholesterol⁶⁵. Thus, by the use of these three mechanisms, the intracellular and extracellular sterol concentrations may be constantly balanced to maintain a stable cholesterol environment within the cell. About half of LDL cholesterol assimilation is accomplished via the receptor pathway in man. Low-affinity routes, e.g. bulk absorptive endocytosis, account for removal of the remainder of the LDL from plasma. One such route is thought to involve the reticulo-endothelial system^{66,67} and when plasma LDL-cholesterol levels are high, macrophages ingest LDL and can become "foam cells" which are characteristic components of atherosclerotic plaques⁶⁸. The exact proportion of cellular sterol which comes from receptormediated LDL uptake in each type of cell is uncertain but it is thought that de novo synthesis is sufficient for cells with a very slow rate of cholesterol turnover²⁵. However, cells which are heavy users of cholesterol, e.g. adrenals ^{30,31} and liver ^{29,50}, preferentially obtain cholesterol from the LDL receptor pathway.

1.2.4. THE LDL RECEPTOR

LDL receptors from bovine adrenal cortex were purified by Schneider et al 69 in 1982 and have been observed in a number of human and animal cells (table 3). The receptor molecule was found to be an acidic glycoprotein, with an apparent molecular weight of 164,000 on SDS

TABLE 3: CULTURED CELLS WHICH OBTAIN CHOLESTEROL VIA THE

LDL RECEPTOR PATHWAY*

HUMAN

Fibroblasts

Fibroblasts transformed by $SV40^{4}$ Smooth muscle cells Endothelial cells Lymphoblasts Burkitt lymphoma cells Acute myelogenous leukemia cells¹² HeLa cells Epidermoid cervical carcinoma(EC-50)¹³ Endometrial adenocarcinoma(AC-258)¹³ Glioma cells²⁰ Hep-G2 Hepatoma cells¹⁴ Non-small-cell lung carcinoma (L-DAN)#

Adrenal cells(Y-1 clone) Teratocarcinoma cells L cells L1210 leukemia cells P388 leukemia cells

Chinese hamster ovary cells

HAMSTER

DOG & SWINE

MOUSE & RAT

Fibroblasts Smooth muscle cells

Fibroblasts

COW

Adrenocortical cells Endothelial cells

From reference 28 unless otherwise stated
Simian Virus 40
This thesis, Section 2.4.8

polvacrylamide gel electrophoresis. The protein chain is composed of 839 amino acids and may be divided into 5 domains, each with a particular function 70 . The NH $_{
m 2}$ -terminal 322 amino acids constitute the first domain. This section contains many cysteine residues whose sulphydryl groups are bound in disulphide links, thus conferring rigidity and stability on this part of the receptor. Overall the first domain is negatively charged and probably provides the site for interaction with the positive arginine and lysine residues known to be crucial in the binding of apo B and apo E⁷¹. Second and third domains are 350 and 48 amino acids long respectively. The third domain is immediately adjacent to the cell and probably serves to extend the receptor so that binding sites are clear of the membrane. The fourth domain spans the plasma membrane. This region of the receptor has 22 amino acids, all of which are apolar. The carboxy terminal end of the receptor forms the fifth domain which projects into the cytoplasm. It is thought that this sequence binds to clathrin or a clathrinassociated protein, enabling the receptor to be incorporated into coated pits²⁸. The carbohydrate chains are attached, two through N-linkages to asparagine and 18 via O-linkages to serine and threonine 72. The exact location of these chains on the protein backbone is unknown. The LDL receptor is synthesised as a lower molecular weight precursor in the rough endoplasmic reticulum and converted to the mature receptor in the Golgi complex by addition and modification of the carbohydrate side chains⁷³. It is inserted into specialised areas of the plasma membrane, which allows it to function as an effective transport These areas are called "coated pits"⁷⁴. In 1976 Anderson⁷⁴, mediator. in collaboration with Goldstein and Brown, showed that, in cultured human fibroblasts, LDL receptors were found almost exclusively in these regions of the plasma membrane. The characteristic feature of these indented segments of the plasma membrane is a cytoplasmic surface lined
predominantly with the protein clathrin of molecular weight 180,000⁷⁵. Continually forming and pinching off, coated pits are transient in nature and their lifespan is less than five minutes⁷⁶. Their rate of formation and internalisation is not affected by the presence of LDL or LDL receptors²⁸ and each pit contains receptors for many different macromolecules²⁸.

Unlike LDL, the receptor molecule is not degraded on fusion with lysosomes within the cell. Instead, it returns to the cell surface and into the coated pits²⁸. This was demonstrated by blocking the synthesis of new receptors with cycloheximide and observing that there was no reduction in the rate of binding or internalisation of LDL by human fibroblasts over several hours⁶⁵. The coated pit, enclosing the LDL-receptor complex, forms an endocytic vesicle which breaks away from the plasma membrane. The endosome, containing the LDL, which has lost its clathrin coat and its complement of receptors²⁸ is carried through the plasma to fuse with a primary lysosome⁷⁴ where its contents are exposed to hydrolytic enzymes. Here apo B is enzymically degraded to its component amino acids and cholesteryl esters are hydrolysed to yield fatty acids and unesterified cholesterol²⁶.

1.2.5 SPECIFICITY OF LDL RECEPTORS IN VITRO

The receptor binds LDL by interaction with apo B on the particle's surface²⁵. This binding is dependent on a limited number of functionally significant arginine and lysine residues and it can be abolished by modification of these particular residues on the lipoprotein⁷⁹. Studies with human fibroblasts by Mahley et al^{78,79} showed that, in addition to apo B, the LDL receptor can bind apo E-containing lipoproteins, viz. chylomicrons and IDL. This property was discovered in experiments where dogs and swine were fed a high cholesterol diet which caused the appearance of an HDL species, HDL,

which is high in cholesteryl ester and contains apo E as the only apoprotein⁷⁸. ¹²⁵I-apo E-HDL_c, when placed in the culture medium of human fibroblasts, was taken up by LDL receptors with 10-25 times the affinity of LDL⁸⁰. It was also observed that the receptors became saturated by HDL at 25% of the number of LDL particles required for saturation. Pitas et al⁸⁰ deduced that the enhanced affinity of HDL_c was due to the fact that four apo E molecules in each HDL particle bound to a receptor on the cell surface.

1.2.6 HEPATIC LIPOPROTEIN RECEPTORS

The liver is known to remove chylomicron remnants, VLDL, IDL, and LDL from the plasma by receptor-mediated mechanisms⁸¹. Although as noted above, chylomicron remnants may be recognised by the LDL receptor most are removed from the plasma by interaction with a receptor which is distinct from the former. Apo E binds to this receptor and initiates endocytosis and degradation of the lipoprotein of which it is part. Cholesterol so obtained by the cell suppresses HMG CoA $Reductase^{82}$ as in the LDL receptor pathway. In dogs ¹²⁵I- apo E-HDL is rapidly cleared by the liver via the "chylomicron receptor pathway". Clearance is retarded by preventing receptor binding by modifying the lysine residues of apo E. Liver uptake of HDL is saturable and subject to competition from chylomicron remnants 48. The observations that LDL does not compete for chylomicron remnant uptake, and LDL receptor-deficient animals and humans have normal chylomicron metabolism, indicate that the two receptors are distinct. This is further illustrated in familial dys- β -hyperlipoproteinemia, where a form of apo E is synthesised which is not recognised by receptors, and remnants and IDL persist in the plasma, but LDL is metabolised as normal⁸⁴.

A summary of the properties of the LDL receptor is given in table 4.

TABLE 4: PROPERTIES OF THE LDL RECEPTOR[#]

LOCATION:	Concentrated in regions of cell surface membrane called clathrin coated pits
BINDING CAPACITY:	Saturation at 50µg of LDL protein per mL
BINDING AFFINITY:	K _m - 2μg of LDL protein/mL at 4°C ·10-15μg of LDL protein/mL at 37°C
BINDING SPECIFICITY:	Affinity for LDL more than 200-fold higher than for HDL. Affinity for canine HDL 23-fold higher than for LDL
NUMBER OF BINDING SITES PER CELL:	7,500-15,000 at 4°C, 1,500-70,000 at 37°C
MOLECULAR WEIGHT:	164,000 ⁶⁹
RATE AT WHICH BOUND LDL IS INTERNALISED INTO THE CELL:	$t_{\chi} - 3$ minutes at $37^{\circ}C$
RATE OF TURNOVER OF RECEPTORS:	$t_{\frac{1}{2}}$ - approx. 20 hours
BINDING CHARACTERISTICS:	Requires divalent cations Abolished by chemical modification of lysine and arginine residues of apoB & apoE ³⁷
BINDING pH OPTIMUM:	7.5
ENZYME TOLERANCE:	Sensitive to proteases. Resistant to glycosidases.
METABOLIC REGULATION:	Receptor synthesis repressed by LDL, cholesterol or oxygenated sterols.

Data from reference 25 except where otherwise indicated

1.2.7 THE LDL RECEPTOR PATHWAY IN VIVO

Results of <u>in vitro</u> experiments examining the regulation of LDL metabolism within the cell cannot be quantitively extrapolated to apply to the <u>in vivo</u> system. In order to calculate the uptakes of LDL by the receptor pathway in various cells <u>in vivo</u> it would be necessary to know the exact LDL concentration at the surfaces of extravascular cells and the extent of receptor expression in each cell type. Also, the LDL concentration in human interstitial fluid⁸⁵ is known to be high enough for suppression of cultured human fibroblast LDL receptors. Other methods have therefore been devised for the study of receptor mechanisms in vivo.

Since the lipids of LDL are constantly exchanging⁸⁶, their labelling yields data of limited value in following the kinetics of the intact LDL particle. Instead, the apoprotein moiety of the LDL is usually trace-labelled since it remains with the particle throughout its lifespan in the plasma, and the amount of apo B per particle is constant ^{39,87}. Thus apo B metabolism reflects the metabolism of the whole LDL particle. In typical experiments donors are injected with their own LDL which has been labelled with 125 I 88 . Blood samples are withdrawn at intervals and the clearance rate of LDL determined from serial measurements of plasma radioactivity. Iodotyrosine, generated in the degradation of LDL is rapidly filtered from the circulation by the kidneys. When results are expressed as percentages of initial radioactivity, a decay curve for LDL catabolism may be constructed (fig.6). This allows calculation of the fractional catabolic rate (FCR) of apo B^{88} . The decay curve for human ¹²⁵I-LDL was found to be biexponential⁸⁹ and this phenomenon may be explained with the use of a simple two-compartment model (fig.7), with intra- and



Figure 6: Plasma decays of Native- and 1,2-cyclohexanedione(CHD)-modified- LDL^r in man⁹³. (FCR : Fractional Catabolic Rate)



central compartment.

extravascular pools. The first compartment contains approximately two-thirds of the total LDL pool. This encompasses LDL in the intravascular pool and in rapidly exchanging tissue spaces. The remaining LDL is located in slowly exchanging extravascular spaces. After 5-6 days the two pools have equilibrated and consequently the decay curve becomes monoexponential⁸⁹. The FCR and apo-LDL concentration together are used to calculate the absolute catabolic rate (ACR), commonly expressed as mg Apo-LDL/day/Kg body weight. In the steady state the ACR is equivalent to the rate of synthesis of apo B^{89} . Although this method allows investigation of total LDL kinetics <u>in vivo</u>, it reveals no information about the extent of the contribution of the receptor pathway to LDL catabolism.

As mentioned previously, the lysine and arginine residues on the binding site of apo B may be chemically modified so that the binding of the apoprotein to its receptors is abolished ^{90,91}. Consequently. LDL modified in this way cannot gain entry to cells by the receptor pathway. It may enter only by non-receptor-dependent mechanisms and therefore modified lipoprotein is cleared much more slowly from the plasma⁹², (fig.6). The clearance of LDL by the receptor pathway may then be expressed as the difference between total LDL removed (native LDL) and the removal by receptor-independent methods Using this approach, it has been shown that the (modified LDL). LDL pathway makes a significant contribution to cholesterol metabolism in man^{93,94}, and several animal species^{91,92,95}. Reports of the proportion of LDL cleared by receptor vary according to the method employed to modify the receptor-recognition site of apo B. Values are in the range 30-60% of the total LDL catabolised. Studies conducted using 1,2-cyclohexanedione (CHD) - modified LDL appear to underestimate receptor activity 91,96,97. The partial reversal of CHD-modification has been shown to occur spontaneously at

pH 7.4 in vitro^{91,98}, but this seems not to affect significantly the results obtained in vivo⁹⁹. Reductive methylation of the lysine residues of apo B yields a stable modified LDL . This complex is. however, rapidly cleared from the plasma in man and dogs, with a half-life of only a few minutes¹⁰⁰ suggesting that it is recognised as a foreign particle. In some species (rats, rabbits, monkeys), however, methylation is successful in retarding the removal of LDL from plasma and gives a reasonable estimate of LDL clearance by non-receptor routes ^{91,92,101}. More recently, LDL whose lysine residues have been glycosylated has been used as a tracer. Estimates of receptor-LDL uptake by this method indicate that more than 50% of plasma LDL is metabolised via the receptor pathway⁹⁴. The above methods have also been used to assess the effects of genetic defects, drugs, and hormones on LDL catabolism 93,94. For example, cholestyramine¹⁰² and thyroxine¹⁰³ have thus been shown to stimulate the LDL receptor pathway in man and animals and thereby increase LDL catabolism.

1.2.8 SITES OF LDL CATABOLISM IN VIVO

Different types of cells have different requirements for cholesterol, therefore, the degree of expression of cellular LDL receptors is variable. Several methods have been developed to assess tissue receptor activity <u>in vivo</u>. Assays on isolated cell membranes³⁰ have shown that the receptors are present on most tissues (table 3), although the majority have relatively low activity⁶¹. Those tissues which actively utilise cholesterol, such as liver, adrenal glands and ovaries are all high in receptors⁶¹. Adrenals have a rapid throughput of cholesterol which they use as a precursor for steroid hormones^{30,31}. Similarly, ovaries require cholesterol as a substrate for hormone synthesis, while the liver utilises LDL cholesterol in the synthesis of bile acids and lipoproteins.

A second method of determining tissue receptor status is to use non-degradable markers of LDL catabolism (fig.8). The radioiodinated LDL commonly used in kinetic studies is degraded by the cell, releasing iodotyrosine which is rapidly excreted by the kidneys. Thus the tissue uptake values obtained using this tracer reflect only the catabolism occurring immediately prior to assessment. Non-degradable tracers have the advantage of remaining inside the cells they have entered and give a cumulative assessment of LDL catabolism over the time-course of the experiment. Pittman et al have used 14 C-sucrose 104 and ¹²⁵ I-cellobiose-tyramine¹⁰⁵ covalently bound to LDL as cumulative probes of the metabolic fate of the lipoprotein. Also, Stein et al 106 have studied the metabolism of LDL whose core has been loaded with non-degradable ¹²⁵I-cholesterol linoleyl ether. Regardless of the tracer used, such experiments have indicated that the adrenal glands are the most active tissues in LDL receptor-mediated uptake per gram weight but, due to its size, the liver makes the largest contribution to receptor catabolism of LDL, removing over 50% of plasma LDL by this 99,104,106,107



CELLOBIOSE TYRAMINE-LDL



CHOLESTERY L LINOLEOYL ETHER

Figure 8:

Non-degradable markers of LDL catabolism.

1.3.1. HYPOCHOLESTEROLEMIA IN CANCER

Hypocholesterolemia has been observed in many cases of cancer^{108,109}. In 1930 Muller reported hypocholesterolemia as a frequent finding in cases of leukemia¹¹⁰. Since then several epidemiological studies have been conducted¹¹¹. Of these 15 provided evidence that men in the lowest quintile of plasma cholesterol concentration were at increased risk of cancer and 12 showed no such relationship. Hence hypocholesterolemia, although common in cancer victims, cannot be used as a predictive factor. Until recently, there was considerable controversy as to whether the low plasma cholesterol observed in cancer patients is a cause of the neoplasm or an effect of its Bases et al¹¹² showed that plasma cholesterol levels presence. returned to normal in leukemic patients receiving successful chemotherapy, so supporting the latter argument. Further studies conducted in 1980¹¹³ and 1982¹¹⁴, indicated that most subjects with low plasma cholesterol who died of cancer, did so within 1-2 years of detection of the hypocholesterolemia and the concensus view now is that hypocholesterolemia in these patients was probably a consequence of cancer present, but undetected, at the time of blood sampling.

1.3.2. PLASMA LDL CHOLESTEROL

Ginsberg et al¹¹⁵ examined the cholesterol content of lipoprotein fractions in groups of hypocholesterolemic-leukemic patients and normal subjects. The lowering of plasma cholesterol in the former was mainly due to a decrease in LDL cholesterol. The plasma concentration and metabolism of VLDL were found to be normal but the catabolic rate of LDL was significantly increased. They postulated that the increased mass of actively proliferating neoplastic cells had elevated cholesterol requirements and so metabolised large quantities of LDL. This agrees with the findings of Gal et al¹⁶ that cancer cells in culture exhibit high LDL receptor uptake and endogenous cholesterol synthesis when in a logarithmic growth phase. Gilbert et al¹¹⁶ later showed that plasma cholesterol levels in a patient with chronic myelocytic leukemic fluctuated inversely with disease activity. This is further evidence that hypocholesterolemia is an effect of cancer rather than a cause.

1.3.3. LDL RECEPTOR ACTIVITY IN NEOPLASTIC CELLS IN VITRO

Ho et al in 1978¹² showed the rate of receptor-mediated degradation of LDL was 3-100 times higher than normal in mononuclear cells from patients with acute myelocytic leukemia, than in those from normal subjects. In these cells there was also a higher rate of endogenous synthesis of cholesterol. Despite the increased cholesterol availability in these cells, the cholesterol:protein ratio was 50% lower than normal. This implies a rapid turnover of cholesterol, which may be attributable to a rapid efflux of the sterol from the cells or, more probably, to a high rate of utilisation of cholesterol for cell growth and replication. The LDL receptor pathway in four human gynaecological cell lines was observed by Gal et al 16 to be subject to all the regulatory mechanisms observed in the pathway in human fibroblasts⁶⁵. Furthermore, the rate of receptor-mediated degradation of LDL was found to vary according to the cell density in the cultures, with cells in log phase exhibiting the highest rate of LDL degradation. These workers later reported fifty times the LDL receptor activity in cultured neoplastic gynaecological tissue as compared to the equivalent non-neoplastic tissue cultures¹³. The cancer cells also had elevated HMG CoA Reductase activity and overall the neoplastic cells were metabolising much larger quantities of

cholesterol than their non-neoplastic counterparts. When the same cell lines were propagated in mice, the tumours had 15-30 times more binding capacity for LDL than any of the vital organs. Tumour cells may therefore exhibit a much higher uptake of LDL than "normal" tissues and LDL, on the basis of these observations may therefore constitute a good vehicle for the delivery of cytotoxic drugs or radioisotopes selectively to cancer cells. The classic LDL receptor pathway has also been observed in human hepatoma cells^{14,117} and Leydig cells¹¹⁸ <u>in vitro</u>, and LDL has been shown to provide cholesterol for steroid hormone synthesis in the latter.

More recently, Vitols et al ¹¹⁹ have correlated leukocyte LDL receptor activity and plasma cholesterol levels in leukemic patients. Mononuclear cells were isolated from the blood of healthy donors and from patients with leukemia. Elevated receptor-mediated degradation of LDL was observed in cells from leukemic patients and there was an inverse relationship between plasma cholesterol and receptor activity. On analysis of the chromosomes of leukemic cells¹²⁰, those taken from patients with significantly higher LDL receptor activity were found to have an extra chromosome 8. On the basis of this finding Lindquist et al^{120} postulated that chromosome 8 may include genes of importance in the expression of the LDL receptor pathway. Further studies 24 revealed that leukemic cells in the bloodstream are capable of the catabolism of 35% of the plasma cholesterol pool per day. They conclude that the increased assimilation of LDL by leukemic cells may be the cause of hypocholesterolemia in acute leukemia.

The factors which influence drug efficacy in the treatment of diseases other than cancer apply equally in the management of cytotoxic therapy. As with any type of drug, absorption, binding, distribution, metabolism and excretion all affect the overall success or failure of treatment but, in addition, other properties of cytotoxic agents are important. For most anticancer agents the difference between an ineffective dose and a lethally toxic dose is very small, that is, these drugs have a very narrow therapeutic range¹²¹(fig.9). Many of the drugs commonly used are relatively unstable and are effective at very low concentrations. Some agents work only on cells at a particular stage of the cell cycle . Major organ dysfunction is not uncommon with cancer and may complicate chemotherapy since the absorption, excretion, etc. of the drug may be consequently modified, leading to increased toxic effects unless the dosage is changed accordingly. One way of improving cancer chemotherapy is to look for highly specific agents, but to-date there are no drugs which are completely specific for malignant cells. This is due to the lack of a qualitative difference between "normal" and "cancer" cells. For specificity, cytotoxic drugs rely on quantitative differences in distribution¹²³, the presence of target biochemical reactions within neoplastic cells¹²⁴, and differences in cell kinetics . In general, tumours possess a large proportion of actively dividing cells and it is these cells which are most sensitive to cytotoxic therapy. Other rapidly dividing cells in the body, including bone marrow and gastrointestinal tract epithelium 128 usually suffer the toxic effects of chemotherapy. A second method of obtaining better results with the anti-cancer compounds has been to take advantage of their dose response curve (fig.9) which has been shown in animal models 121 to be very steep. The number of tumour cells which

DOSE RESPONSE CURVE



Figure 9:

Dose response curve for cytotoxic drugs

survive may vary by several orders of magnitude between toxic and nontoxic doses. It is on this basis that high-dose chemotherapy is justifiable, despite the often inevitable myelosuppression and mucositis.

1.4.1 PHARMACOKINETICS

In order for any cytotoxic agent to be therapeutically effective, it must penetrate the tumour cell cytoplasmic membrane. Once inside, it must be present in sufficient concentrations and must be exposed to tumour for an adequate length of time. Physical and chemical properties of the drug determine tissue distribution and dictate its metabolism and excretion. These factors contribute to the concentration-time profile of the drug within body fluids, different tissues, cells and subcellular compartments.

The observed kinetics of many antineoplastic agents can be explained by a simple two compartment model (fig.7). This model assumes that the drug is distributed unevenly between a central compartment, of small volume, and a peripheral compartment of larger volume. Drug enters the central compartment and is rapidly distributed among the body fluids and tissues which it comprises. Although the two compartments are "theoretical spaces" and do not correspond directly to specific anatomical sites, the central compartment can be said to consist of serum and the extracellular fluid of well-vascularised tissues. Drugs in the serum and tissues within this compartment equilibrate rapidly. The peripheral compartment, in contrast, equilibrates very slowly with This compartment is thought to consist of the more poorlyserum. perfused tissues e.g. muscle, skin, fat and some tumours. Drug may enter or leave the system only by the central compartment. The apparent volume of drug distribution for each compartment is dependent on the extent of blood flow to each tissue, the ability of

the drug to penetrate the tissues and the drug's affinity for them. The peripheral compartment serves as a reservoir for the central compartment. As drug is excreted from the central compartment, it is replaced by drug from the peripheral compartment.

1.4.2. DRUG ADMINISTRATION

· · · · · · · ·

There are several routes by which antineoplastic agents may be administered. The method by which a particular drug is given depends on its stability, solubility and sclerosant characteristics. Some drugs are unsuitable for oral administration because they are destroyed by enzymes or low pH in the gastrointestinal tract, or precipitate at acidic pH. Others cannot be given intravenously because they are insoluble in aqueous solvents.

Oral administration is convenient and suitable for many drugs, e.g. chlorambucil. However, gastrointestinal (GI) function is frequently abnormal in cancer patients as a result of pre-existing GI complaints, the presence of abdominal tumour or side effects of treatment. Even in the absence of GI complications there is great variability in absorption of drugs between patients.

Intramuscular, intraperitoneal, intrathecal and subcutaneous routes are less commonly used. Anthracyclines, for instance, cannot be given by these routes since the high local concentration of drug generated may cause bleeding and tissue necrosis around the injection site¹²⁹. Intrathecal injection is useful in cases where drug would not otherwise penetrate the blood-brain barrier in sufficient concentrations. Intravenous drug administration is the most common of parenteral methods. Drug is injected directly into the central compartment and a stable high drug concentration may be maintained over a defined time by continuous infusion. Drawbacks to this approach are the possibility of thrombosis or infection and drug extravasation may cause necrosis at the

injection site. Intra-arterial injection may be given to induce a high local drug concentration at the tumour. However, this requires surgery and a sizeable tumour artery must be located. This procedure, like IV administration, carries the risks of infection and thrombosis. Also, the administration of drugs like cyclophosphamide by this method is of no added benefit, since the drug requires enzymic activation in liver microsomes^{130,131}.

1.4.3. DISTRIBUTION, METABOLISM AND EXCRETION

The distribution of cytotoxic drugs in different tissues of the body depends on the chemical structure and physical properties of the drug in question. Polar drugs have difficulty in crossing lipid membranes and hence are largely excluded from the central nervous system. The concentration of drug in a particular organ is dependent on the size of the organ, the rate of blood flow through it, and the lipid solubility and polarity of the drug. Certain sites are relatively impermeable to polar drugs, e.g. central nervous system, testicles, large tumours. There may be other pharmacological "sanctuary sites" in which the penetration of particular drugs is limited. Response rates to chemotherapy in metastic breast cancer are higher in some metastases than others¹³². This may be a result of biochemical and kinetic differences between sites, but it is likely that drug distribution is important. A "third space" has been described where drugs may equilibrate very slowly with the central compartment. This third compartment probably includes pleural or peritoneal effusions. The rate of drug elimination from patients with these "extra" reservoirs may be so greatly prolonged as to produce increased toxic side effects. Many cytotoxic drugs are extensively metabolised. Some require enzymic activation, e.g. cyclophosphamide becomes a potent alkylating agent only after metabolism by microsomal enzymes in the liver 130,131 .

Nucleic acid antimetabolites are activated in several sites in the body, including tumour cells¹³³. It is important that the sites of activation function properly. Organ dysfunction^{123,124} may result in a prolonged half-life of the ineffective form of drug, with a failure to attain sufficiently high levels of the metabolised active agent. Conversely, in the case of drugs destroyed on metabolism or inefficiently excreted, toxic levels of drug may persist in the body for dangerously long periods.

Some drugs are excreted as the active drug, therefore impairment of the excretion of these drugs leads to increased toxicity and doses must be reduced accordingly. Anthracyclines are excreted in bile¹³⁵ and consequently any blockage of the biliary tract, e.g. gallstones, reduces or prevents excretion and anthracycline concentrations may rise above the safe therapeutic limit. Similarly, drugs excreted via the kidney are more likely to reach toxic levels and may even produce nephrotoxicity if administered to patients with kidney dysfunction¹³⁶.

1.4.4. CELL KINETICS

Growing tumours may be composed of both proliferating and nonproliferating cells, and within the former subpopulation, the cells may be at various phases of the cell cycle¹²² (fig.10.) Thus in the optimisation of drug schedules account must be taken of the kinetics of tumour growth, since certain agents may be markedly more effective at particular stages of the cell cycle. As an alternative to classification by their mode of action, anticancer drugs are commonly classified according to the stage of the cell cycle during which they are most toxic. Thus three classes of cytoxic agent are defined:-

- 1) cell cycle non-specific
- 2) cell cycle and phase-specific
- 3) cell cycle non-phase-specific



Figure 10:

Cell phase and cycle specificity of cytotoxic drugs

(Adapted from "Basic Principles of Cancer Chemotherapy", Calman, K.C. et al 1980 Pub. McMillan, LONDON.1 The first of these classes includes Y-irradiation and nitrogen mustard, and these agents are toxic to both dividing and non-dividing cells. Methotrexate, vinblastine and cytosine-arabinoside belong to the second category. Drugs of this type are ineffective against cells which are not actively proliferating. The third type of drug e.g. cyclophosphamide, nitrosoureas, actinomycin D, is much more toxic to proliferating than non-proliferating cells. Since most tumours are composed of cells in various stages of the cycle, drugs from each of the above categories are commonly given in combination in order to maximise the proportion of tumour cells under attack during treatment. Consequently combination chemotherapy is usually more effective than single-agent therapy.

1.4.5. DRUG RESISTANCE

Drug resistance is often a complicating factor in the treatment of The resistance may occur in normal tissues as well as cancer. tumour and may be intrinsic (e.g. bleomycin and cytosine arabinoside are not effective against bone marrow and bowel cancer respectively) or acquired. Resistance may be conferred by a combination of pharmacological, biochemical and kinetic factors¹³⁷. One beneficial aspect of acquired resistance is that normal cells may develop an increased tolerance of chemotherapy with repeated drug treatments. Acquired resistance¹³⁸ is often due to induction of drug metabolism¹³³ and changed kinetics, in both normal tissues and tumour. Another mechanism of resistance in tumours is believed to be a result of mutation, whether induced by drugs or irradiation therapy. In general, normal tissues are less susceptible to mutation since they contain a relatively smaller proportion of dividing cells than tumours. The frequency of mutation increases with tumour size and is variable between different tumour types.

1.4.6 DRUG TARGETING

Thus there are a number of problems which beset the clinical oncologist in his efforts to rid the patient of cancer cells and retain normal ones. There are two obvious ways of improving the benefit/risk ratio of cytotoxic drug therapy:-

Improved Drug Design - drugs or prodrugs should be designed so that (1)they have a high affinity for the site at which their action is required, or are only converted to their active form at that site. This should markedly increase the efficacy of therapy while minimising any harmful side effects. However, this approach is limited by the strength of the target/agent relationship which is generally far from ideal. The Use of Targeting Vehicles - As long ago as 1906, Ehrlich¹³⁹ (2)proposed that molecules with an affinity for certain tissues could be used as carriers for therapeutic agents. By encapsulating drug within a vehicle, the distribution, uptake and overall efficacy of the drug may be beneficially altered so that tumour cells are damaged substantially more than normal cells. The ideal vehicle would deliver drugs only to their site of action and, once there, they would be passively released or internalised upon recognition of their target.

In addition to drug selectivity, numerous other factors, e.g. distribution, kinetics may act to reduce the intrinsic activity of a highly selective drug. Thus it is preferable to target drugs within some sort of carrier system. Drug vehicles should be non-toxic, biodegradable and of size and shape to allow the incorporation of many drug molecules, without evoking a response from the host's immune system. They should have a high affinity for their target site, while ignoring irrelevant areas. There must be a mechanism by which the drugs enter the cells along with, or after release from, their carrier. Several types of drug carrier have been investigated. These fall into three main categories: drug-antibody complexes; drug-macromolecule complexes; and drug microspheres.

1.4.7. DRUG-ANTIBODY COMPLEXES

Drug-antibody complexes often have improved toxicity against malignant cells^{5,140,141,142}. Binding of the immunoglobulin to its antigen on the cell surface is thought to initiate endocytosis¹⁴¹, followed by lysosomal degradation of the antibody-drug complex and release of the free drug inside the cell. Isolation and purification of target-specific antibodies, however, remains difficult and it is also possible that the antigenic specificity of the tumour may change during its progression. Furthermore, drugs may distort the antigen-binding site, and render the complex susceptible to removal by the reticulo-endothelial system. The drug-antibody interaction may be subject to competition from endogenous circulating antibodies against target cells and the host may even develop an allergic reaction to the drug-antibody complex.

1.4.8. DRUG-MACROMOLECULE COMPLEXES

Phagocytes and some malignant cells are able to actively endocytose macromolecules. Among the many macromolecular carriers which have been studied are DNA, albumin, liposomes and LDL. Deprez-de Campeneere and co-workers¹ have studied daunorubicin (DNR) and doxorubicin (DOX) linked covalently to DNA. In mice with leukemia, both complexes were more effective than the native drugs and the overall toxicity of DOX-DNA was significantly less than that of doxorubicin. Cornu et al¹⁴³ have shown drug-DNA complexes to be effective on certain tumours in man. Proteins such as albumin have also been used as carriers. β -amantin coupled to albumin¹⁴⁴ was selective in damaging kidney proximal tubules and liver sinusoidal cells <u>in vivo</u>. Daunorubicin covalently bonded to albumin³ was found to be more effective than the

free drug, both <u>in vitro</u> and <u>in viv</u>o. The major drawback of drugprotein carriers is that liver and kidneys will suffer during chemotherapy since they constitute the main sites of protein endocytosis.

A great deal of work has been done with drug-liposome complexes, where the drugs are carried entrapped in lipid vesicles⁶. Although liposomes provide a protective environment for the drug, they are leaky and small drug molecules may easily diffuse out into the plasma. This has been overcome to some extent by incorporating drugmacromolecule complexes e.g. DNA, poly-L-Glu¹⁴⁵ into the liposomes. In early work, the size of the lipsomes used (µm) precluded their entry to small capillaries and induced their rapid clearance from the plasma by cells of the R.E. system⁸⁻¹¹. Recently, much smaller, more stable particles have been synthesised, but the liver, spleen and lungs remain the major targets for liposomes in vivo⁸⁻¹¹. Inclusion of monoclonal antibodies in the liposome surface has increased the specificity of the carrier and, consequently, the efficacy of the drug¹⁴⁶. In the case of methotrexate in liposomes conjugated with antibody to L929 murine fibroblasts⁷, a 10-fold increase in drug efficacy in vitro was observed. Another promising macromolecular drug carrier is LDL. LDL is, in effect, a "natural liposome" and as such has many of the advantages of liposomes, with the added benefit that its removal from the circulation by the R.E. system is likely to present less of a problem. The potential of LDL as a drug vehicle is discussed fully later (1.4.10).

1.4.9. MICROSPHERES

These are relatively large (0.2-300 μ m) microspherical colloids of polymers such as albumin¹⁴⁷. They are employed in the controlled localised release of drugs. Their target is dictated largely by their

size. Very small microspheres are cleared from the circulation by macrophages of the R.E. system and are ultimately delivered to Kuppfer cells in the liver. Particles 7-12 µm become trapped in the lungs. Liver and spleen filter out particles 2-12 µm. Particles larger than 12 µm stick in the first capillary bed they encounter. Thus large microspheres will become entrapped in tumour vasculature if they are injected into the arterial supply of the tumour¹⁴⁸. A high local concentration of drug is therefore achieved with reduced effects on non-target tissues. Microspheres 1-2 µm may be coated with magnetic substances¹⁴⁹ and directed to the required sites by external application of two-dimensional magnetic fields. Senyei et al¹⁵⁰ demonstrated that magnetic carriers of doxorubicin can achieve high levels of drug in the tumour with no drug in any other organ.

1.4.10 STRUCTURAL AND FUNCTIONAL ADVANTAGES OF LDL AS A VEHICLE FOR CYTOTOXIC DRUGS

Low density lipoproteins are particles composed of an apolar core of cholesteryl esters and triglycerides surrounded by a phospholipid monolayer containing free cholesterol and apoprotein $B^{27,34}$. The proportions of each of the lipids is shown in table 1. The apolar lipid core contains approximately 1500 cholesteryl ester molecules and provides an excellent site for sequestration of lipophilic molecules. Since the studies of Rudman et al¹⁵¹, it has been observed that numerous lipophilic agents partition into the lipoprotein fractions in vivo. Gal et al¹³ indicated that LDL might prove to be a suitable vehicle for targeting of cytotoxic agents. Membrane fractions from human tumour cells (EC50), after propagation in mice, were found to have 15-30 times the affinity for LDL expressed by vital organs of the mouse. Studies in our own laboratory¹⁵² and by other workers^{153,154} indicate too that mouse tumours <u>in vivo</u> have a high affinity for LDL,

which they obtain by the high-affinity receptor pathway. Cell experiments have shown that the LDL pathway in neoplastic cells^{13,14} has all the characteristics of the classic LDL receptor pathway described by Goldstein and Brown in cultured human fibroblasts⁶¹. On binding to its receptors on the cell membrane, LDL is rapidly endocytosed and degraded in the lysosomes liberating its charge to the inside of the cell (fig.5). LDL has the potential to act like the "ideal" drug vehicle in many respects¹⁵⁵, providing adequate amounts of drug for therapeutic activity can be incorporated into the lipoprotein.

1.4.11 IN VITRO STUDIES WITH DRUG-LDL COMPLEXES

Recently numerous groups have reported successful arrest of fibroblast or tumour cell growth <u>in vitro</u> using antineoplastic agents carried in LDL.

Using "reconstituted" LDL¹⁵⁶, Mosley et al¹⁸ delivered the photosensitising agent, pyrene, via the LDL receptor pathway. Subsequent exposure to U.V. light led to cell death, since pyrene activated in this way produces oxygen radicals. Human A431 epidermal carcinoma cells, SV40-transformed human fibroblasts and chinese hamster ovary cells were among those killed using this method. Fibroblasts from a patient with familial hypercholesterolemia (FH) which had no LDL receptors, were unable to take up LDL by the receptor pathway. Consequently, pyrene accumulation in these cells was very low and they were not killed after exposure to U.V. light. In later experiments¹⁹, chinese hamster ovary cells pre-incubated with pyrene were exposed to LDL containing photoprotective azo dyes, scavengers of singlet oxygen. After U.V. irradiation, cells exhibiting high LDL receptor activity were undamaged by the pyrene and it was established that the photoprotective dyes were delivered to the cells by a receptor-dependent process.

In 1983, Rudling et al²⁰ showed that cellular accumulation of LDL containing the anthracycline Aclacinomycin A was dependent on LDL receptor activity in cultured human glioma cells. LDL accumulation was found to be highest when the cells were proliferating rapidly. The receptor-dependent uptake of drug-LDL was subject to competition from native LDL but not HDL. Chloroquine, an inhibitor of LDL degradation caused increased LDL receptor uptake. The drug-LDL complex inhibited growth in glioma cells but not in FH fibroblasts.

Firestone et al¹⁵ synthesised a number of lipophilic derivatives of cytotoxic agents with a view to reconstituting them in LDL. Two of these derivatives, both steroid nitrogen mustard carbamates, reconstituted well with LDL and were able to arrest the growth of chinese hamster ovary cells and SV40-transformed human fibroblasts at low drug concentrations. Using one of these compounds, $(N-[[2-[3\beta-(oleoyoxy)-(oleoyox)-(oleoyoxy)-(oleoyox)-(oleoxy)-(oleox)-(oleoxy)-(oleo$ and rost-5-en-17 β -yl] propyl] oxy] carbonyl]-N,N-bis (2-chloroethyl)amine), the growth of 98% of the cells was arrested at 10 $\mu g/mL$ of the However, only 50% of these cells were actually killed. drug in LDL. Increasing the drug concentration gave no enhancement in cytotoxicity. There were two possible reasons for this: 1) saturation of the LDL pathway; and 2) inefficient hydrolysis of the drug-steroid bond in the A second derivative (N-[[[4-[3β -(oleoyloxy)androst-5-enlysosomes. 17β -yl <u>pentyl</u> oxy carbonyl -N, N-bis-(2-chloroethyl)amine) was synthesised, in which the steric congestion around the drug-steroid link was reduced, so facilitating the rapid enzymic liberation of free drug within the cell. This derivative, when reconstituted with LDL was able to kill 100% of the cells at concentrations of 5 $\mu g/mL$ and above.

Lipophilic derivatives of adriamycin were used by Vitols et al^{21,22} for reconstition in LDL. In cultured human fibroblasts, accumulation of (N-(N-retinoyl)-L-leucyl doxorubicin-14-linoleate) -LDL²¹ was measured

and found to be equivalent to the theoretical uptake implied by the observed rates of LDL assimilation and degradation. As in the aforementioned studies, FH fibroblasts showed little or no uptake of the drug-LDL complex. Drug-LDL complexes whose receptor-binding ability had been abolished by methylation were not assimilated by the cells. The drug-LDL complex was found to inhibit growth of normal fibroblasts. In human white blood cells from normal and acute myelocytic leukemia patients, the uptake of AD-32-LDL²² was many times higher than the amount which could have been assimilated by the receptor pathway alone. Therefore, in this system, the authors concluded that non-specific mechanisms were responsible for the accumulation of a large proportion of the drug. This was analgous to the results of Remsem and Shireman 157 who observed that the carcinogen benzo(a)pyrene, when carried within LDL, was able to enter cultured fibroblasts in significant amounts even in the absence of LDL receptors, apparently via a rapid distribution of the compound between lipoprotein and cell membrane.

Daunomycin was incorporated into LDL by Iwanik et al²³ using a simple incubation procedure. After fluorescence quenching with DNA and KI, they concluded that the daunomycin had partitioned into two domains of the LDL particle viz. the core and the surface. An interesting finding was that the complex was toxic to both daunomycin-sensitive and daunomycin-resistant P388 leukemic cell sublines. In both cases more drug entered the cells when daunomycin-LDL was administered than when free daunomycin was used. Subfractionation of subcellular organelles showed the drug to be located mostly (86%) in insoluble nuclear material when native daunomycin was used. After exposure to the drug-LDL complex, some of the drug was found in the plasma membrane and nuclear membrane fractions, but the majority (65%) was in the mitochondrial-microsomal-lysosomal membrane fraction and insoluble

nuclear pellet. The daunomycin distribution in organelles correlated well with the distribution of 125 I-labelled native- and daunomycin-LDL. In subsequent work 158 , the same group showed the LDL-drug complex to be more cytotoxic than the native drug at shorter exposure times. Uptake of daunomycin-LDL was subject to competition from LDL but not HDL. On analysis of the DNA content of treated cells it was apparent that cells exposed to daunomycin-LDL had accumulated in the G_2 -M phase of the cell cycle, whereas those given daunomycin were evenly distributed in each phase, thus illustrating the enhanced cytotoxic effects of the daunomycin-LDL delivery system.

Halbert et al¹⁷ linked methotrexate (MTX) covalently to the apo B moiety of LDL. The complex was thirty times less active against murine leukemia <u>in vitro</u> than native drug. It is likely that the drug binding to apo B caused distortion of the protein molecule, so preventing its recognition by the LDL receptor, although it has been reported¹⁵⁹ that other covalent methotrexate-macromolecule complexes are less active than MTX when tested under similar conditions. Another possible explanation for this decreased cytotoxicity is that methotrexate is not being efficiently hydrolysed from LDL and insufficient concentrations of free MTX are being released inside the cell.

1.4.12 LDL AS A CYTOTOXIC DRUG VEHICLE IN VIVO

From the results of the many cell experiments with drug-LDL complexes, it is apparent that LDL has potential as a drug vehicle. It carries drug in appreciable quantities primarily to cells which express specific LDL receptors. On binding to the receptors, the drug, inside LDL, is internalised by endocytosis and transferred to the lysosomes. Here the complex is degraded releasing the drug, free to apply its cytotoxic action within the cell. LDL is an endogenous particle and as

a carrier, it is not susceptible to the possible resistance mechanisms and allergic responses that some other macromolecular drug vehicles evoke.

For LDL to act as a drug-vehicle in vivo it is crucial that:

- the structure of the particle is not changed in such a way that it is rendered "foreign";
- the conformation of apo B is maintained to allow efficient binding to LDL receptors.

Some of the above studies^{15,18,19,21,22,157} used LDL which had been freeze-dried and reconstituted with drug by Kreiger's method¹⁶⁰. This procedure was originally developed for replacing cholesteryl esters in the core of LDL with their labelled counterparts¹⁵⁶. The particles produced act like native LDL <u>in vitro</u> but, when injected <u>in vivo</u>, they are recognised as foreign and cleared rapidly from the circulation by cells of the R.E. system.

Masquelier et al¹⁶¹ found that when AD-32-LDL was injected into mice, 98% of the complex was removed from the plasma within 90 minutes. By comparison only 40% of injected native LDL was removed. High relative uptakes of AD-32-LDL were found in liver, kidneys and spleen, indicating that removal of the complex was mediated largely by the R.E. system. In an effort to reduce the rate of clearance of the complex, the method of drug incorporation was modified. Lyophilisation of LDL with 25% sucrose instead of starch, followed by reconstitution with AD-32 as before, produced a particle which acted like native LDL <u>in vivo</u>. The possibility of using a complex in which AD-32 was covalently linked to lysine residues on apo B was examined. As reported by Mahley⁸², modification of more than 15% of the lysine residues greatly inhibits apo B-receptor interaction. Consequently, degradation of the drug-LDL particle decreased with increasing conjugation of drug to protein. The authors concluded that drug could be incorporated into LDL, producing particles which acted like LDL in vivo.

The use of LDL as a drug vehicle is, of course, not without its draw-Drug-LDL has the potential to harm any of the cells in the backs. body, since they all utilise LDL to some extent. Organs which, under normal circumstances, use large quantities of LDL, namely adrenal glands and liver, are likely to suffer along with tumour during chemotherapy, unless steps are taken to protect them. Drug toxicity will be limited by the rate of the receptor mechanism and it is therefore possible that the receptor pathway could become saturated before toxic drug levels are reached within the target cells. Drug uptake via LDL would also be subject to dietary influence, since cellular drug accumulation would obviously be lowered when receptor expression had been downregulated in response to dietary intake of cholesterol, and Another potential drawback is the current lack of vice versa. techniques for making synthetic LDL which acts like the endogenous particle in vivo. Preparation of native human LDL requires fresh, unfrozen plasma, which would ideally come from the patient who is to receive chemotherapy. However, this may not always be practical, and unless a substantial proportion of the 1500 cholesteryl esters in each LDL particle is replaced by drug, the quantity of LDL required to administer a therapeutic drug dose may be too large in terms of both the volume of plasma required and the induced rise in the level of plasma cholesterol following re-injection of the drug-laden LDL.

1.5 CHLORAMBUCIL, ADRIAMYCIN AND DAUNOMYCIN

Three drugs, namely chlorambucil, daunomycin and adriamycin - (fig.ll), were incorporated into LDL in the course of this project and the following section outlines the modes of action, toxicity and pharmacokinetics of each of them.

1.5.1. CHLORAMBUCIL

Sulphur mustard poisoning in World War I was observed to cause a combination of gastrointestinal tract ulceration, lymphoid tissue dissolution and bone marrow aplasia. The toxic effects of nitrogen analogues of sulphur mustard on lymphoid tissue were studied by Goodman, Gilman and Dougherty in 1942 and their observations on transplanted lymphosarcoma in mice¹⁶² prompted them to undertake clinical trials with the nitrogen mustards, the first of the antineoplastic chemotherapeutic agents. Since then many nitrogen mustard derivatives have been synthesised, although few have had any more cytotoxic activity than the original compound.

All nitrogen mustards are considered chemically unstable. However, modification of the side chain increases the stability of some analogues. For example, chlorambucil (fig.ll), the phenylbutyric acid derivative of mechlorethamine (fig.ll), is stable enough to be prepared in tablet form and administered orally, while cyclophosphamide (fig.ll), is stable and inactive prior to activation by the cytochrome P450 system in the liver^{130,163}. In contrast, mechlorethamine is so hygroscopic that it is prepared as dry crystals and reconstituted with water only, immediately prior to injection.

1.5.2. CHEMICAL ACTION OF CHLORAMBUCIL

The nitrogen mustards act by virtue of their ability to form carbonium ions which subsequently participate in electrophilic attack on

Chemical structures of some alkylating agents and anthracyclines. cock, R HO ANTHRACYCLINES R = H, DAUNOMYCIN R = OH, ADRIAMYCIN Б NH. Ю ٠£ 0: OCH. CH₃ -N CH₁-CH₂-CI CH1-CH2-CI ALKYLATING AGENTS CYCLOPHOSPHAMIDE **MECHLORE THAMINE** CHLORAMBUCIL CH¹CH¹CH¹ ъ Figure 11:

biological molecules¹⁶⁴ (fig.12). They readily form covalent bonds with amino, sulphydryl, hydroxy, carboxy and imidazole groups. Many cell sites are susceptible to this method of alkylation, including DNA, RNA and enzymes. Therefore, cell function and viability may be greatly affected by nitrogen mustards. The alkylation of DNA, however, is thought to be the main mechanism of the cytotoxic action of these agents. By altering DNA, alkylating agents disrupt basic cellular process. They interfere with mitosis and cell division in rapidly proliferating tissues.

Nitrogen mustards exert their effects on cells at all stages of the cell cycle but toxicity is usually expressed in the S-phase with blockage at the G₂ (pre-mitotic) phase¹⁶⁵. However, synchronised cells in culture are generally more sensitive in late G, or S-phases than G_2 , mitosis or early G_1 . This is possibly because polynucleotides are more susceptible to alkylation in the unpaired state, during DNA replication (S-phase), than in the - - helical conformation. Cells accumulated in the G2 phase continue to synthesise cellular materials, without dividing. Lethal toxicity may be the result of damage to many cellular sites, for example, glycolysis, respiration, protein synthesis and nucleic acid synthesis are all processes which may be affected. Non-proliferating cells may later differentiate, without further mitosis, to adult cells. Slowly proliferating cells are more likely to recover from alkylation since the effects of alkylated DNA may not be of great importance prior to mitosis. Slowly dividing cells therefore have a much longer time in which to attempt DNA repair and so overcome the effects of the alkylating agents.

Interaction of bifunctional agents like chlorambucil with DNA tends to have toxic effects, whereas monofunctional agents cause only mutations or carcinogenesis¹⁶⁴. Bifunctional molecules have the ability to cross-link two DNA chains. This is potentially more harmful than



The mechanism of action of alkylating agents 164 Figure 12:

mono-alkylation, where the modified DNA may be compatible with the continued life of the cell and may be successfully passed to the next generation. DNA repair may recover the monoalkylated DNA, whereas highly cross-linked DNA is much more difficult to free from the damaging alkyl groups. High-dose chemotherapy circumvents the DNA repair system to some extent by causing formation of extensively cross-linked DNA, whereas DNA damage during low-dose therapy may be easily corrected.

Due to their properties of lipophilicity, acid dissociation, and stability in aqueous solutions, alkylating agents are able to cross cell membranes. This is crucial to their activity, as are the two (2-chloroethyl) groups. Several different groups have been used as the third substituent on the nitrogen in an attempt to increase drug selectivity and decrease overall toxicity, with varying degrees of success¹⁶⁴. A degree of specificity may be obtained because of the inherent difference in proliferation rates between normal and neoplàstic tissues. Since normal tissues proliferate more slowly, the effects of alkylating agents are not immediately apparent and these cells may repair their DNA successfully before division occurs.

1.5.3. CHLORAMBUCIL - THERAPEUTIC USES AND TOXICITY

Chlorambucil is commonly used in the treatment of leukemia, Hodgkin's Disease and malignant lymphomas. Administration of the drug is usually oral, in doses of 0.1-0.2 mg/Kg daily. Absorption from the gastrointestinal tract is variable but is generally complete within 2 hours. There is little information on the metabolism, distribution and elimination of chlorambucil. Toxic effects of chlorambucil are most apparent in bone marrow, gastro-intestinal tract, skin and occasionally in the liver¹⁶⁶.
1.5.4. RESISTANCE TO ALKYLATING AGENTS

There is little information available on the possible mechanisms of resistance to alkylating agents but it has been widely observed that acquired resistance to one alkylating agent confers cross-resistance to all others¹³⁷. The acquisition of resistance has generally been observed to be slow, and is thought to be a result of several mutations, not just a single one as is the case with antimetabolites. In resistant cells, there is a decrease in the permeability of the cell membrane to the drug. Production of nucleophiles is increased to compete with DNA for alkylation e.g. administration of cysteine¹³⁷ reduces the antineoplastic efficacy of alkylating agents and several resistant animal tumour cells have increased free thiol concentrations as compared to sensitive cells. It is likely too that resistance is potentiated by an increase in the activity of the DNA repair system.

1.5.5. PHARMACOKINETICS OF CHLORAMBUCIL

Alberts et al¹⁶⁷ observed in man that chlorambucil was rapidly absorbed from the gastrointestinal tract and cleared from plasma with a terminal phase half-life of 92 minutes. On administration of chlorambucil at 0.6 mg/Kg, the peak drug concentration was found to be 1.1 μ m/mL plasma. The major metabolite of chlorambucil, phenylacetic acid mustard, was seen in plasma within fifteen minutes of drug administration. Comparison of gastrointestinal and IV routes of administration¹⁶⁸ of chlorambucil showed peak plasma concentrations to be higher after IV injection. Peak plasma concentrations were observed 40-70 minutes after gastrointestinal administration. Thereafter, decay curves for the drug were identical in both cases. Two metabolites were detected, one of which was the β -oxidation product of chlorambucil 2[4-N,N,-bis-(2-chloroethyl)-aminophenyl] -acetic acid (phenyl acetic acid mustard),

a compound known to have alkylating action in animals.

1.5.6. ANTHRACYCLINES

These are relatively new antineoplastic agents. The anthracyclines (fig.11) are antibiotics and were first isolated in the early 1960's from fermentation products of the fungus Streptomyces peucetius, var. caesius. Daunomycin was simultaneously discovered by groups in France¹⁶⁹ and Italy¹⁷⁰ and as a result it is known by several different names (daunorubicin, rubidomycin and daunomycin). Adriamycin was later isolated in Italy¹⁷¹ and differs from daunomycin in only one hydroxy substituent. Structures for both drugs are shown in figure 11. The anthracycline molecule consists of a tetracyclic ring structure joined to the sugar daunosamine by a glycosidic linkage (fig.11). Although their structures are almost identical, the two drugs are effective in the therapy of quite different neoplasms. Daunomycin is used mainly in the treatment of leukemia and lymphomas, whereas adriamycin is effective against a much wider spectrum of tumours.

1.5.7. CYTOTOXIC ACTION OF ANTHRACYCLINES

The mechanism of action of anthracyclines is not fully understood. X-ray diffraction studies and DNA model building 171a suggest that anthracyclines bind tightly to DNA. It is widely accepted that these drugs intercalate the strands of DNA 172 . The \prec helix of the DNA molecule is thus disrupted and no longer forms an effective template for RNA synthesis 173 . Adriamycin has a great affinity for myocardial DNA and this is thought to be a cause of the cardiotoxic properties of this drug 174 . Both drugs may also have cytoplasmic routes of action. Adriamycin is thought to cause intracellular damage as a result of free-radical formation 130 and cell membrane behaviour may also be affected. As expected, maximal cytotoxic effects are expressed, after exposure to these drugs, during the S-phase of the cell cycle, although some cytotoxicity is observed at other stages 129,175.

1.5.8. METABOLISM OF ANTHRACYCLINES

The reason for the observed differences in the spectra of cancers against which these agents are active is largely unexplained, but differences in their metabolism have been observed. Both complexes are extensively metabolised ^{176,177}. The major metabolites are adriamycin-ol and daunomycin-ol which are the results of enzymic reduction of the C-13 ketone¹⁷⁶ and both express antitumour activity. In acute myelocytic leukemia, the activity of "daunomycin reductase" has been directly correlated with patients' response to daunomycin therapy. Two other metabolites of daunomycin¹⁷⁶ and five of adriamycin , all aglycone fragments of the drugs, have been observed. Major sites of anthracycline absorption are heart, lungs, kidneys, liver and spleen^{123,178,179}. Anthracyclines do not appear to cross the blood-brain barrier and, when administered by intrathecal injection¹⁸⁰, were found to be toxic to cells of the central nervous system. The liver is the primary site of anthracycline metabolism and excretion, with 60% of adriamycin and 20% of daunomycin 178 removed here and subsequently excreted in bile¹⁷⁹. These drugs are also excreted in urine, 25% and 6% of adriamycin¹⁷⁹ and daunomycin respectively are removed from the body by this route 178.

1.5.9. THERAPEUTIC USES AND TOXICITY OF ANTHRACYCLINES

Both adriamycin and daunomycin are effective in the treatment of acute leukemias and malignant lymphomas. Daunomycin is also effective in the treatment of solid tumours in children but, unlike adriamycin, it is of limited usefulness against solid tumours in adults. Cardiotoxicity¹⁷⁴ is a major dose-limiting factor in the administration of anthracyclines, and consequently the cumulative dose of these drugs is generally kept below 550 mg/m². Dose regimens for both drugs vary depending on the tumour type and the extent of its progression. Daunomycin is usually given at a dose of 30-60 mg/m² daily for 3 days or once weekly¹⁶⁶. The recommended dose for adriamycin is 60-75 mg/m², administered as a single IV infusion and repeated after 21 days¹⁶⁶. Toxic effects of anthracyclines include stomatitis, alopecia, gastrointestinal disturbances and cardiomyopathy. Myelosuppression is frequently the main dose-limiting factor.

1.5.10 RESISTANCE TO ANTHRACYCLINES

In common with numerous other antineoplastic agents, acquired resistance to anthracyclines is frequently observed and the reasons for it are poorly understood. As in the case of alkylating agents, resistance to one anthracycline confers complete resistance to the other anthracyclines^{138,164}. Cross-resistance has also been reported between anthracyclines and other natural products viz. dactinomycin and the vinca alkaloids^{181,182}. It is likely that this type of cross-resistance is the result of modifications in cellular permeability. Cellular drug accumulation has been found to be lower in several resistant sublines when compared to the sensitive sublines^{183,184}. It has not been established whether this is a consequence of increased drug efflux or decreased influx, although a number of authors have shown the former to be more important in several tumour models in <u>vitro</u>, and <u>in vivo</u>¹³⁸.

1.5.11 PHARMACOKINETICS OF ANTHRACYCLINES

Anthracyclines are administered intravenously and so are immediately distributed in the central compartment. The plasma decay curve for daunomycin is biphasic with half-lives of 45 minutes for the initial

rapid distribution phase, and 55 hours for the terminal phase¹⁷⁸. Adriamycin removal is triphasic, reflecting the participation of a larger number of metabolites in adriamycin pharmacokinetics. Halflives of the drug in the three clearance phases are 12 minutes, 3.3 hours and 29.6 hours, as measured by Bachur et al¹⁷⁹.

SECTION 2 .

.

.

: MATERIALS, METHODS AND RESULTS

Suppliers of the materials used in the following experimental sections are detailed below. All other reagents and solvents were of Analar grade.

Adriamycin and daunomycin were supplied as gifts from Farmitalia Carlo Erba, Milan, Italy, and May & Baker Ltd., Dagenham, U.K., respectively. Chlorambucil was donated by The Wellcome Foundation, Crewe, U.K., or purchased from Sigma Chemical Co. Ltd., Poole, Dorset.

Radioisotopes (^{125}I , ^{131}I , $[7-^{3}H]$ -cholesterol, and $[4, 5-^{3}H]$ -leucine) were obtained from the Radiochemical Centre, Amersham, Bucks.

MATERIAL

SUPPLIER

Bio-Rex 70 Cation Exchanger	Bio-Rad Laboratories, Watford, Herts
Brij-35	Sigma Chemical Co.,
	Poole, U.K.
Cellobiose	Sigma
Cholesterol	British Drug Houses (BDH)
	Poole, U.K.
Cholesterol Oleate	Sigma
Cholesterol Tosylate	11
Cholic Acid	BDH
Cyanuric Chloride	Sigma
1,2-Cyclohexanedione	Fluorochem Ltd., Glossop, Herts.
Dimethyl Hydrazine	Sigma
Dimyristoyl Phosphatidyl Choline	"
Dipalmitoyl " "	n
Na ₂ EDTA .	BDH

SUPPLIER

Dulbecco's Modified Eagle's Medium	Gibco Biocult Ltd., Paisley, U.K.
Folin-Ciocalteu Reagent	BDH
Glycine	11
Ham's F10 Medium	Gibco
Human Serum Albumin	Sigma
Hydrogen Peroxide Solution(40% w/v)	BDH
Hydrocortisone sodium succinate for injection	Organon Laboratories, Newhouse, Scotland
Iodine Monochloride	Sigma
Lecithin	11
Microtitre Cell Culture Plates	Corning Medical Halstead, U.K.
Millipore Filters	Millipore Corpn., Bedford, Mass.
N-Nitrosobis(2-oxopropyl)amine	Ash Stevens Inc.,
	Detroit, USA.
PD10 Columns	Pharmacia(GB)Ltd, London
QAE-A50 Anion Exchanger	11
Scintillation Liquid (Dimilume 30)	Packard Instruments
Silica	Sigma
Silica Gel60 $_{F254}$ TLC Plates	BDH
Sodium Dodecyl Sulphate	BDH
" Iodoacetate	Sigma
" Potassium Tartrate	BDH
" Taurocholate	Sigma
Spectrapor Dialysis Tubing	Spectrum Medical Industries Los Angeles, USA.
Tissue Culture Flasks	Flow Labs., Irvine, Scotland
Tris(hydroxymethyl)methylamine	Sigma
Triton X-100	Sigma
Trypsin .	Gibco

EQUIPMENT

Beckman L5-65 Centrifuge and rotors	Beckman Instruments Ltd., Glenrothes, Scotland	
Isco UA ₅ U.V. Absorbance Monitor	MSE Scientific Instruments Crawley, W. Sussex	
Amicon B15 Concentrating Cells and Membranes	Amicon Corpn. Lexington, Mass., U.S.A.	
Packard Scintillation Counters	Packard Instruments Illinois, U.S.A.	
Iridectomy Scalpel & Trocar	MacCarthy's Surgical Instruments Dagenham, U.K.	

ANIMALS & DIETS

NMRI and CFLP mice were obtained from inbred colonies at the Departments of Medical Oncology, Glasgow University, and Pathology, G.R.I., respectively. NIC B6D2F1 mice were purchased from Bantin and Kingman Ltd., Hull, U.K.

Rats, Sprague-Dawley and Wistar were purchased from Strathclyde University and Lions Lab., Ringwood, Hants., respectively.

New Zealand White rabbits were purchased from Leslie Moore Ltd., Bradford, U.K. and fed diet SG1 (Oxoid Ltd., Basingstoke, U.K.). Standard small animal diet was obtained from Special Diet Services, U.K, Cholesterol and bile salt-enriched diets were prepared as described in Section 2.1.13.

.

2.1.1. PREPARATION OF LDL

LDL was separated from human plasma by rate-zonal ultracentrifugation by the method of Patsch et al¹⁸⁵. A diagram of the apparatus used in this procedure is shown in figure 13. Using a Beckman L5-65 ultracentrifuge and Ti-14 zonal rotor, the LDL from 40-60 ml of human plasma was separated from other plasma constituents by flotation through a linear sodium bromide gradient, density range 1.0-1.3 Kg/L, at 45,000 rpm and 10°C for 110 minutes. Eluant from the rotor was collected in fractions of 14 mL and the optical density at 280 nm monitored using an Isco UA5 Ultraviolet Absorbance Monitor. A typical elution profile is shown in figure 14. The peak fractions containing LDL were pooled and concentrated by pressure filtration in an Amicon ultraconcentration cell using an XM100 Å filter, under nitrogen pressure at 25 p.s.i. Excess sodium bromide was removed by washing with working buffer (0.15M NaCl, 0.01% EDTA, pH 8.1) and reconcentrating, several times, or by exhaustive dialysis against working buffer.

2.1.2. PREPARATION OF HDL AND d>1.21 Kg/L PLASMA FRACTION

HDL was prepared from human plasma by sequential ultracentrifugation at 1.063 Kg/L and 1.21 Kg/L¹⁸⁶. Human plasma was made up to density 1.063 Kg/L by the addition of 0.0834 g/mL of solid potassium bromide, overlayed with solution of the same density and centrifuged at 45,000 rpm and 10°C for 18 hours in a Beckman Ti60 rotor and L5-65 ultracentrifuge. Buoyant lipoproteins of density less than 1.063 Kg/L were aspirated. The solution containing HDL and plasma proteins at the bottom of the centrifuge tubes was made up to density 1.21 Kg/L by the addition of 0.236 g/mL of solid potassium bromide, overlayered with 1.21 Kg/L density solution and centrifuged overnight as above.







PREPARATION OF HUMAN LDL

HDL (d 1.063 - 1.21 Kg/L) was aspirated from the tops of the tubes and desalted, either by dialysis against working buffer or gel matrix filtration through Pharmacia PD10 columns with working buffer (0.15 M NaCl, 0.01% EDTA, pH 8.1) as eluant.

The remaining fraction at the bottom of the tube was dialysed against distilled water for 24 hours at 4°C, prior to further dialysis against working buffer (0.15M NaCl, 0.01% EDTA, pH 8.1) with several changes of solution.

2.1.3. RADIOIODINATION OF LDL

The protein moiety of LDL was labelled using a modification 93 of the iodine monochloride method of McFarlane¹⁸⁶. This involves the substitution of iodine at the meta-position on the phenyl ring of tyrosine residues of the apoprotein. Since this reaction is only 40% efficient with LDL, a 2.5-fold excess of ICl is required in order to label each mole of apoprotein, taken as 64000 daltons, with one mole of iodine. Apoprotein concentration was determined by the method of Lowry et al¹⁸⁸, Section 2.1.9. or estimated by measuring absorbance at 280 nm on a Beckman Du-7 Spectrophotometer and relating 1 optical density unit to a concentration of 1 mg/mL. (It should be noted here that, since this method was first described, the molecular weight of apoprotein B has been reported to be 500,000 Daltons. Thus the estimation of apo B concentration here was purely empirical and overestimates the amount of protein present by a factor of 7.8. Consequently each mole of apoprotein is labelled with 7.8 moles of iodine). To each mL of LDL solution (protein 2-5 mg/ml) prepared as described in Section 2.1.1. was added 250 μl of 1M glycine buffer, pH 10.

After addition of the required amount of ^{125}I or ^{131}I , usually 1-2 mCi in 10-20 μ L, the appropriate volume of 25 nM ICl was added, giving a final ICl concentration of 125-315 nM.. The reaction is complete in a few seconds and glycine and unreacted iodide were removed by gel filtration through a Pharmacia PD10 column, using working buffer (0.15M NaCl, 0.01% EDTA, pH 8.1) as eluant.

2.1.4. SYNTHESIS AND IODINATION OF CHOLESTERYL LINOLEOYL ETHER

Cholesteryl Linoleoyl ether (CE), figure 8, was synthesised as described by Halperin and Gatt¹⁸⁹. The cholesteryl ether was labelled by mixing 15 mg of the ether, in 250 μ L of chloroform, with 250 μ L methanol, 2mCi of ¹²⁵I or ¹³¹I and 250 μ L of 25nM ICl. This mixture was washed several times with distilled water to remove unreacted iodide. The labelled ether was further purified by thin-layer chromatography on silica gel 60_{F254} plates. These were developed in heptane: ethyl acetate (9:1) and the cholesteryl ether band (which migrated near the solvent front leaving iodide at the origin) was scraped off into a plastic bag. The radioiodinated cholesteryl ether (I-CE) was extracted into chloroform (3 x 5 mL) and transferred to a pre-weighed glass tube before evaporation of the solvent under nitrogen at 40°C. The yield of I-CE was usually ca. 10 mg.

2.1.5. INCORPORATION OF LABELLED CHOLESTERYL LINOLEOYL ETHER INTO LDL

This procedure (figure 15) has been adapted from Stoffel's method¹⁹⁰ for loading HDL with labelled cholesteryl esters and fatty acids. In brief, the I-CE was co-sonicated with cholic acid and lipids to generate micelles which, when mixed with HDL undergo exchange of lipids. The ester-containing HDL was purified and mixed with LDL and the d > 1.21 Kg/L fraction from human plasma which was a source of cholesteryl ester transfer protein¹⁹¹.



Figure 15: Protocol for the preparation of radioiodinated cholesteryl ether-LDL (I-CE-LDL) by the modified Stoffel method, Section 2.3.2. The labelled LDL was purified by centrifugation, and desalted by gelmatrix chromatography.

A) PREPARATION OF CHOLATE MICELLES¹⁹⁰

Phospholipid, cholesterol and I-CE, in chloroform, were mixed in molar ratios of 80 : 24 : 53 and dried into the surface of a conical glass tube under a stream of nitrogen. This required 10.9 mg lecithin (13.6 μ mol), 1.6 mg cholesterol (4.1 μ mol) and 10 mg I-cholesteryl ether (9.0 μ mol). Cholic acid was dissolved in tris buffer (0.02 M tris(hydroxymethyl) methylamine, 0.02 M NaCl, 0.02 M EDTA, pH 8.2) to a final concentration of 4.6 mg/mL. Equivalent weights of cholic acid and lipids (i.e. 5 mL cholate solution to above quantities of lipids) were mixed and sonicated at 0°C for 30 minutes.

B) LOADING OF HDL WITH CHOLESTERYL ETHER

Cholate micelles and HDL were mixed 1 : 1 (weight/weight) and stirred overnight at room temperature. After 18-24 hours cholate micelles were removed by filtration of the mixture through 0.22 µ^m millipore filters. The HDL solution was made up to density 1.063 Kg/L by the addition of 0.0834 g/mL of KBr, overlayered with d1.063 Kg/L density solution and centrifuged at 45,000 rpm for 18 hours in a Beckman Ti60 rotor and L5-65 centrifuge. Fragments of HDL and lipid micelles and density solution were aspirated from the top of the centrifuge tube, leaving HDL labelled with I-CE at the bottom. The I-CE-HDL was desalted by filtration through a Pharmacia PD10 column, eluting with working buffer (0.15M NaCl, 0.01% EDTA, pH 8.1).

C) TRANSFER OF CHOLESTERYL ETHER FROM HDL TO LDL

By utilising the cholesteryl ester transfer protein found in human $plasma^{57,58,191}$ the I-CE was transferred from HDL to LDL. Fifteen mL of d>1.21 Kg/L bottom fraction was incubated with I-CE-HDL and 5 mg LDL (protein) for 24 hours at 37°C.

D) PURIFICATION OF LABELLED LDL

The above mixture, after incubation, was made up to density 1.063 Kg/L and overlayered with d1.063 solution prior to centrifugation at 45000 rpm and 10°C in a Beckman Ti60 rotor. Labelled LDL was removed by aspiration from the tops of the centrifuge tubes and desalted on a Pharmacia PD10 column using working buffer or phosphate - buffered saline, (PBS; 0.15M phosphate; 0.15 M NaCl pH 7.4) as eluant.

2.1.6. TRANSFER OF CHOLESTERYL ETHER FROM LDL TO OTHER LIPOPROTEINS

Man and many animals possess a cholesterol ester transfer protein^{57,58,191} which facilitates the exchange of cholesteryl esters between the lipoprotein classes. If I-CE-LDL is to be used as a tracer of LDL catabolism, it is essential that the animal whose metabolism is being studied does not have high cholesteryl ester transfer activity. It was therefore necessary to establish which of the animal models used had suitably low transfer activity.

 $^{125}\text{I-cholesteryl}$ ether HDL was prepared as described above (2.1.5.) One mL of mouse, hamster, rat, rabbit and human plasma were each incubated with 10 μL of $^{125}\text{I-CE-HDL}$ (approx. 20,000 cpm) at 37°C for 24 hours.

The LDL from each sample was separated by centrifugation at 1.063 Kg/L in a Beckman Ti40.3 rotor at 39K rpm and 4° C overnight. On completion of the spin, 1 mL fractions were removed from each tube and their 125 Iactivity measured using a Packard Twin-Channel Autogamma Liquid Scintillation Counter. The percentage of the total activity found in each fraction is shown in figure 16. It was concluded that hamster and rabbit plasma have high transfer activity, like human plasma, whereas mouse and rat have much lower activity. The use of I-CE-LDL as a probe of LDL catabolism is therefore likely to be confounded by transfer of the cholesteryl ether between lipoproteins in the former two species.

2.1.7. CELLOBIOSE-TYRAMINE-LDL:SYNTHESIS AND LABELLING

Cellobiose is a disaccharide $(4-0-\beta-D-Glucopyranosyl-D-\leftarrow Glucopyranose)$ which has a reducing aldehydic group available on the second ring. Tyramine is a primary amine which will react rapidly with aldehydes to form a Schiff's base. This linkage can then be reduced to make a stable bond to LDL-lysine residues via cyanuric chloride (figure 8). Cellobiosetyramine (CT) was synthesised in a manner similar to that described by Pittman et al¹⁰⁵. 1 mmol each of cellobiose and tyramine and 5 mmol of sodium cyanoborohydride were suspended in 10 mL of distilled water and dissolved by heating to 50-60°C in a stoppered glass tube. The mixture was allowed to react at 37°C for 24 hours. Unreacted tyramine was removed by chromatography using a 1.6 x 100 cm column of Sephadex G15 in 0.1 M ammonium acetate. The reaction mixture was applied to the column and eluted with ammonium acetate buffer at a flow rate of 20 mL/hr. Fractions were collected and assayed for tyramine, by measuring absorbance at 280 nm in a Beckman DU-7 spectrophotometer, and sugar by the anthrone reaction. Two peaks were obtained, the first contained 75% of the

ACTIVITY OF PLASMA CHOLESTEROL ESTER TRANSFER PROTEIN IN VITRO



and centrifugal separation of HDL from LDL as described in Section 2.1.5 The presence of radioactivity in fraction 1 indicates cholesteryl ester plasma following incubation of the labelled lipoprotein with the plasma Percentages of total 125 I-CE found in fractions of animal and human transfer activity.

Figure 16:

applied OD_{280} and all of the sugar, the second contained the remainder of the OD₂₈₀, that is, non-conjugated tyramine. Excess celloboise was removed by ion exchange chromatography using Bio-Rex 70 Cation (Na⁺ form) Exchanger, 200-400 mesh. The resin was washed in 50 volumes of 0.12N HCl, three times, followed by three washes with 50 volumes of water. The column was packed and eluted with water until the eluate was of pH 5-6. Peak I from the sephadex column was applied in three aliquots, each interspersed with 5 mL water. The unretained peak was eluted with water and was found to contain The water elution was continued until there was no sugar. detectable absorbance at 280 nm. The retained, positively charged conjugate was eluted using 3.5% ammonia solution. Ammonia and water were removed by overnight lyophilisation leaving pure cellobiosetyramine.

To 250 nmol of CT in 50 μ L of distilled water was added 10 μ L of 0.3M KH₂PO₄, pH 4.5, 10 μ L of Na ¹²⁵I or Na ¹³¹I solution (lm Ci) and 10 μ L of ICl (250 nmol). The mixture was allowed to stand for 15 minutes before the addition of 500 μ L of distilled water. Unreacted iodide was removed on a 0.9 x 6 cm column of QAE-A50 (Anion Exchanger) in distilled water. The I-CT was eluted in 3-4 mL of water and lyophilised overnight.

Linkage of CT to LDL : A dried aliquot of the freshly made I-CT was dissolved in 5 x 10 μ L of distilled water and 10 μ L of the solution mixed with 20 μ L of cyanuric chloride (2.5 mg/mL in acetone), 15 μ L of 0.008 M NaOH and 10 μ L of distilled water. This was allowed to stand for 15 seconds before the rapid addition of 5 mg LDL protein in 1 mL of working buffer. After 2-3 hours at room temperature, the I-CT-labelled LDL was purified on a Pharmacia PD10 column, eluting with working buffer.

2.1.8. CHEMICAL MODIFICATION OF LDL

In order to prevent binding of LDL to its receptors, the arginine residues of apo B were modified by 1,2- cyclohexanedione (CHD), as described by Mahley et al⁹⁰, (figure 17). 0.15M CHD was made up in 0.2 M sodium borate buffer pH 8.1. Two volumes of CHD solution were mixed with each volume of the LDL solution (protein concentration 2-5 mg/mL), and incubated at 35°C for two hours. Sodium borate and unreacted CHD were subsequently removed by gel filtration through Pharmacia PD10 columns, eluted with working buffer. This process modifies at least 55% of the arginine residues in the apoprotein B⁹⁰, so preventing CHD-LDL from binding to LDL receptors.

Reversal of modification <u>in vitro</u> is possible by incubating the CHD-LDL with a strong nucleophile such as hydroxylamine^{90,98}. For this reason tris buffers should not be used in the preparation of CHD-LDL.

Reaction of the arginine molecules with CHD removes the positive charge characteristic of arginine residues. As a result, the modified LDL has substantially more anodic mobility than native LDL on agarose gel electrophoresis.

2.1.9. PROTEIN DETERMINATION

The apoprotein content of LDL was assayed by the method of Lowry et al¹⁸⁸ using human serum albumin as standard. Reagents were prepared as follows: Solution A - 2% NaHCO₃ in 0.1N NaOH;

Solution B - 2% NaK Tartrate in deionised water;

Solution C - 1% $CuSO_A$ in deionised water.

Biuret reagent - to 100 mL of solution A was added 1 mL of solution B and 1 mL of solution C.



Chemical modification of LDL with 1,2-cyclohexanedione(CHD), DHCH - dihydroxycyclohexyl

Figure 17:

Folin-Ciocalteu reagent - the stock reagent was diluted 1 : 1 with distilled water.

Standards

Human serum albumin 1 mg/mL was diluted with deionised water to give 25-100 μg protein / 400 μL . Bovine serum albumin, 0.3 mg/mL, was diluted 1 : 4 and used as a quality control.

Method

To each 400 μ L sample, control and standard was added 2 mL of Biuret reagent. Following vortexing and a 10-minute stand, 200 μ L of Folin-Ciocalteu reagent was added to each tube during further vortexing. Colour was allowed to develop for 30 minutes and the optical density at 750 nm was measured using a Beckman DU7 spectrophotometer. A standard curve of OD₇₅₀ vs. protein concentration was constructed and unknowns were read from the curve.

2.1.10. STERILISATION OF SAMPLES FOR INJECTION

All solutions used for injection into animals were sterilised by \cdot filtration through sterile millipore filters, pore size 0.22 μ m.

2.1.11. COLLECTION OF BLOOD SAMPLES

Blood removed from a peripheral vein was placed in 10 mL glass tubes containing 0.1% K_2 EDTA, unless serum was required, in which case 10 mL glass tubes containing no anticoagulant were used.

2.1.12. INDUCTION OF TUMOURS

A) NMRI/MAC13 - This was carried out at the Department of Oncology at Glasgow University with the cooperation of Dr J Welsh. Solid MAC-13 tumour was passaged in two-month old NMRI mice from an inbred colony at the above department by subcutaneously injecting each animal with approx 1mm^3 of finely minced tumour. After 10 days the tumour had grown to approximately 1.5 g in weight $(1.49 \stackrel{+}{-} 1.13\text{g}, n=14)$. At this stage the animals were used in the study, or tumour was passaged through another group of mice.

B) CFLP/PAROTID ADENOMA - Parotid adenomata were induced in these mice by inoculating them subcutaneously in the neck with A2LP polyoma virus within 24 hours of birth, as described by Lamey et al¹⁹². Following a latent period of 10-12 weeks the tumours became visible and therafter the animals were ready for use in ca.3 weeks. At this stage the tumours weighed approximately 0.5 g $(0.43^+ 0.35 \text{ g}, n=8)$.

C) SPRAGUE - DAWLEY RATS/ COLONIC TUMOURS - Rats weighing 300g were injected for 16 weeks with dimethyl hydrazine in 0.05% EDTA solution, at a dose of 15 mg/Kg. Controls were injected with EDTA solution only¹⁹³. Two to three weeks after the last injection, tumours were approximately 1g in size $(0.92 \pm 0.87, n = 25)$.

D) WISTAR RATS/WALKER2.56 Tumour - Walker256 tumour was maintained in serial subcutaneous passage, in a colony of inbred Wistar rats, by Dr K Fearon at the Department of Oncology, G.U. Tumour was cut into fragments of 100 mg and placed under the skin in the flanks of rats weighing 200-250g. After 10 days, the tumours had grown to approximately 8g $(8.01^+3.75, n=8)$. The rats were used at this stage. This type of tumour was used for implantation in the subrenal capsular assay, described later, Section 2.3.14.

E) HAMSTERS/PANCREATIC TUMOURS - Pancreatic tumours were induced in syrian hamsters by repeated weekly injections of N-nitrosobis (2-oxopropyl) amine in normal saline at a dose of 5 mg/Kg over 23 weeks as described by Pour et al¹⁹⁴.

2.1.13. ANIMALS AND DIETS

Unless otherwise stated, rats, mice and hamsters were maintained on standard small animal diet and allowed free access to drinking water (containing 1% potassium iodide for 1 day before and during LDL turnover studies, to inhibit thyroidal uptake of radioiodide).

Sodium Taurocholate - Enriched Diet

Standard small animal diet was spray-coated with a saturated solution of sodium taurocholate in ethanol. The ethanol was allowed to evaporate and the preparation had a final sodium taurocholate composition of 1% (W/W).

High Cholesterol Diet

Cholesterol was heated in coconut oil (1g in 10) and used to coat standard diet pellets. The final cholesterol content of the diet was 1% (W/W).

2.1.14. MEASUREMENT OF LDL RECEPTOR ACTIVITY IN VIVO

LDL was prepared by rate-zonal ultracentrifugation of human plasma, as previously described (2.1.1.). Aliquots of the concentrated and desalted LDL were labelled, one with ¹²⁵I and the other with ¹³¹I (2.1.3.). One aliquot, usually the ¹³¹I-labelled, was then chemically modified with cyclohexanedione. (2.1.8.) Tumour and control animals were each injected with 0.1 mL of solution containing 5 μ Ci of each LDL tracer via a tail vein in the mice, and by jugular vein in the rats. After 18 hours, the animals were sacrificed and exsanguinated. The organs were removed, rinsed in saline, blotted, weighed and ¹²⁵I-LDL and ¹³¹I-CHD-LDL radioactivity measured using a Packard Twin-channel Autogamma Liquid Scintillation Counter. Results were expressed as counts per minute/ gram of tissue per mL of plasma for each isotope.

This approach relies on the observation that CHD-modified LDL is cleared more slowly than native LDL from the plasma (figure 6) since it is unable to interact with LDL receptors and cannot therefore be catabolised via the LDL receptor pathway⁹⁰. CHD-modified LDL is consequently cleared only by receptor-independent mechanisms. Whereas native LDL is subject to catabolism via receptor-dependent and receptor-independent routes. Thus LDL receptor activity may be expressed as the difference between native- and CHD-LDL clearances (figure 18), if it is assumed that the distribution of modified LDL is identical to that of native LDL in the plasma and extravascular spaces.

Tissue radioactivity due to uptake of native LDL is a composite of receptor-bound, internalised and degraded LDL, and a non-specific component of LDL trapped in interstitial fluid. Since no

LDL RECEPTOR ACTIVITY

RECEPTOR-INDEPENDENT routes uptake by RECEPTOR-DEPENDENT and 11 NATIVE-LDL uptake

uptake by RECEPTOR-INDEPENDENT routes only 11 CHD-LDL uptake

2

.....

thus

1 - 2 = RECEPTOR ACTIVITY

Calculation of LDL receptor activity using native and 1,2-cyclohexanedione (CHD)-modified LDL Figure 18:

difference is assumed between the distribution of native and chemically modified lipoproteins in interstitial fluid, calculated CHD-LDL uptake contains this same non-specific component. The calculated receptor-mediated LDL uptake into each tissue may then be attributed to receptor- bound, internalised and degraded LDL. The low-molecular weight LDL degradation products after extraction of tissue homogenates with trichloroacetic acid were found to correspond to less than 10% of the total radioactivity in all tissues except kidney. Ninety percent of the observed LDL uptake was then a result of the presence of high molecular weight, presumably intact lipoprotein. In the kidney 35% of the total radioactivity consisted of degradation products, reflecting the role of this organ in the excretion of iodotyrosine. Radioactivity assimilated via the receptor pathway generally accounted for 25-40% of the total tissue activity. The major breakdown products of LDL labelled in this way are rapidly excreted from the cell. Thus LDL receptor activity measured using I-LDL as a tracer reflects only the LDL metabolism occuring immediately prior to assessment in any particular tissue. In order to measure cumulative LDL metabolism, it is necessary to employ a tracer which is not excreted so rapidly from the cells which it enters. To this end, several cumulative probes of LDL catabolism have been investigated. Stein et al¹⁰⁶ have used radioiodinated cholesteryl ether and, more recently, Pittman et al have studied radioiodinated cellobiosetyramine.

2.2.1. MEASUREMENT OF RELATIVE LDL RECEPTOR ACTIVITY IN ANIMAL TUMOUR MODELS USING NATIVE AND 1,2-CYCLOHEXANEDIONE-MODIFIED LDL.

Eighteen hours after injection of native and CHD-modified LDL into control and tumour-bearing animals, the radioactivity in their tissues and plasma was measured and used to calculate relative receptor-mediated LDL uptakes (2.1.14). The values obtained for each tumour type are shown in tables 5,7-9 and figures 19,21-23.

A) MAC-13 TUMOUR IN NMRI MICE

All tissues examined showed some degree of receptor-mediated LDL uptake (figure 19). The liver is clearly the most active in receptor expression per gram of tissue (table 5), and per organ (figure 19). Apparently high receptor activity is found in the kidney (table 5) although this is partly a result of this organ's role in the excretion of the degradation products of iodinated LDL, namely iodotyrosine and free iodide - note the high TCA-soluble activity in figure 19. In animals with tumours, receptor-mediated hepatic assimilation of the lipoprotein fell (p<0.05, Wilcoxon Rank Test) while uptake into other tissues remained the same as in the control situation. The tumour was moderately active in taking up LDL by the receptor pathway, and when account was taken of its mass (figure 19) it was found to be second only to liver in activity. The presence of MAC-13 tumour in these mice did not perceptibly change their total plasma cholesterol value (table 6), nor did it change the whole body clearance of native or CHD-modified LDL (figure 20).



Figure 19:

Assimilation of LDL via the receptor pathway, in NMRI mouse tissues and MAC-13 tumours,. measured as described in Section 2.1.14. The low molecular weight degradation products of the lipoprotein were extracted from tissue homogenates using trichloroaceticacid(TCA).

	RELATIVE RECEPTOR-MEDIATED LDL UPTAKE	
TISSUE	CONTROL GROUP (n=11)	TUMOUR GROUP (n=15)
Liver	30±11	21 <u>+</u> 9 **
Spleen	14 <u>+</u> 11	15 <u>+</u> 7
Kidney	23 <u>+</u> 12	26 <u>+</u> 5
Heart	13 <u>+</u> 5	19 <u>+</u> 6
Lung	14±9	ND [*]
Muscle	6 <u>+</u> 3	11 <u>+</u> 3
Tumour	-	16 <u>+</u> 8

Table 5 : Receptor-mediated LDL uptake into the tissues of NMRI/MAC 13 mice¹⁵².

- * ND : Not determined.
- ** Wilcoxon Rank Test, p<0.05

GROUP	PLASMA CHOLESTEROL (mmol/L)
Control* (N=11)	2.91-1.07
Control, cholesterol fed (n=12)	4.62-0.64
Tumour* (n=15)	2.72-0.81
Tumour, cholesterol fed (n=15)	4.44-0.79

* Tissue uptakes of LDL for these groups are shown in Table 13.

Table 6 : Effects of cholesterol feeding on plasma cholesterol concentrations in control and MAC-13 tumour inoculated NMRI mice¹⁵².

DECAY OF NATIVE - AND CHD - I.DL IN CONTROL AND TUMOUR - BEARING NMRI MICE



 $^{125}\mathrm{I}_{-\mathrm{native}-}$ and $^{131}\mathrm{I}_{-\mathrm{CHD}-}$ LDL in control and MAC-13 tumour-inoculated NMRI mice Plasma decays of Figure 20:

B) POLYOMA VIRUS-INDUCED ADENOMATA IN CFLP MICE

The pattern of receptor-mediated LDL assimilation observed in NMRI mice was repeated in this model. Uptakes of LDL per gram of tissue are shown in table 7 . Again the liver, the most active organ in the control group, was suppressed by the presence of tumour (p<0.05, Wilcoxon Rank Test). In these animals the opportunity was taken to measure adrenal LDL uptake. As expected, it was very high per gram of tissue (table 7), but because of its size, made a negligible contribution to overall LDL catabolism (figure21). The tumour was also very active on a per gram basis and was matched only by hepatic uptake when its mass was taken into account (figure 21).

C) DIMETHYLHYDRAZINE-INDUCED COLONIC TUMOURS IN RATS

As in the mouse models, liver was the most active normal organ in receptor-mediated uptake of LDL, whether uptake was expressed per gram (table8) or per organ (figure 22). The presence of this type of tumour in the rats did not suppress hepatic LDL uptake (figure 22), however, there was a significant (p<0.05, Wilcoxon Rank Test) reduction in receptor-mediated assimilation of LDL by the ovaries of animals with tumours. Once again, both adrenals and ovaries had high LDL receptor activity per gram (table 8) although their contribution to overall LDL catabolism is minimal (figure 22). Tumour made a large contribution to receptor-mediated LDL catabolism with higher uptake of LDL by this pathway than any other organ.

	RELATIVE RECEPTOR-MEDIATED LDL UPTAKE (CPM/GRAM TISSUE/mL PLASMA)	
TISSUE	CONTROL GROUP (n=9)	TUMOUR GROUP (n=9)
Liver	60 <u>+</u> 14	27 <u>+</u> 9 [»]
Spleen	22 <u>+</u> 18	22 <u>+</u> 7
Kidney	36 <u>+</u> 12	28 <u>+</u> 10
Heart	31 <u>+</u> 5	23 <u>+</u> 7
Lung	36 <u>+</u> 16	24 <u>+</u> 7
Muscle	8 <u>+</u> 2	7 <u>+</u> 15
Adrenal	51 <u>+</u> 36	36 <u>+</u> 24
Tumour	-	31 <u>+</u> 9

. Table 7 : Receptor mediated LDL uptake into the tissues of CFLP/viral adenoma $\rm{mice}^{152}.$

Wilcoxon Rank Test, p<0.05</p>



Receptor mediated organ uptake of LDL in CFLP/adenoma mice

Figure 21:

Uptake of LDL via the LDL receptor pathway, in CFLP mouse tissues and polyoma virusinduced tumours, measured as described in Section 2.1.14.
	RELATIVE RECEPTOR-MEDIATED LDL UPTAKE (CPM/GRAM TISSUE/mL PLASMA)					
TISSUE	CONTROL GROUP (n=3)	TUMOUR GRÓUP (n=3)				
Liver	88 <u>+</u> 3	104 <u>+1</u> 5*				
Adrenal	71±10	63 1 16				
Spleen	58 <u>+</u> 13	57 - 10				
Ovary	70 * 3	29-7 *				
Lung	72 ± 13	50 * 8 #				
Kidney	70±6	83 + 24				
Gut	64 <u>+</u> 2	64 - 12				
Heart	28_+5	26 + 6				
Muscle	8±0.2	8+3				
Fat	5 ± 1	14-4				
Tumour	-	16035				
		· · · · · · · · · · · · · · · · · · ·				

- Table 8 : Receptor-mediated uptake of LDL by the tissues of normal rats and those with dimethylhydrazineinduced colonic tumours.
 - * Wilcoxon rank test vs controls, p<0.05,
 - # Not significant



Receptor-mediated uptake of LDL by rat tissues and colonic tumours, Figure 22:

measured as described in Section 2.1.14.

D) WALKER-256 TUMOUR IN WISTAR RATS

This tumour was very active in receptor-mediated LDL assimilation. When compared to rat vital organs, tumour came second only to liver (figure 23) and gram for gram (table 9) tumour was more active than any other type of tissue assayed. Adrenals and spleen had high uptake per gram of tissue, but as organs they catabolised only a small proportion of the LDL injected (figure 23).



Figure 23:

Relative receptor-mediated uptake of LDL by rat organs and Walker256 tumour implants.

	RELATIVE RECEPTOR-MEDIATED LDL UPTAKE
TISSUE	(CPM/GRAM_TISSUE/mL_PLASMA)
Tumour	36 – 12
Liver	. 29–5
Spleen	29 ⁺ 5
Adrenal	17-10
Lung	. 13-5
Kidney	10-3
Ovary	8-2
Gut	7-10
Skin	5-4
Muscle	1-2
Fat	1-1

Table 9 : Relative Receptor-Mediated uptake of LDL into tissues of 8 rats with Walker 256 Tumour implants.

.

2.2.2. MEASUREMENT OF LDL CATABOLISM USING CUMULATIVE NON-DEGRADABLE MARKERS

In an attempt to improve the assessment of LDL uptake into tumour tissues 125 I-cellobiose-tyramine-LDL 105 and 125 I-cholesteryl linoleyl ether-LDL 106 (figure 14), were used as probes. It has been reported 105,106 that, once assimilated by cells, the leakage rate of these tracers is less than 35% per day compared with the instantaneous release of iodotyrosine from cell-ingested radioiodinated LDL.

Animals were injected with native ¹²⁵I-CT-LDL (2.1.7.) or ¹²⁵I-CE-LDL (2.1.5.) and after 18 hours tumour and tissues were removed, rinsed, blotted and assayed for radioactivity. The uptake of labelled LDL into each tissue was expressed as a percentage of the injected dose per organ and per gram of tissue. In NMRI mice with MAC-13 tumours receptor activity was measured after injection of ¹²⁵I-native- and ¹³¹I-CHD-modified-cholesteryl ether LDL.

A) MAC-13 TUMOURS

In the control group adrenal uptake of CE-LDL (per gram) by the receptor pathway was highest of all tissues (figure 24). Uptake by adrenals in tumour-inoculated animals was significantly lower (p<0.05 Wilcoxon Rank Test), table 10, a phenomenon which was not observed using I-LDL as a probe (figure 20). Most tissues in tumour animals showed slightly lower LDL assimilation by both receptor-dependent and receptor-independent routes. Using this marker of LDL catabolism, relative tumour LDL receptor activity was quite low, although it was second only to liver when considered as a whole.



Figure 24:

	RELATIVE RECEPTOR-M	MEDIATED LDL UPTAKE DOSE) x 100		
TISSUE	CONTROL GROUP (n=6)	TUMOUR (n=3)		
Liver	14.1 <u>+</u> 1.1	10.7 <u>+</u> 0.2 *		
Spleen	0.35 <u>+</u> 0.51	0.44 <u>+</u> 0.03		
Kidneys	0.12 <u>+</u> 0.01	0.08 <u>+</u> 0.009*		
Adrenals	0.05 <u>+</u> 0.005	0.02 <u>+</u> 0.007*		
Heart	0.03 <u>+</u> 0.007	0.02 <u>+</u> 0.002		
Lung	0.09 <u>+</u> 0.008	0.13 <u>+</u> 0.01		
Thymus	0.03 <u>+</u> 0.002	0.04 <u>+</u> 0.001		
Gonads	0.018 <u>+</u> 0.004	0.018 <u>+</u> 0.002		
Tumour	-	2.13 <u>+</u> 0.18		

Table 10 : Relative receptor-mediated uptake of

Cholesteryl ether-LDL by NMRI mouse organs and MAC-13 tumour.

. •

* Wilcoxon rank test vs controls, p<0.05.

In the rats, spleen had the highest uptake of 125 I-CT-LDL/g (table 11). It is likely that this high activity largely represents the accumulation of denatured tracer material, which is less than 5% of the injected dose (figure 25) but is directed specifically at this organ and accumulates there. Next highest in LDL assimilation was liver, where there was comparatively higher hepatic LDL uptake in the tumour group than the control group (p<0.05,Wilcoxon Rank Test) - exactly the opposite of the situation in both mouse models studied (tables 5 and 7). Tumour uptake was surpassed only by liver and spleen (table 11).

C) PANCREATIC TUMOURS

As a whole, liver had considerably higher overall LDL uptake than any other tissues in these animals (table 12). When uptakes were expressed per gram of tissue, spleen had the highest value (figure 26), as observed in the rat model above. In tumour animals splenic uptake was suppressed (p<0.05,Wilcoxon Rank Test) as compared to control rats. In both cases the overall contribution of the spleen to LDL catabolism was relatively small. Adrenals in the tumour group had high LDL assimilation, whereas the activity in control animals was considerably smaller (p<0.05,WilcoxonRank Test). In this model a number of tissues were quite active in assimilation of LDL (figure 26) and tumour appears to be only moderately active.

	TOTAL LDL UPTAKE (CPM/ORGAN/DOSE) X 100					
TISSUE	CONTROL GROUP (n=18)	TUMOUR GROUP (n=22)				
Tumour		0.70 <u>+</u> 0.23				
Spleen	3.15 <u>+</u> 1.07	2.60 <u>+</u> 1.18				
Liver	1.00 <u>+</u> 0.19	1.24±0.20*				
Adrenal	0.60 <u>+</u> 0.28	0.62±0.17				
Ovary	0.33 <u>+</u> 0.24	0.51 <u>+</u> 0.28				
Kidney	0.43 <u>+</u> 0.07	0.43±0.07				
Thymus	0.23 <u>+</u> 0.09	. 0.40±0.09 [*]				
Gut	; _	0.38 0.22				
Lung	0.40 <u>+</u> 0.13	0.31±0.07				
Heart	0.19 <u>+</u> 0.09	0.18±0.04				
Skin	-	0.08±0.02				
Fat	0.09 <u>+</u> 0.07	0.06±0.06				
Muscle	-	0.05±0.04				
	·	,				

. •

Table 11 : Uptake of ¹²⁵I-cellobiose-tyramine-LDL by rat organs and dimethyl hydrazine-inducedcolonic tumours.

.



UPTAKE OF ¹²⁵ I-CT-LDL BY RAT TISSUES AND COLONIC TUMOURS EXPRESSED AS A

Figure 25:

	<u>TOTAL LI</u> (CPM/ORGAN/	DL UPTAKE /DOSE) X 100
TISSUE	CONTROL GROUP (n=5)	TUMOUR GROUP $(n=4)$.
Tumour	-	0.88 <u>+</u> 0.77
Liver	3.68 <u>+</u> 2.13	6.2 <u>+</u> 1.6
Spleen	0.11 <u>+</u> 0.05	0.26 <u>+</u> 0.04*
Kidneys	0.70 <u>+</u> 0.30	1.03±0.25
Adrenals	0.06 <u>+</u> 0.03	0.77 <u>+</u> 0.03 [*]
Heart	0.28 <u>+</u> 0.15	0.33±0.14
Lungs	0.69 <u>+</u> 0.26	0.72±0.07
Thymus	0.10±0.10	0.10±0.07
Gonads	0.11 <u>+</u> 0.02	0.11±0.01

Table 12 : Uptake of 125 I-cellobiose-tyramine-LDL

by Hamster organs and BOP-induced pancreatic tumours.

* Wilcoxon Rank Test vs. controls, p<0.05</pre>



TOTAL UPTAKE OF ¹²⁵ I-CT-LDL BY HAMSTER TISSUES AND PANCREATIC TUMOUR

Figure 26:

2.2.3. FEEDING EXPERIMENTS

In all of the preceding animal models, liver has been the major organ of receptor-mediated LDL catabolism, both as a whole organ and per gram of tissue. It is likely therefore that chemotherapy using LDL as a vehicle would be greatly damaging to this essential organ. In order to reduce the potential toxicity we have looked at a number of ways of minimising hepatic uptake of LDL during therapy (figure 27).

Also identified as organs at risk during "LDL chemotherapy" are the adrenal glands. Although they are very small and their overall contribution to LDL catabolism is relatively low, they rapidly utilise LDL in the synthesis of essential steroid hormones. As a result, adrenals have very high LDL receptor activity per gram, and are likely to be damaged during chemotherapy using LDL as a vehicle. Hydrocortisone has been used to reduce adrenal LDL uptake.

A) CHOLESTEROL FEEDING

NMRI mice with and without tumours were fed a diet containing 1% cholesterol (g/g) (2.1.13.) for 7 days prior to the experiment. Plasma cholesterol concentrations were measured at the time of sacrifice in groups of animals fed the normal diet or the higher cholesterol diet and these are shown in (table 6). Despite the increase in plasma cholesterol levels, a similar pattern of LDL catabolism was observed in both groups of animals. No significant suppression of receptor-mediated assimilation was apparent in any of the mouse organs or tumour (table 13).



Figure 27:

	RELATIVE RECEPTOR-MEDIATED LDL UPTAKE						
	(CPM/GRAM TISS	SUE/mL PLASMA)					
TISSUE	CONTROL GROUP (n=12)	TUMOUR GROUP (n=15)					
Liver	26 ⁺ 7	20 ⁺ 12					
Spleen	22 ± 8	17-7					
Kidney	53 ± 32	30-+20					
Heart	21 <i>±</i> 6	27-11					
Muscle	12 - 6	10 ⁺ 7					
Tumour	-	20-10					
·							

Table 13 :

Effects of cholesterol feeding on receptor-mediated LDL assimilation by the tissues of control and MAC-13 tumour inoculated NMRI mice¹⁵².

Both groups received a diet containing 1% cholesterol (W/W) for 7 days prior to the experiment. Plasma Cholesterol values for these animals are given in table 6.

B) BILE SALT SUPPLEMENTATION

It is known that bile acid feeding suppresses hepatic sterol synthesis and LDL receptor activity in the $dog^{195,196}$ and in the rat.¹⁹⁷ When sodium taurocholate was administered to NMRI mice (table 14) at a dose of 1% of the dietary intake (g/g) a fall incurred in receptor-mediated LDL uptake by the liver (p<0.05, Wilcoxon Rank Test). As expected this agent did not affect LDL uptake into other organs, including the adrenal gland.

C) HYDROCORTISONE THERAPY

In an attempt to inhibit adrenal uptake of LDL, hydrocortisone was administered daily by intraperitoneal injection at a dose of 5mg per day to NMRI mice for 7 days. This effectively reduced (p<0.05,Wilcoxon Rank Test), the receptor-mediated adrenal uptake of LDL (table 14) by ca. 50% in the control mice and by a similar amount in the tumour-inoculated group. In rats with colonic tumours, hydrocortisone reduced adrenal uptake by 40% in control animals and 30% in tumour bearing animals (table 15).

D). COMBINATION OF BILE SALTS AND HYDROCORTISONE

When sodium taurocholate feeding was combined with hydrocortisone injections in NMRI mice, adrenal receptor activity fell dramatically (by 80% in controls and 60% in tumour-bearing mice) table 14. In the rats, the fall in adrenal LDL receptor activity (table 15) remained the same whether hydrocortisone was administered alone or in conjunction with bile salts.

		RELATIV	<pre>'E RECEPTOR M' (cpm/gm/l</pre>	EDIATED TIS	SUE UPTAK a)	ы		
GROUP	LIVER	SPLEEN	KIDNEY	HEART	LUNG	MUSCLE	ADRENAL	TUMOUR
Control (N=4)	64-5	26 <mark>-</mark> 8	33-12	32+3	31-12	5-+3	66-16	I
Control + Na taurocholate (n=4)	48 - 9*	26 <mark>+</mark> 13	44+22	36+17	31-18	+- 6-4	65-4	I
Control + hydrocortisone (n=4)	91 + 2	43+11	52-16	47-18	53-20	11+2	34-16*	I
Control + Na taurocholate + hydrocortisone (n=4)	61 ⁺ 1	41-2	57-13	37±10	29-15	16 ⁺ 10	13 ⁺ 16*	I
Tumour (n=5)	53+8*	30+8	42+9	50-34	45-7	24-18	71-14	78-15
Tumour + Na taurocholate (n=5)	44-2*	27-7	34+19	23-5	35-15	7+3	41-22*	99 + 56
Tumour + hydrocortisone (n=5)	54-17	28 <mark>+</mark> 10	47-12	32+6	33+13	11-5	50-8*	84-25
Tumour + Na taurocholate + hydrocortisone (n=2)	34-2*	14-6	30+5	20-3	29-6 29-6	2+ 2+	29 - 3*	59-17

The effects of diet or steroid supplementation were assessed by the Wilcoxon Rank Test.

* Significantly different from untreated group, p < 0.05

Effects of Na taurocholate and hydrocortisone sodium $gy_Scinate$ on relative receptor mediated LDL uptake by tissues of control and MAC+13 inoculated NMRI mice. Table 14 :

		RE	LATIVE H	RECEPTOR-	MEDIATED	UPTAKE (DF LDL 1	3Y RAT 1	ISSUES A	ND COLON	IC TUMOU	RS	
						pm/g tis:	sue / m.	l plasme					
GROUP	LIVER	OVARY	LUNGS	ADRENAL	SPLEEN	KIDNEY	GUT	SKIN	THYMUS	HEART	MUSCLE	FAT	TUMOUR
Control (n=4)	53-10	53-12	49-6	48-12	46-4	38-12	35-11	28-15	27-9	20-5	5-4	3-0.6	1
<pre>Control + Indrocortisone (n=4)</pre>	43+1	16-2*	38+7	29+5	38+3 8- 8	28+1	21+2	12+2	3-1 * 3-1	14-1	4-0.7	2+0.2	I
Control + sodium taurocholate + Jydrocortisone	1 50 ⁺ 15	16-9*	46 ⁺ 11	39 - 12	54-14	48±12	39+11	14-1	14-5	21-6	3+1 3+1	2-0.5	1
(n=4) Tumour + 1ydrocortisone	31-2*	10+2*	34+1*	33-11	45-5	31-2* 31-2*	21-1	14-0.3	6 - 0.3*	17-1	3-0.1	2 ⁺ 0.2	34+6
<pre>(n=4) Tumour + sodium taurocholate + nydrocortisone (n=4)</pre>	43-43	1 8 1 4+ 3 4+	37 - 3*	29 - 1*	39-4 4	40+4	11-16	15-1	21-5	18+1	4-0.1	4-0.7	30-9 30-1

Wilcoxon rank test vs. controls, p<0.05.

*

Effects of sodium taurocholate and hydrocortisone sodium succinate on relative receptor-mediated uptake of LDL by rat tissues and dimethyl hydrazine-induced colonic tumours. •• Table 15

Assimilation of LDL by the tumour did not change significantly during either hydrocortisone or bile acid therapy, or as a result of their simultaneous administration. The following section describes the methods employed for incorporation of cytotoxic drugs into LDL, and assessing the cytotoxicity of the drug-LDL complexes so formed.

2.3.1. SYNTHESIS OF CHLORAMBUCIL CHOLESTERYL ESTER (CBE)

Molar equivalents of chlorambucil, cholesterol and dicyclohexylcarbodiimide and (when required) 1 mCi $\left[7^{-3}H\right]$ -cholesterol were dissolved in dry diethyl ether in a pre-dried round-bottomed flask. The flask was flushed with nitrogen, sealed and its contents stirred at room temperature for 24 hours. The progress of the reaction was monitored by thin-layer chromatography (TLC) using chloroform : ethyl acetate 8 : 1 as eluant, on silica gel F_{254} TLC plates. Chlorambucil remained near the origin ($R_{f} = 0.35$) while the ester migrated near the solvent front ($R_f = 0.92$) and cholesterol between the two ($R_f = 0.65$). On completion of the reaction, precipitate was removed by filtration and the solvent evaporated from the remaining solution. The residue was dissolved in chloroform, prior to removal of unreacted starting materials on silica and basic alumina columns. Elution was achieved using chloroform and carbon tetrachloride alternatively in 5 mL aliquots. The purified cholesteryl ester was dried, weighed and infrared and ¹H-NMR spectra obtained (figure 28). The infrared spectrum showed a characteristic peak for the ester carbonyl stretch at 1740 ${\rm cm}^{-1}$ and confirmed that the product was free of unreacted cholesterol (no H-O absorption). The reaction yield was typically 60-80%.



chlorambucil prepared as described in Section 2.3.1.

2.3.2. INCORPORATION OF CBE INTO LDL BY EXCHANGE FROM CHOLATE/LIPID

MICELLES

Initially we used an adaptation of Stoffel's method¹⁹⁰ for loading HDL with labelled cholesteryl esters. This procedure has been described in detail in Section 2.1.5. In order to improve the overall efficiency of drug incorporation, several other modifications were made to the method (figure 29).

A) "RECONSTITUTED" HDL

HDL (d1.063 - 1.21Kg/L) was prepared from human plasma as described in Section 2.1.2. An aliquot containing 23mg of HDL protein was extracted twice with two volumes of chloroform : methanol (3:1), washed with ether and lyophilised overnight. Phospholipid, cholesterol, and $[7-^3H]$ cholesterol chlorambucil ester were added to give the molar ratios of each found in HDL (80:24:53, lipid:protein,1:1,W/W) and the mixture sonicated in 5 mL 0.02M tris, 0.02M NaCl, 0.02M EDTA buffer, pH 8.2 for 15 minutes at 0°C. For comparison, an aliquot of HDL containing the same amount of HDL protein was loaded with $[^3H]$ -CBE by exchange from cholate micelles as outlined in Section 2.1.5., replacing I-cholesteryl ether with $[^3H]$ -CBE. Both $[^3H]$ -CBE preparations were purified by filtration through 0.22 µm millipore filters and centrifugation at 1.063 Kg/L. Fractions of 5 mL were removed from each centrifuge tube and assayed for CBE by $[^3H]$ -specific activity.



Figure 29:

The modified Stoffel method, section 2.3.2, used to load LDL with chlorambucil cholesteryl ester (CBE).

B) EXCHANGE OF CBE FROM CHOLATE MICELLES DIRECTLY TO LDL

5 mL of drug-laden cholate micelles, Section 2.1.5, was stirred overnight at room temperature with 5 mg human LDL (protein) and 15 ml d>1.21 Kg/L plasma fraction (prepared as described in Sections 2.1.1. and 2.1.2. respectively). Chlorambucil ester-laden LDL was purified as described for I-cholesteryl ether LDL (2.1.4.)

2.3.3. DEGRADATION OF CBE DURING INCUBATION WITH LDL

Mixtures contained 2mg of CBE in lmL of micelles, Section 21.1.5. and 4mL of either PBS, heat-inactivated¹⁹⁸ d >1.21 Kg/L plasma fraction (prepared as in section 2.1.2., heated for 1 hour at 60°C and filtered through a 0.45 μ m filter), or sodium iodoacetate-treated d >1.21 Kg/L plasma fraction (pre-incubated overnight with 5 mM NaIOAc and dialysed against PBS to remove free NaIOAc)¹⁹⁹. Each mixture was stirred gently at room temperature for 24 hours and samples were removed at 0,1,2,4 and 24 hours and assayed for CBE by HPLC (2.3.6.)

2.3.4. TRANSFER OF CBE FROM SYNTHETIC MICROEMULSIONS TO LDL

As an alternative to the Stoffel¹⁹⁰ method (2.3.2.) drug/phospholipid/ cholesteryl oleate microemulsions were made by the method of Craig et al 200 (figure 30) using the molar proportions of each component suggested by Parks et al²⁰¹. 2.4 mg each of dimyristoyl pholphatidyl choline (DMPC) and CBE were added to 2.37 mg of cholesteryl oleate, in chloroform, the solvent evaporated and the mixture dried under vacuum for at least 1 hour. The dried lipids were dissolved in 200 μ L isopropanol (predried over molecular sieves). Sixty-six µL of the lipid mixture were injected quickly into 1 mL of rapidly vortexing PBS, with the entire system kept at 46°C throughout the procedure. Several millilitres of microemulsions were prepared in this way and mixed with equal volumes of LDL (protein 2-6 mg/mL), prior to incubation for 5 hours at 37°C. Thereafter, solid KBr, 0.012 mg/mL, was added to the solution making its density 1.006 Kg/L. The preparation was overlayered with d1.006 solution in Beckman 40.3 centrifuge tubes and the drug-laden LDL separated from the microemulsions by overnight centrifugation a 45 K rpm and 25°C in Beckman 40.3 rotor.



ester(CBE)-LDL using drug-lipid microemulsions as described in Section 2.3.4. Microemulsions and density solution were aspirated and the infranatant CBE-LDL desalted by gel filtration on a PD10 column, eluting with PBS. The CBE-LDL was sterilised by filtration through a 0.22 μ m millipore filter. This treatment also removed aggregated LDL and microemulsions which were of density greater than 1.006 g/mL. The protein and drug concentrations of the CBE-LDL complex were measured by Lowry protein assay (2.1.9.) and HPLC (2.3.6.) respectively. Identical incubations were performed in the absence of LDL and centrifuged as described above for the LDL-containing preparation. The infranatant was then filtered through a 0.22 μ m millipore filter and assayed for CBE to ensure that no microemulsions remained after purification.

2.3.5. INCORPORATION OF ANTHRACYCLINES INTO LDL

A) Daunomycin was incorporated following the method of Iwanik et al²³ (figure 31). Drug (20-50 mg) was dissolved in methanol and dried onto the surface of a glass tube under N_2 , 5 mL of LDL (protein 2-10 mg/mL) in buffer (5mM sodium phophate, 0.15M NaCl, 0.3 mM EDTA, pH7.3) was added and the mixture flushed with nitrogen and stirred gently at room temperature for 2-3 hours in darkness. The daunomycin-LDL complex was separated from the unbound drug by gel filtration through PD10 columns, using PBS as eluant. LDL-containing fractions were pooled, filtered through a 0.22 μ m millipore filter and assayed for drug (2.3.6.) and protein (2.3.7.).

In procedures 2.3.5. B-G, LDL prepared by zonal (2.1.1.) was used without concentrating or de-salting in an attempt to reduce aggregation of the LDL during incubation with adriamycin.

PREPARATION OF DAUNOMYCIN-LDL



PBS: 5mM sodium phosphate, 0.15 M NaCl, 0.3mM EDTA, pH7.3

Figure 31: Protocol for the preparation of daunomycin-LDL as described in Section 2.3.

B) Cholate Method: As described for CBE (2.3.2.), adriamycin, phospholipid, cholesterol, cholate micelles were prepared using 5 mg adriamycin, 10.9 mg lecithin and 1.6 mg cholesterol per 5mL of cholic acid buffer.
To each mL of micelles was added 1 mL of LDL (0.5 mg protein/mL) and the solution stirred overnight at room temperature.

C) DMSO Method: 2 mg adriamycin was dissolved in 2 mL of DMSO buffer 202 (0.15 M NaCl, 0.3 mM EDTA, pH 7.2 containing10% DMSO,v/v) and warmed to 37°C for 10 minutes before adding 2 mL LDL solution (protein 0.1-0.5 mg/mL). The mixture was incubated overnight at 37°C, in darkness.

D) PBS Method: Adriamycin (2 mg) was dissolved in 2 mL buffer (5 mM sodium phosphate, 0.15 M NaCl, 0.3 mM EDTA pH 7.3) and 2 mL LDL solution added (protein 0.1-0.5 mg/mL). The solution was stirred in darkness at room temperature for 2-3 hours.

E) Dialysis : LDL (0.1 mg/mL) in dialysis tubing, molecular weight cut-off 2,000-8,000 was surrounded by various concentrations of adriamycin (2-150 μ g/mL) in DMSO buffer. The concentrations of free and bound adriamycin in the dialysis bags were measured at intervals by HPLC (2.3.6.). Free adriamycin was removed by chromatography on PD10 columns, with PBS as eluant.

F) Incubation in Glycine Buffer: Adriamycin-LDL was prepared by overnight incubation of 5 mg of drug with 2 mL LDL (0.5 mg protein/mL), and 5 mL NaCl/DMSO buffer, pH 7.2 or 5 mL 1M Glycine/DMSO buffer, pH 10. The complexes were concentrated by centrifugation at 38 K for 4 hours and at 10°C in Beckman SW41 rotor, after making the density of the solutions up to 1.4 g/mL by the addition of solid KBr. The drug-LDL complexes were desalted on PD10 columns using PBS as eluant. G) Surfactants: Following the method of Tucker et al 203 surfactants were used to enlarge the LDL particles. Ten per cent solutions (w/v) of Brij-35, Triton x-100 and SDS were prepared in PBS and filtered through 0.22 µm filters prior to use. The surfactants were each added gradually to a different aliquot of LDL (protein 2 mg/mL in PBS), giving final surfactant to phosholipid molar ratios of 1:1. The mixtures were left to react for 30 minutes before adding two volumes of adriamycin solution (1mg/mL in PBS). After 24 hours incubation at room temperature, surfactant and free drug were removed by PD10 filtration.

In all of the above procedures, anthracyclines were assayed by HPLC (2.3.6.) and protein determined as described (2.3.7.)

2.3.6. HPLC ASSAYS

. . .

Chlorambucil²⁰², chlorambucil cholesteryl ester, adriamycin²⁰³ and daunomycin were measured by reversed-phase high pressure liquid chromatography (HPLC). A 20 x 0.5 cm stainless steel column, slurry packed in the laboratory with 5 μ m Hypersil-ODS, was used in all assays. Typical chromatograms and standard curves for each system are shown in figures 32 and 33.

HPLC OF ANTHRACYCLINES



Figure 32: HPLC profiles of adriamycin and daunomycin



Figure 33:

HPLC profiles of chlorambucil and chlorambucil- \propto -cholesteryl ester obtained using the system outlined in Section 2.3.6.(C).

A) ADRIAMYCIN ASSAY²⁰³

APPARATUS:

INJECTION VALVE	:	Ryeodyne 7125 with 20 µL loop.
PUMP	:	Varian 8500
DETECTOR	:	Pye-Unicam LC-FL Fluorescence monitor, fitted
		with interference filters for excitation at 450 nm
		and emission at 550nm.
RECORDER	:	Servoscribe single pen chart recorder.
MOBILE PHASE	:	Acetonitrile: 10 mM Phosphoric acid (65:35)
		containing 6 mM Brij-35.
FLOW RATE	:	2 mL/minute.

EXTRACTION PROCEDURE: To 1 mL of solution was added 200 ng of daunomycin (as internal standard), in methylene chloride and the solution mixed with 1 mL 10 mM phosphate-buffered saline, pH7.8, followed by extraction with 2 x 5 mL methylene chloride. After centrifugation for 5 minutes at 2000 rpm, the organic layer was removed and evaporated to dryness at 40°C under a stream of nitrogen. The residue was dissolved in 100 μ L of mobile phase and 20 μ L injected onto the column.

STANDARDS: A standard solution of 10 μ g adriamycin /mL was used to spike LDL solution, giving final drug concentrations of 10 ng-1 μ g/mL. Extraction of standards was carried out as described for samples and a standard curve plotted of adriamycin: daunomycin peak height <u>vs</u> adriamycin concentration (figure 34).



Each point is the mean of four assays.

Daunomycin was measured either on the system used for adriamycin or on the following system:

APPARATUS:

INJECTION VALVE	:	Rheodyne 7125 fitted with 100 μL loop
PUMP	:	Gilson 300
DETECTOR	:	Gilson Holochrome U.V. Monitor, absorption
		measured at 330 nm.
RECORDER/INTEGRATOR	:	Schimadzu
MOBILE PHASE	:	Acetonitrile : 10 mM phosphoric acid (68:32)
		containing 6 mM Brij-35.
FLOW RATE	:	2 mL/minute

EXTRACTION PROCEDURE AND STANDARDS: Daunomycin was extracted from samples and standards as described for adriamycin (2.3.6 A) omitting the addition of 200ng of daunomycin. A standard curve was constructed by plotting daunomycin peak area against concentration (fig.35).

C) CHLORAMBUCIL AND CHLORAMBUCIL CHOLESTERYL ESTER ASSAY²⁰²

APPARATUS: as described for the daunomycin assay, with the following amendments:

DETECTION	:	254 nm
MOBILE PHASE	:	Methanol:water (65:35) containing
		10mM phosphoric acid
FLOW RATE	:	1mL/minute


EXTRACTION PROCEDURE : Each mL of plasma or LDL solution was extracted with 2 volumes of chloroform and centrifuged, twice. The organic phase was aspirated and dried down under a stream of nitrogen at 40°C. The extract was dissolved in methanol and injected onto the column. STANDARDS: Chlorambucil cholesteryl ester, purified by TLC, or chlorambucil was dissolved in chloroform and spiked into LDL solutions and extracted by the same procedure as the samples. Peak areas were plotted against drug concentration (figure 36).

2.3.7. LDL PROTEIN DETERMINATION IN SOLUTIONS CONTAINING ANTHRACYCLINES

The apoprotein content of the various lipoproteins prepared was assayed by the method of Lowry et al¹⁸⁸, as described in section 2.1.9. In the case of LDL containing anthracyclines, protein concentration was observed to be much higher than expected (figure 37). The Lowry assay conditions are basic and this changes the colour of the anthracyclines from orange to purple. A scan of adriamycin absorption in the ultraviolet and visible regions (figure 38) showed however that the high pH was not causing the "extra" absorbance at 750 nm. It is likely that the Lowry reagents form a complex with these drugs in much the same way as they do with peptides, so giving rise to an enhancement of the blue colour of the solution and consequently elevated estimates of protein concentration. Apoprotein B was therefore precipitated from solution using 1 volume of 1,1,3,3-tetramethyl urea, delipidated by extraction with 2 volumes of chloroform : methanol (2:1) and dried with ether before solvation in 0.5M NaOH and Lowry protein assay.

92





Each point is the mean of four assays

*



Figure 37: Apparent protein concentration in LDL solutions of constant LDL content and increasing adriamycin concentration.

U.V./VISIBLE ABSORPTION OF ADRIAMYCIN IN NEUTRAL AND BASIC CONDITIONS



PBS: 5 mM sodium phosphate, 0.15 M NaCl, 0.3 mM EDTA, pH 7.3

Figure 38: Absorption spectra of adriamycin in neutral and basic conditions. Note that there is no enhancement of the drug's absorption at 750nm on changing pH.

 \star

¹³¹I- or ¹²⁵I- labelled LDL or drug were administered to New Zealand White rabbits by injection into a canulated marginal ear vein. Blood samples were taken after 5 minutes and at intervals over the following 24 hours. Isotope dilution in the 5-minute sample gave an indication of plasma volume. Plasma clearance of the radioactive species was directly related to the disappearance of radioactivity with time. Re-isolation of the labelled LDL was unnecessary since non-protein bound iodine has been reported to correspond to less than 1% of the total radioactivity and apoprotein B does not transfer to the other plasma lipoproteins²⁰⁶. Iodide was administered, 0.1g/L in drinking water, for two days prior to and throughout these studies. Radioactivity remaining in the plasma at each time point was expressed as a percentage of the level of radioactivity in the 5-minute sample.

2.3.9. CELL CULTURE MEDIA

HeLa cells were maintained in Ham's F10 medium containing 20mM HEPES and supplemented with 10% foetal bovine serum, 2mM glutamine, 5% CO₂ and 50 units/mL of penicillin and streptomycin. For culture of human glioma(G-CCM) and human lung carcinoma(L-DAN), medium containing 50% Ham's F10 and 50% Dulbecco's Modified Eagle's media supplemented with serum etc., as above, was used. Tumour samples were collected in balanced salt solution with antibiotics as above prior to dissection.

2.3.10 DETACHMENT OF CELLS FROM CULTURE FLASKS

Cells were detached from culture flasks using 0.25 % trypsin in balanced salt solution(BSS). After removing medium, cells were washed

with BSS and soaked in trypsin for one minute. The trypsin solution was aspirated and the flasks sealed and incubated at 37°C until the cells had detached from the surface of the flask. This usually required 10-15 minutes, after which the cells were re-suspended in medium prior to counting cell density using a Coulter Counter or haemocytometer.

2.3.11 PREPARATION OF LIPOPROTEIN-DEFICIENT SERUM

Fresh human blood was obtained from healthy volunteers and allowed to clot in 10mL glass tubes, and the serum was aspirated following centrifugation at 2500 rpm for 15 minutes. The density of the serum was adjusted to 1.25 Kg/L by the addition of 0.371g KBr/mL. This solution was overlayered with d1.25 density solution and centrifuged at 45000 rpm for 24 hours in a Beckman Ti60 rotor at 10°C. Lipoproteins and density solution were aspirated and the lipoprotein-deficient serum dialysed, first briefly against distilled water, and then against PBS, pH 7.4.

2.3.12 MICROTITRE DRUG SENSITIVITY ASSAY

Freshney and co-workers have established microtitre assays for testing cytotoxic drugs against human glioma²⁰⁷ and HeLa²⁰⁸ cell lines. This procedure has a number of advantages over the commomly used cloning experiments²⁰⁹ for assessing cytotoxicity : large numbers of replicates can easily be handled, giving better reproducibility; and the long drug-exposure times enable all of the cells in culture to pass through the cell cycle at least once, thus excluding the cycle-dependence of some drugs as a variant in the assay. Protein synthesis was identified as a fundamental metabolic process without which the cells could not

survive. Since leucine is an essential amino acid it is unlikely that protein synthesis could carry on for long within the cell without the incorporation of leucine obtained from the surrounding medium. Thus, the failure of cells to incorporate $[^{3}H]$ -leucine into protein several days after drug exposure will almost certainly correlate with cell death.

Protocols for HeLa and Glioma microtitre assays are shown in figure 39. Cells were seeded at a density of 5 X 10^3 cells/mL, i.e. 10^3 cells/well, in the central 60 wells of 96-well microtitre plates. The edge wells were filled with cell-free medium. Throughout the experiment the cells were incubated at 37°C in a humidified 5% $\rm CO_2$ atmosphere. After being allowed to adhere to the surface for 3 days the cells were re-fed with fresh medium. Medium and drugs were administered on day 5 in the case of HeLa cells, and days 4, 5 and 6 for glioma cells. Drugs were added to the plate wells in 10-50 μL of PBS in serial dilutions of 1:10 or 1:5, with duplicate incubations at each concentration. Twelve wells of cells in each plate were left free of drugs to serve as controls. Following drug exposure the medium was replaced and the cells allowed to recover for 4 days(HeLa) or 5 days (glioma) before $[4, 5-^{3}H]$ -leucine (1 µCi/mL) was added in fresh medium. During this recovery period medium was replaced at least every 48 hours. The cells were exposed to $\begin{bmatrix} 3 \\ 4 \end{bmatrix}$ -leucine for 24 hours prior to washing with PBS, extraction with 10% (w/v) trichloroacetic acid (2 X 15 minutes) rinsing with distilled water, fixing with methanol and drying. The contents of each well were dissolved in 200µL of 0.3M NaOH containing SDS (1% w/v). An aliquot from each well was placed in a 1mL Sarstedt tube, neutralised and liquid scintillant added. Each tube was well mixed and counted for $\begin{bmatrix} {}^{3}H \end{bmatrix}$ -activity in a Packard β -Counter. The counts obtained for each well were corrected for background and expressed

m	plates, 10 ³ cells per well					and add drugs			Recovery Period		vith medium containing • 1 NCi/ML	ot, fix and dry, dissolve cell		rporation by scintillation counting			
lela MICROTITRE ASSAY ²⁰⁶	AY 0 Seed microtitre	1	cJ	3 Replace medium	4	5 Replace medium	6 Replace medium	7	8 Replace medium	თ	10 Replace medium v [<u>4</u> ,5- ³ H -1encine	11 Wash, TCA extract	residue in NaOH	12 Measure ³ H-inco			
MICROTITRE ASSAY ²⁰⁷	Seed microtitre plates, 10^3 cells per well			Replace medium		Replace medium	• add drugs	Replace medium		Replace medium Recovery Period			Replace medium with medium containing	4,5- ³ H -leucine, 1 µCi/mL	Wash, TCA extract, fix and dry, dissolve cell residue in NaOH.	Measure ³ H-incorporation by scintillation counting	
GLIOMA	DAY O	Ч	N	с	4	S	9	7	ω	თ	10	11	12		13	14	

٠

Protocols for HeLa and Glioma Microtitre Drug Sensitivity Assays Figure 39:

TCA : trichloroacetic acid

as disintegrations per minute (DPM). The mean DPM for controls was taken to be 100% survival and the $\begin{bmatrix} {}^{3} \text{ H} \end{bmatrix}$ -leucine uptakes for other wells expressed as a percentage of this value. A plot of log(% cell survival) against log(drug concentration(M)) was constructed for each of the drugs and drug-LDL complexes studied. From these curves, the doses which corresponded to 50% inhibition (ID₅₀) and 90% inhibition(ID₉₀) of cell growth, compared to controls, were calculated for each drug.

An extra plate of cells was seeded on day 0 for use in monitoring the cell growth rate. At intervals during the experiment several wells were trypsinised as described in Section 2.3.10 and the number of cells per mL measured using a haemocytometer. This was to verify that the majority of cells was in the logarithmic growth phase throughout the assay. This is essential if the maximum relative changes in $[^{3}H]$ -leucine uptake between control and drug-treated cells are to be assessed. Biased results would be obtained if $[^{3}H]$ -leucine incorporation was measured while controls had reached stationary phase and drug-treated cells were in log. phase. Growth curves for G-CCM and HeLa cells plotted from data obtained in this way are shown in figure 40.

2.3.13 L-DAN CLONOGENIC STEM CELL ASSAY AND SPHERIOD MODEL SYSTEM

This work was carried out with the collaboration of Dr. D.J. Kerr of the Department of Medical Oncology, Glasgow University. L-Dan cells were maintained in exponential growth in monolayer culture prior to disaggregation using trypsin (Section 2.3.10). Thereafter cells were seeded for clonogenic assay²⁰⁹ or used for initiation of tumour spheroids²¹⁰ (fig. 42).

96

MICROTITRE ASSAY GROWTH CURVES



Figure 40:

Growth curves for HeLa and glioma (G-CCM) cells in the microtitre assays (Section 2.3.12). Cell number was measured using a haemocytometer and each point is the mean of duplicate counts for a pool of the cells from twelve wells of the culture plate.

CLONOGENIC ASSAY

cell monolayers in exponential growth

Drug Exposure 1 hr

harvest with 0.25% Trypsin in BSS

wash in ice-cold medium

seed at 200 cells/mL in 5cm petri dishes

Incubate 37°C 2%CO₂ 12 days

wash & fix

count colonies > 40 cells

Figure 41: Protocol for the clonogenic stem cell assay used in the testing of free and LDL-encapsulated daunomycin against L-DAN lung tumour cells, as described in Section 2.3.13.





Figure 42:

Outline of the experiments performed with native and LDL-encapsulated daunomycin on lung tumour (L-DAN) cell spheroids as described in Section 2.3.13. Monolayers were exposed to varying concentrations of daunomycin and daunomycin-LDL for one hour. Following treatment, the monolayers were harvested with trypsin, as described in Section 2.3.10, centrifuged, and washed with ice-cold medium, after which the cells were diluted in medium and seeded at 200 cells/mL in 5cm Petri dishes. The dishes were incubated for 12 days at 37°C in a 2% CO₂ atmosphere before fixing and staining of colonies with methylene blue. Colonies of more than 40 cells were counted. The cloning efficiency of the untreated cells was defined as 100% survival and the cloning efficiency of treated cells expressed as a percentage of control survival.

Spheroids were prepared²¹⁰, with a mean diameter of 350 μ m, at 200-300 spheroids/glass universal tube and subjected to similar drug concentrations and exposure times as in monolayer, at 37°C with intermittent agitation. The spheroids were then allowed to sediment and washed with ice-cold medium, re-suspended in medium and subdivided for further analysis.

One aliquot of spheroids was incubated with 0.125% trypsin in PBS for 14 minutes at 37°C, after which the trypsin solution was replaced with medium. Disaggregation of the spheroids was acheived by repeated pipetting of the suspension. The clonogenic assay was performed as described above. A second aliquot of spheroids was used for a growth delay experiment. Intact spheriods were transferred to agar-coated wells on a plastic tissue culture multidish, 1 spheroid per well, using a pasteur pipette. The cross-sectional areas of 24 spheroids from each treatment group were measured twice weekly using a "micromeasurements" image analysis system coupled via a television camera to an inverted optical microscope²¹¹. These measurements were converted to volumes, assuming spherical geometry. Treatment-induced growth delay was defined as the time taken for median spheroid volumes to increase by a factor of 10 above initial size.

2.3.14 THE SUBRENAL CAPSULAR (SRC) ASSAY

A six-day assay has been described by Bogden et al²¹² for assessing the sensitivity of primary human tumour explants to chemotherapeutic agents in normal immunocompetent mice. Following this protocol (fig. 43), with the cooperation of Dr. D. Cunningham, Medical Oncology, G.R.I., the anti-tumour effects of some drug-LDL complexes have been tested.

Tumour samples were removed from patients or animals and immediately placed in sterile balanced salt solution containing penicillin and streptomycin, 50 units/mL. The tumour was cut into fragments measuring approximately 1mm³. The mice (NIC B6D2FI) were weighed and anaesthetised and a lateral incision was made over the left kidney, which was then exteriorised. Tumour was inserted under the renal membrane using a trocar, after a small incision had been made in the membrane with a Ziegler iridectomy knife. This procedure was carried out under magnification with a Zeiss stereoscopic microscope. Two perpendicular diameters of the tumour fragment were measured in situ using a micrometer scale, fitted to the microscope and calibrated so that 10 micrometer units were equivalent to 1mm. Kidney was gently replaced and the incision closed using 9mm autoclips.

Six animals were implanted with tumour in each treatment group. One group of six mice was left untreated to act as controls, and the rest received drugs by tail vein injection on the following day, i.e. day 1, and on days 3 and 5. On day 6 the animals were sacrificed and the tumour dimensions remeasured, before the kidneys were placed in

SRC ASSAY

Measure 2 perpendicular diameters of tumour in situ Implant 1 mm³ of tumour under renal capsule Inject drugs or LDL via a tail vein 0 DAY 3 ŝ DAY 6 DAY ДАΥ ДАΥ

Sacrifice mice , re-measure tumour diameters Calculate tumour volume : $0.5(L \times w^2)$ Express as % of initial volume formol saline, pending histological analysis. The change in tumour volume, $\frac{1}{2}(\text{length X width}^2)$, was expressed as a percentage of the original volume.

2.3.15 MEASUREMENT OF PLASMA WHITE BLOOD CELL DENSITY

Blood was removed from the mice used in the SRC assay, on day 6, via the vena cava, and placed in glass tubes containing K_2^{EDTA} . White blood cell density was measured at the Department of Haematology, G.R.I. on an automated Coulter Counter system.

2.3.16 HISTOLOGICAL ANALYSIS OF TUMOUR SECTIONS

Three micron thick sections from mouse kidney and implanted tumour were stained with haematoxylin and eosin and examined by Dr. A. Jack of the Pathology Department, G.R.I.

2.3.17 LDL RECEPTOR ACTIVITY ASSAY IN VITRO

High-affinity LDL degradation was measured as the difference in cellular degradation of $^{125}I-LDL$ in the absence and presence of an excess of unlabelled LDL (fig. 44)²¹³. Twenty-four-well culture plates were seeded on day 0 with cells in medium containing foetal bovine serum, Section 2.3.9, and incubated in a 5% CO₂ atmosphere at 37°C. Two wells of each plate were filled with medium only. On day 3, the medium was replaced with medium containing 10% lipoprotein-deficient serum, prepared as described in Section 2.3.11, instead of foetal bovine serum. Fresh lipoprotein-deficient medium was added to the cultures on day 6 and $^{125}I-LDL$ was added at concentrations of 0-100 µg LDL protein/mL with duplicates of each concentration. . Identical incubations were set up

LDL DEGRADATION ASSAY



Figure 44: Scheme for assessing LDL degradation and internalisation in monolayer cell cultures.

and excess LDL added to a concentration of 0.5 mg LDL protein/mL in the culture medium. After a further 24 hours incubation, the medium was removed from the cells and its degraded LDL content measured. Cellfree incubations contained 125 I-LDL at 100 µg LDL protein/mL and were subsequently analysed for spontaneous degradation of the tracer. This amounted to less than 3% of the total tracer used and was subtracted from the values obtained in other incubations. High molecular weight, presumably intact, LDL was precipitated by adding 1,1,1-trichloroacetic acid to the medium to a final concentration of 10% (w/v). The supernate was separated from the pellet by centrifugation and free iodide extracted from it by the addition of excess potassium iodide (10 µL, 40% solution) and 10 µL of 30% w/v hydrogen peroxide, followed immediately by 2mL chloroform. After thorough mixing and centrifuga-last tion, the aqueous phase, containing degraded LDL, was separated and assayed for ¹²⁵I-activity in a Packard Autogamma Twin-Channel Scintillation Counter.

The cells remained attached to the surface of the culture wells and, after washing with PBS, extraction with 10% trichloroacetic acid (2 x 15 minutes), rinsing with distilled water, and fixing with methanol, the cell residue was dissolved in 1.0 M NaOH. Aliquots of the dissolved cellular material were assayed for protein by the Lowry method, Section 2.1.9, and for ¹²⁵I-activity using a Packard Scintillation Counter as above. This procedure isolated intact intracellular lipoprotein and the amount of LDL internalised was expressed per microgram of cell protein for each incubation.

100

2.4.1. CHLORAMBUCIL CHOLESTERYL ESTER-LDL

Chlorambucil cholesteryl ester (CBE) was loaded into LDL by the modified Stoffel method¹⁹⁰ described in Section 2.3.5. Approximately 50% of the drug-ester (as measured by specific $\begin{bmatrix} 3 \\ H \end{bmatrix}$ activity) was transferred at each stage (table 16), producing LDL which had approximately 10% of its cholesteryl ester core replaced with CBE.

In an effort to increase the net amount of drug incorporated into HDL and subsequently into LDL, reconstituted HDL containing increased amounts of CBE, was generated as described in Section 2.3.2 (A). The percentages of total drug-ester found in each fraction after centrifugation at 1.063 g/mL are shown in fig.45. The HDL formed by the reconstitution procedure was of a wide density range and would prove difficult to separate from LDL after the next step in the procedure. For this reason, reconstituted HDL was not used in any subsequent preparations.

The CBE-LDL complex was tested on human glioma cells in the microtitre assay outlined in Section 2.3.12. Results (fig.46) indicated that the drug-LDL complex was inactive. It is likely that the drug itself had become inactive during the incorporation procedure, since the G-CCM cells had already been reported²⁰ to express high LDL receptor activity under similar conditions. Possibly tris (tris(hydroxymethyl)_ methylamine), a nucleophile in the cholate sonication buffer, Section 2.3.2., was attacking the carbon attached to the terminal chlorine of the drug and rendering it inactive. Replacing tris with phosphate in the sonication buffer appeared to reduce the inactivation of the drug PREPARATION OF CHLORAMBUCIL-LDL BY STOFFEL'S METHOD AND OTHERS

mg/mg nmol/nmol PROTEIN RATIO CBE TO LDL 143 63 184 547 181 0.19 р 0.25 0.08 0.24 0.73 Protein concentration was calculated assuming a molecular weight of 500,000 Daltons for the apo mg/mL nmol/mL CONCENTRATION LDL PROTEIN* 0.0 4. 0 4.6 1.6 .10.0 APPARENT 5.0 3.0 2.0 2.3 8.0 CONCENTRATION OF nmol/mL 1813 847 876 252 861 CBE IN LDL mg/mL 0.58 0.17 1.22 0.59 0.57 % CBE IN HDL LDL 29 4 ω 10 15 15 61 40 36 I Cholate → HDL → LDL (Tris Buffer) Buffer) Buffer) content of each LDL particle. = : (PBS ▶LDL (PBS = = = -= Ξ Ξ = Cholate METHOD = = = *

Apparent concentration of Chlorambucil Cholesteryl Ester (CBE) in LDL and HDL at various stages of the preparation of CBE-LDL by the methods described in section 2.3.2. •• Table 16



native and "reconstituted" lipoprotein.





Figure 46: Comparative cytotoxicities of LDL, chlorambucil, and chlorambucil ester-LDL (prepared by the method used to prepare radioiodinated cholesteryl ether-LDL, fig. 15) on human glioma cells in the microtitre assay. Each point is the mean of duplicate incubations.

since the drug-LDL complex prepared in this way had cytotoxic activity (fig. 47) against human glioma cells in the microtitre assay, Section 2.3.12.) However, the buffer change also reduced the overall efficiency of the LDL loading procedure by a factor of 2 (table 16). Thus the 0.02M phosphate buffered saline, 0.02M EDTA, pH 8.2 was used in all of the following CBE-LDL preparations by Stoffel's method. In order to increase the net amount of drug recovered in LDL, the exchange step from cholate micelles to HDL was omitted, that is, the drug-laden micelles were incubated immediately with LDL and d>1.21 Kg/L bottom fraction, Section 2.3.2. This increased the overall incorporation of the drug-ester 3-fold (Table 16), with no obvious effects on the integrity of LDL. CBE-LDL prepared in this way was not obviously different from either native-LDL or CBE-LDL prepared by transfer of drug to LDL from cholate micelles via HDL. Plasma decays (figs. 48(a) and (b)) for each of the aforementioned LDL species made using radioiodinated LDL, Section 2.1.3, were measured in rabbits as described in Section 2.3.8. and simultaneously compared with the plasma decay of native LDL.

2.4.2. <u>COMPARISON OF CBE CONCENTRATION MEASURED BY SPECIFIC</u> [³ H] -ACTIVITY AND HPLC.

Until equipment was available for the assay of chlorambucil by HPLC, the drug-ester was synthesised with trace amounts of $\begin{bmatrix} 7-^{3} H \end{bmatrix}$ - cholesterol and "apparent" drug concentration was measured by specific $\begin{bmatrix} ^{3} H \end{bmatrix}$ -activity. This approach assumed that the drug-ester link was not being hydrolysed during the incorporation procedure - an assumption which later proved to be false.

Chlorambucil-ester-laden LDL was prepared as described in Section 2.3.2.

GLIOMA MICROTITRE ASSAY



Figure 47: Comparative cytotoxicities of chlorambucil and chlorambucil ester-LDL (prepared by the method outlined in fig. 15, using phosphate bufferedinstead of tris buffered- saline. Each point is the mean of duplicate incubations.



Figure 48: Plasma decays of native-LDL and chlorambucilester(CBE)-LDL prepared by exchange of drug from a) micelles to LDL via HDL ('HDL'-CBE-LDL) or b) micelles to LDL ('cholate'-CBE-LDL)

and aliquots of the solution were assayed for CBE by $\begin{bmatrix} 3 & H \end{bmatrix}$ -activity and HPLC and found to contain 0.72 mg/mL and 2.5 µg/mL by the respective methods. Lowry protein assay showed the solutions to have a protein concentration of 0.56 mg/mL. Thus only 3 molecules of drug had been incorporated per LDL particle, not 950 as calculated by specific activity. The CBE experiments already reported (figs. 45-48, table 16) then gave overestimates of the drug-ester incorporation into lipoproteins and consequently the results of these studies may be interpreted only qualitatively.

In all subsequent experiments with CBE, the drug-ester concentration was measured by HPLC, Section 2.3.6.

2.4.3. DEGRADATION OF CBE DURING INCUBATION PROCEDURES

Measurement of the CBE concentration in the cholate micelle mixture during the course of its incubation with LDL, showed that a large proportion of the ester had been degraded, leaving only 20-40% of the original amount of ester in the solution after 24 hours.

The d>1.21 Kg/L fraction of human plasma, Section 2.1.2., was used in the incubation as a crude preparation of cholesteryl-ester transfer protein, a plasma constituent which mediates the transfer of cholesteryl esters between LDL and other lipoproteins. Since this fraction contains numerous other enzymes, including LCAT, it was thought possible that the drug-ester was being hydrolised enzymically. Sodium iodoacetate (NaIOAc), an inhibitor of LCAT ¹⁹⁹ was added to the LDL/cholate/d>1.21 Kg/L fraction mixture in an attempt to reduce the hydrolysis rate, Section 2.1.3. Another approach to reduce enzyme degradation of the compound was to inactivate a number of enzymes by

103

heating the d>1.21 Kg/L plasma fraction to 60° C, which leaves cholesteryl ester transfer protein unaffected¹⁹⁸.

The degradation of the drug-ester with time under each of the conditions used is shown in fig. 49. None of the agents gave any increase in the amount of intact drug-ester remaining after 24 hours, on the contrary, they increased its degradation. It is interesting to note that there was no concurrent increase in the concentration of the free chlorambucil in any of the incubations, suggesting that the drug-ester degradation was not simply a hydrolysis procedure.

If therapeutic amounts of CBE are to be incorporated into LDL, other methods of incorporation must be considered, since the cholate method gives a final concentration of only a few micrograms of CBE per mg of LDL protein.

2.4.4. TRANSFER OF CBE FROM SYNTHETIC MICROEMULSIONS TO LDL

CBE-microemulsions were prepared as outlined in Section 2.3.4., using either DMPC or DPPC as the phospholipid component. Using DMPC emulsions, the final amount of drug in LDL was slightly higher, ... DMPC was used in all subsequent preparations of CBE-containing microemulsions. Recovery of the drug-ester at each stage of the preparation of CBE-LDL using these particles is shown in table 17. Even when contained within microemulsions, and in the absence of plasma proteins, 50% of the drug-ester is degraded during overnight incubation, table 17.

Typical recoveries of drug and LDL are shown, for each stage of the microemulsion/LDL preparation, in table 18. Less than 15% of the initial drug used is recovered in LDL but this is 100-times more than



DEGRADATION OF CBE UNDER VARIOUS CONDITIONS Figure 49:

sodium iodoacetate, ¥ plasma fraction

STAGE OF PREPARATION

DMPC MICROEMULSIONS

Incubation	Mixture	(t= 0 hrs.)	100
**	11	(t= 4 hrs.)	60
11		(t=24 hrs.)	48.6
Microemuls	ions only	(t=24 hrs.)	49
Chlorambuc: (after cen	il-Ester-1 trifugatio	LDL on)	48

DPPC MICROEMULSIONS

Incubation	Mixture	(t= 0 hrs.)	100	
**	11	(t= 4 hrs.)	59.5	
11	11	(t=24 hrs.)	51.9	
Microemuls	ions only	(t=24 hrs.)	49	
Chlorambuc (after cen	il-Ester- trifugati	LDL on)	40.8	

Table 17 : Comparison of incorporation of chlorambucil cholesteryl ester into LDL by exchange from microemulsions containing drug, cholesterol oleate and Dimyristoyl Phosphatidyl Choline (DMPC) <u>or</u> Dipalmitoyl Phosphatidyl Choline (DPPC),

STAGE OF PREPARATION	% CBE REMAINING	% LDL PROTEIN REMAINING
Incubation mixture (t= 0 hrs.)	100	100
Incubation mixture (t= 5 hrs.)	78	100
CBE-LDL (after centrifugation)	44	75
CBE-LDL (De-salted)	35	62
CBE-LDL (after 0.22um filtration)	28	55

Table 18 : Recoveries of chlorambucil cholesteryl ester (CBE) and LDL protein at each stage of the preparation of CBE-LDL as described in section 2.3.4. Values shown are the mean of 3 preparations. the drug recovery from the cholatc procedure.

The proportions of drug and LDL protein in several preparations of CBE/LDL by this method are variable, as shown in table 19. On average, 100 drug molecules (92^+63) are incorporated per LDL particle.

2.4.5. DEGRADATION OF CBE IN LDL DURING STORAGE AT 4^OC

Aliquots of CBE-LDL (135 μ g CBE/mg LDL protein) were stored at 4°C. Every 24 hours for 3 days, an aliquot was removed and assayed for CBE by HPLC, Section 2.3.6. The decay profile for the drug-ester under these conditions is shown in figure 50. In the first 24 hours, the ester decayed rapidly, leaving only 35% of the original ester in solution. Thereafter, a further 5% of the drug is lost over the following 48 hours. The absence of free chlorambucil in the solution at any time implies that the drug-ester is not being hydrolysed to yield free drug.

2.4.6. ADRIAMYCIN-LDL

Several methods of incorporating adriamycin into LDL were employed, Section 2.1.5. As reported by Iwanik et al²³, incubation of LDL with Adriamycin in sodium chloride/phosphate buffer causes immediate aggregation of LDL and drug. In an attempt to reduce this precipitation we used LDL solution immediately from the zonal preparation, Section 2.1.1., without concentrating it or removing any of the sodium bromide. This did not completely prevent aggregation in any of the buffers used but about 10% of the LDL and some drug remained complexed and in solution, and was subsequently purified and assayed for drug, Section 2.3.6, and protein, Section 2.3.7. Proportions of drug and

CHLORAMB CONCEN	UCIL ESTER TRATION	LDL P CONCEN	ROTEIN [*] TRATION	DRUG TO	PROTEIN TIO
(µg/mL)	(nmol/mL)	(mg/mL)	(nmol/mL)	(gram/gram)	(nmol/nmol)
	·		····		
181	269	5.5	11.0	0.033	24
752.	1120	7.8	15.6	0.096	72
530	788	4.6	9.2	0.115	86
221	328	1.7	3.4	0.130	97
1130	1679	5.0	10.0	0.226	168
180	267	7.5	15.0	0.024	18
1580	2348	6.5	13.0	0.243	180
	•				

- * LDL protein concentration calculated assuming a molecular weight of 500,000 Daltons for the apoprotein B content of each LDL particle.
- Table 19 : Proportions of chlorambucil cholesteryl ester and protein in CBE-LDL complexes prepared as described in section 2.3.4. On average, 92⁺63 drug molecules were incorporated per LDL particle.





Figure 50:

Degradation of chlorambucil ester in a preparation of the drug-LDL complex containing 371 µg of drug ester and 3.8 mg of LDL protein per mL of solution. The complex was stored at 4°C for several days and aliquots removed at intervals for drug quantitation by HPLC as described in Section 2.3.6.(C).

protein in adriamycin-LDL complexes prepared by the methods described in Section 2.3.5 are shown in table 20. After 4 hours incubation in DMSO and PBS buffers, only a few drug molecules had partitioned into each LDL particle. This number was increased by prolonging the incubation time to 24 hours. On the basis of these experiments, 24 hours incubation of drug in DMSO buffer, Section 2.3.5.(c), produced LDL with the highest adriamycin content. Incubation of drug and LDL at higher pH, Section 2.3.5.(f), produced no increase in incorporation. On the contrary, slightly less drug was included, table 20, and significant amounts of adriamycin degradation products were apparent, fig. 51, in the pH 10 incubation mixture when analysed by HPLC, Section 2.3.6. after 24 hours. However, it is interesting to note that only Adriamycin had been incorporated into LDL. Surfactants were used to enlarge the LDL particle and so facilitate the entry of adriamycin into the core space. Tucker et al²⁰³ reported that several different surfactants, including those used here, caused rapid increases in the Stokes radius of the LDL particle. Adriamycin-LDL complexes prepared after incubation of LDL with Brij-35, Triton X-100, and SDS, Section 2.3.5.(g), were, disappointingly, not much higher in drug content than those prepared without surfactants, table 20. As with the other methods, the incorporation of the drug was quite erratic. SDS caused around 80% of the LDL protein to precipitate, before the adriamycin was added. No aggregation was observed in any of the other surfactant/LDL mixtures prior to drug addition.

By subjecting LDL to gradually increasing concentrations of adriamycin, via dialysis; it was hoped that the drug would be incorporated into the LDL particle without such a high degree of aggregation. The actual amount of drug which bound to LDL was less than 1% of the amount available, even in the most concentrated adriamycin solutions, fig. 52.

106

ADRIAMYCIN-LDL

METHOD	INCUBATION	ADR1A CONCENT	MYCIN RATION	LDL PR CONCENT	OTEIN RATION	DRUG TO RAT	PROTEIN IO
	Hours	(JmL)	(nmol/mL)	(Jm/gu)	(nmol/mL)	(gram/gram)	(nmol/nmol)
PBS	4	0.15	0.27	60	0.12	0.0027	2.3
PBS	24	1.40	2.41	40	0.08	0.035	30.1
DMSO	4	0.08	0.14	29	0.06	0.0028	2.3
DMSO	24	2.75	4.74	24	0.05	0.114	94.8
CHOLATE	24	3.20	5.51	43	0.09	0.074	61.2
DMSO/GLYCINE pH 10	24	0.40	0.69	260	0.52	0.0015	1.32
DMSO/NaCl pH 7.2	24	1.1	1.89	300	0.60	0.0036	3.15
DMSO	24	0.20	0.34	30.4	0.06	0.0067	5.6
SURFACTANTS							
BRIJ. 35	24	5.52	9.51	192	0.38	0.029	25
BRIJ. 35	24	1.40	2.4	260	0.52	0.005	4.6
TRITON X-100	24	2.92	5.03	180	0.36	0.016	14
TRITON X-100	24	1.30	2.30	264	0.53	0.006	4.3
SDS	24	0.29	0.50	50	0.10	0.005	58

••

Protein concentration was calculated assuming a molecular weight of 500,000 Daltons for the apo B content of each LDL particle. *

Adriamycin and protein content of adriamycin-LDL complexes prepared by the various methods described in section 2.3.5. Table 20 :


RETENTION TIME (mL)

Figure 51:

HPLC profiles of adriamycin/LDL incubation mixtures after 24 hours at 25 °C in buffers containing 10% dimethyl sulphoxide and (A) 0.15 M NaCl, pH 7.2 (sample diluted 1: 10) <u>or</u>

(B) 0.10 M glycine, pH 10.

A : adriamycin, B : daunomycin internal standard

At higher concentrations, the incorporation of drug into LDL as a function of the free adriamycin concentration, fig. 53, began to reach a plateau. The aggregation of LDL was not significantly reduced by the dialysis procedure, which yielded adriamycin-LDL containing up to 18 drug molecules per particle.

2.4.7. DAUNOMYCIN-LDL

.

Daunomycin-LDL was prepared as described in Section 2.3.5.(a). The degree of daunomycin incorporation was variable, table 21, as with the other drugs studied. On average, 120 daunomycin molecules (range 40-190) were sequestered within each particle, giving drug concentrations up to 0.73 mg/mL.

2.4.8. IN VITRO TESTING OF CYTOTOXICITY OF DRUG-LDL COMPLEXES

Daunomycin-LDL (D-LDL), adriamycin-LDL (A-LDL) and chlorambucil-ester-LDL (CBE-LDL) complexes were prepared, as outlined in Sections 2.3.4 and 2.3.5. The cytotoxicities of these complexes were compared to those of the native drugs. In all experiments, controls were set up containing no drugs. Identical plates were prepared with native LDL or LDL which had undergone the drug incorporation procedure, in the absence of drug. These controls induced no significant difference in cell survival from that seen in the cells to which neither drug nor LDL had been added. Additionally LDL and uncomplexed drug, added separately to the medium induced dose response exactly like cells treated only with native drug. All of the complexes studied had cytotoxic effects, although the extent of enhancement of cell kill conferred by the use of LDL as a vehicle was variable.

In G-CCM glioma cells, daunomycin-LDL was less toxic than an equal

Incorporation Of Adriamycin Into LDL By Dialysis



Figure 53: Incorporation of adriamycin into LDL by dialysis into the lipoprotein solution, as a function of the free adriamycin concentration.

DAUNOMYCIN CONCENTRATION		LDL PROTEIN* CONCENTRATION		DRUG TO PROTEIN RATIO	
(µg/mL)	(nmol/mL)	(µg/mL)	(nmol/mL)	(gram/gram)	(nmol/nmol)
95	168	650	1.3	0.146	· 130
300	531	1600	3.2	0.188	· 166
280	496	6500	13.0	0.043	38
400	709	2500	5.0	0.160	142
379	673	5000	10.0	0.076	67
725	1285	3400	6.8	0.213	189

- * LDL protein concentration calculated assuming a molecular weight of 500,000 Daltons for the apoprotein B content of each LDL particle.
- Table 21 : Proportions of daunomycin and protein in daunomycin-LDL complexes prepared as described in section 2.3.4. On average 122⁺58 drug molecules were incorporated per LDL particle.

..

dose of native drug, fig.54. When tested in the microtitre system, Section 2.3.12, complex and drug had ID_{00} 's of 3.5 x 10^{-6} M and 5.6 x 10^{-7} M respectively. When tested on L-DAN cells by clonogenic stem cell assay, Section 2.3.13, free and LDL-bourne drug were almost indistinguishable in their cytotoxicity, table 22, despite the presence of high receptor-mediated LDL assimilation in L-DAN cells, fig. 55, Section 2.3.17. The clonogenic ID_{50} 's for intact spheroids which were disaggregated after treatment, Section 2.3.13, showed that spheroids were significantly less sensitive to both daunomycin and daunomycin-LDL than their monolayer counterparts, table 22. Spheroid growth delay Section 2.3.13, was significantly longer for daunomycin-LDL compared to native daunomycin at a drug concentration of 5 μ g/mL, table 22. Qualitative fluorescence microscopy on the spheroids after drug exposure showed that D-LDL, fig.56 had diffused further into the spheroids than free daunomycin, fig. 57. In HeLa cells (microtitre assay, Section 2.3.12) the D-LDL complex was marginally more cytotoxic than the free drug, fig. 58.

Adriamycin, when carried in LDL, was active at a lower concentration than free drug, fig. 59, with ID_{90} 1.1 x 10^{-8} M compared to 1.6 x 10^{-7} M for native adriamycin, when tested on HeLa cells in the microtitre assay Section 2.3.12.

Chlorambucil was more effective as a cytotoxic agent when carried in LDL, fig.60. The ID_{50} of the drug fell from $1 \times 10^{-5} M$ to $2 \times 10^{-7} M$ when transported within LDL.



Figure 54: Comparative cytotoxicities of daunomycin and daunomycin-LDL on human glioma cells in the microtitre assay. Each point is the mean of duplicate incubations in three experiments.

EXPOSURE TO

	Daunomycin	Daunomycin-LDL
MONOLAYER		
CLONOGENIC SURVIVAL (ID ₉₀)	1.5 µg/mL	l.6 µg/mL
SPHEROIDS		
CLONOGENIC SURVICAL (ID ₅₀)	1.9 µg/mL	1.3 µg/mL
GROWTH DELAY	20 days	23.2 days

Table 22 : Comparative cytotoxicity of daunomycin and daunomycin-LDL, assessed in monolayer and spheroid culture as described in section 2.3.13.

.



Illiant Internalisation and degradation of ¹²⁵I-labelled-LDL via high- and low- affinity routes in L-DAN cells in monolayer culture.

Figure 55:

Figure 56:

Fluorescence micrograph of an L-DAN spheroid following exposure to daunomycin-LDL.

Figure 57:

Fluorescence micrograph of an L-DAN spheroid following exposure to native daunomycin.



HeLa MICROTITRE ASSAY



Figure 58: Relative cytotoxicities of native and LDLencapsulated-daunomycin against HeLa cells in the microtitre drug sensitivity assay. Each point is the mean of duplicate incubations in three experiments.



Figure 59: Relative cytotoxicities of native- and LDL encapsulatedadriamycin against HeLa cells in the microtitre drug sensitivity assay. Each point is the mean of duplicate incubations in three experiments.

Daunomycin-LDL or daunomycin were given by IV injection, at a dose of 4 mg/Kg, to groups of mice implanted with primary gastric tumour, as described in Section 2.3.14. The apparent increases in tumour size for each group are shown, along with those for untreated animals, in fig. 61. A significant inhibition in tumour growth (p<0.05, Wilcoxon) was observed only in the mice treated with D-LDL. However, histological examination of tumour sections revealed that the apparent tumour growth was actually a measure of lymphocyte infiltration, and very few viable tumour cells were present on day 6 in any of the tumour implants, whether treated or untreated, fig. 62. Thus. the observed reduction in "tumour" size in mice treated with D-LDL was actually a result of drug induced immunosuppression. The SRC assay has since been shown to be invalid for primary gastric tumours 214. A second group of mice was implanted with rat Walker256 tumour (obtained as described in Section 2.1.12) and treated with daunomycin, D-LDL, LDL or left untreated. Despite the presence of a large number of viable tumour cells and only a small proportion of lymphocytes, no difference in tumour growth was observed in any of the drug or LDL treated mice when compared to controls.

Mice implanted with Walker256 tumour were treated with chlorambucil, CBE-LDL (8 mg of chlorambucil per Kg body weight) or LDL. Results obtained are shown in fig. 63. Both chlorambucil and chlorambucil-LDL were successful in suppressing growth of tumour. The LDL-treated group did not show any significant difference in tumour size when compared to the control group. Histology showed viable tumour cells to be present in all tumour implants, but variable degrees of lymphocyte infiltration, necrosis and fibrosis, fig. 64, were observed in a

Glioma Microtitre Assay



Figure 60: Comparative cytotoxicities of native chlorambucil and chlorambucil ester-LDL against human glioma cells in the microtitre drug sensitivity assay. Each point is the mean of duplicate incubations in three experiments. Daunomycin-LDL or daunomycin were given by IV injection, at a dose of 4 mg/Kg, to groups of mice implanted with primary gastric tumour, as described in Section 2.3.14. The apparent increases in tumour size for each group are shown, along with those for untreated animals, in fig. 61. A significant inhibition in tumour growth (p<0.05, Wilcoxon) was observed only in the mice treated with D-LDL. However, histological examination of tumour sections revealed that the apparent tumour growth was actually a measure of lymphocyte infiltration, and very few viable tumour cells were present on day 6 in any of the tumour implants, whether treated or untreated, fig. 62. Thus. the observed reduction in "tumour" size in mice treated with D-LDL was actually a result of drug induced immunosuppression. The SRC assay has since been shown to be invalid for primary gastric tumours²¹⁴. A second group of mice was implanted with rat Walker256 tumour (obtained as described in Section 2.1.12) and treated with daunomycin, D-LDL, LDL or left untreated. Despite the presence of a large number of viable tumour cells and only a small proportion of lymphocytes, no difference in tumour growth was observed in any of the drug or LDL treated mice when compared to controls.

Mice implanted with Walker256 tumour were treated with chlorambucil, CBE-LDL (8 mg of chlorambucil per Kg body weight) or LDL. Results obtained are shown in fig. 63. Both chlorambucil and chlorambucil-LDL were successful in suppressing growth of tumour. The LDL-treated group did not show any significant difference in tumour size when compared to the control group. Histology showed viable tumour cells to be present in all tumour implants, but variable degrees of lymphocyte infiltration, necrosis and fibrosis, fig. 64, were observed in a





mice implanted with human primary gastric tumour in the subrenal capsular assay. Apparent growth-inhibitory effect of daunomycin-LDL compared to daunomycin on Figure 61:

Figure 62:

-

Haematoxylin and eosin stained section of mouse kidney and gastric tumour implant on day 6 of the SRC assay. Note the large degree of lymphocyte infiltration (A) and lack of viable tumour cells (B).



Growth-inhibitory effects of chlorambucil, chlorambucil ester-LDL, and native-LDL on Walker256 tumour implants in the mouse subrenal capsular assay.

Subrenal Capsular Assay - Walker 256 Tumour

Figure 63:

FIGURE 62



Figure 64 (I-IV): Haematoxylin and eosin stained sections of mouse kidney and Walker256 tumour implants on day 6 of the SRC assay.

Note the areas of :-

- (A) lymphocyte infiltration
- (B) necrosis
- (C) fibrosis
- (D) viable tumour cells





random manner in each group, making assessment of the tumour size data qualitative rather than quantitative. In this group of animals the opportunity was taken to measure white blood cell counts, Section 2.3.15. Chlorambucil-treated animals had significant immunosuppression, fig.65, with the mean value only 25% of control. As expected, native LDL-treated mice showed no reduction in white blood cell number. The drug carried in LDL appears to have reduced the extent of immunosuppression with mean white blood cell count 81% of control(not significant, Wilcoxon) in CBE-LDL treated animals, while maintaining the same degree of tumour kill as an equivalent dose of native chlorambucil.

WBC Counts After Treatment With Drugs In The SRC Assay In



capsular assay. Tumour size data for these animals are shown in figure 63. plasma white cell counts of mice treated with these agents in the subrenal Effects of native chlorambucil, chlorambucil ester-LDL, and native-LDL on Figure 65:

SECTION 3

DISCUSSION

:

3. DISCUSSION AND CONCLUSIONS

3.1. LDL RECEPTOR ACTIVITY IN VIVO

Numerous in vitro experiments 12-22 have shown that LDL may be a useful vehicle for the selective delivery of drugs or radioisotopes to cancer cells. However, until recently, little was known of the likely degree of specificity which LDL would have as a selective targeting vehicle in vivo. Growing tumour cells, like other rapidly dividing cells, require a considerable amount of cholesterol for membrane synthesis. In many cell types it has been reported that LDL, the major cholesterol-carrying lipoprotein in human plasma, is the preferred source of the sterol when available. Alternatively cellular cholesterol requirements are met by de novo synthesis from acetate²⁵. In this study a double-isotope procedure was used to assess relative receptor-mediated and receptor-independent catabolism of LDL by individual organs and tumours in mouse and rat models. This approach has been shown previously 67,93,94 to be adequate to detect changes in LDL catabolism induced by diet or lipid-lowering drugs. In the animals examined in the present study, as in the human, the liver and adrenal gland are by far the most active in assimilating LDL via both receptor and non-receptor routes. Because of its size, the liver is responsible for the catabolism of a large proportion of plasma LDL and so plays a dominant role in LDL metabolism in control and tumourbearing animals. Nevertheless, tumour cells were active in taking up The Mac-13 tumour exhibited substantial receptor-mediated uptake LDL. of the lipoprotein (fig.19) and was second only to liver in activity. This was also true of the polyoma virus-induced adenoma in mice (fig.21). Colonic tumours generated by administration dimethylhydrazine¹⁹³ to rats was considerably more active in LDL assimilation via receptors

than any other tissue examined, including liver (fig.22) and Walker-256 rat tumour was also active in receptor-mediated LDL catabolism, assimilating by this route almost as much LDL as liver (fig.23). In the experiments using hamster pancreatic tumours (fig.26) only total LDL uptake (i.e. uptake by both receptor-independent and receptormediated routes) was measured. Consequently no definite conclusion can be drawn regarding the receptor activity of these tissues. Here the tumours' acquisition of LDL was modest in comparison to the relatively high levels of LDL assimilation in other hamster tissues. In the rat model (fig.22), total LDL uptake by colonic tumours was surpassed only by liver and spleen. The high activity of the spleen probably reflects the removal of denatured tracer material, which is mediated by cells of the R.E. system. Although this corresponds to less than 5% of the injected dose, the organ appears to be particularly avid since the denatured tracer is directed specifically at it.

Thus all of these tumours, although of varying aetiology, exhibited substantial LDL catabolic activity and are potentially susceptible to LDL-mediated cytotoxic therapy. However, specificity remains a problem and it is likely than the liver and adrenal glands would suffer during chemotherapy, although it should be noted that the presence of tumour in mice appeared, by some undefined mechanism, to reduce receptor-mediated hepatic uptake of LDL (tables 5 & 7). Certainly, there were no macroscopically apparent metastases in the animals' livers which could have contributed to this effect. Consequently, it is necessary to protect these organs from the toxic effect of LDL therapy by reducing their receptor expression. This is possible since the number of receptors expressed by the cells of each organ can be influenced by changing the intracellular pool of cholesterol. Kovanen et al⁴⁰ have shown that, in mice, adrenal LDL receptor activity is markedly activated by injections of ACTH and, in the dog, it has been demonstrated that hepatic receptor activity can be suppressed by dietary supplements of cholesterol¹⁹⁶ or bile acids¹⁹⁵⁻¹⁹⁷. However, in the groups of mice examined here there was little impact of cholesterol feeding on receptor-mediated hepatic LDL uptake (table 13) despite a substantial increase in plasma cholesterol in response to diet (table 6). Interspecies variation in the ability of cholesterol to inhibit receptors is well recognised. For example, dietary sterol supplementation is ineffective in changing rat plasma cholesterol levels since the bulk of excess sterol is excreted in the bile^{29} . In man^{215} on the other hand, the primary response appears to be to export cholesterol into the plasma as LDL rather than retain it in the liver and modulate receptor activity; and bile acid feeding in the mice used in the present investigation was more successful in reducing the expression of hepatic LDL receptors than was dietary cholesterol supplementation (table 14). This parameter fell by 25% (p<0.05, Wilcoxon rank test) in the control mice and by 17% (p<0.05) in the tumour-bearing animals. Presumably the reduced activity is consequent upon suppression of bile acid production from cholesterol, leading to increase in the hepatic sterol pool and downregulation of the an liver membrane receptors (fig.27). In rats bearing colonic tumours however, bile acid feeding did not produce a similar effect (table 15). Taurocholate-treated tumour-bearing mice also showed reduced receptormediated adrenal LDL uptake, a response which was not observed in the control group (table 14). This unexpected phenomenon, like the reduced hepatic receptor activity seen in tumour-bearing mice, presumably reflects a change in lipoprotein metabolism which is secondary to the presence of tumour.

LDL uptake by the adrenal gland was successfully suppressed by the

administration of hydrocortisone (table 14). In the control and tumourinoculated mice, injection of the steroid lowered adrenal LDL receptor activity by 48% and 30% respectively (p<0.05, Wilcoxon rank test, in both cases). In rats with colonic tumours (table 15) hepatic uptake of LDL was reduced by 40% (p<0.05, Wilcoxon rank test) after hydrocortisone administration and adrenal receptor activity was lowered by 40% and 30% (both p<0.05) in control and tumour animals respectively.

When hydrocortisone was combined with taurocholate feeding, both hepatic and adrenal LDL uptakes were lowered in the mice without concomitant lowering of the activity of the tumour (table 14). This regimen also successfully lowered adrenal uptake in the tumourbearing rat model (table 15). It is interesting to note that the ovaries of rats receiving hydrocortisone alone or with bile salts had significantly suppressed LDL receptor activity. This was not merely a secondary response to tumour presence since it also occurred in the matched control group (table 15). The mechanism for this effect is unclear.

3.2 EVIDENCE FOR LDL UPTAKE IN TUMOURS USING CUMULATIVE MARKERS OF LDL CATABOLISM

LDL labelled with radioiodinated cholesteryl linoleoyl ether¹⁰⁶ or radioiodinated cellobiose-tyramine¹⁰⁵ was used to obtain a cumulative index of the extent of catabolism of the lipoprotein by the tumour. These tracers have been reported to be internalised by the cell in an identical manner to native LDL, but once inside lysosomes they resist degradation and are retained there. Less than 35% leaks back into the circulation over a period of 24 hours, in contrast to the immediate release into the plasma of iodotyrosine from conventionally labelled I-apo B-LDL. In MAC-13-inoculated mice, the tumour as a whole was second only to liver (fig.24) in assimilating labelled LDL. Hepatic uptake of I-CE-LDL by the receptor pathway was again suppressed by the presence of tumour (p<0.05, Wilcoxon rank test), as was originally observed using I-apo B-LDL as tracer (fig.19). LDL receptor activity in kidneys and adrenals was also suppressed in the tumour-bearing animals.

The uptake of LDL via both receptor-dependent and receptor-independent pathways by rat colonic tumours (fig.25) and hamster pancreatic tumours (fig.26) was measured using ¹²⁵I-CT-LDL. In both models tumours were highly active in this regard, although several normal tissues in the hamster model showed a similar response.

In assessing the above results, it must be borne in mind that heterologous, human, LDL was used in the animal models because of its ease of preparation and known biological properties. Such an approach may produce different results than might be obtained if homologous LDL were used, but these should be of minor quantitative significance.

The LDL particle consists of an apolar core of cholesteryl esters and triglycerides surrounded by a phospholoid monolayer containing free cholesterol and apoprotein $B^{27,34}$. This provides several sites for the insertion of drug molecules, depending on their lipid solubility. The core is by far the best site, since chemically sensitive drugs sequestered there would be well protected from serum hydrolases and the potentially damaging effects of an aqueous Furthermore, the core has the capacity to store a enviro ment. large number of drug molecules. However, only very hydrophobic agents are suitable for inclusion in this domain. A second site for drug intercalation is the phospholipid monolayer. Agents which have both polar and lipid soluble components may partition here, between the apolar core and the aqueous environment, but in this location, they are less well protected from serum hydrolases and water and further, may be able to stimulate the immune system to remove the drug-LDL complex from the circulation. A third approach would be to attach the drug covalently to the apoprotein moiety of the LDL particle 17 . However, there are a number of immediately obvious drawbacks here since:-

- Drug attachment to apo B may distort the protein and reduce or abolish completely its binding to LDL receptors.
- The drug-LDL link must be successfully hydrolysed to liberate the active agent within the cell.
- 3) The potential of each LDL particle for storing

drug in this manner is considerably less than the capacity of the core.

- Drug carried in this way is exposed to water and serum enzymes.
- 5) Distortion of apo B in linking the drug molecules may render the complex susceptible to removal by the immune system.

Several procedures have been detailed as adequate for the incorporation of trace amounts of apolar substances into the core space of the LDL particle. However, only Kreiger's method ¹⁵⁶, which, in effect, produces "synthetic LDL", has been shown to insert significant amounts of exogenous substances into the lipoprotein. "Kreiger particles", although they react like native LDL <u>in vitro</u> are rapidly removed from circulation by cells of the R.E. system when injected <u>in vivo</u>. This method however is still potentially useful since Masquelier et al¹⁴⁷ have modified Kreiger's method to produce an LDL species containing a lipophilic anthracycline (AD-32) which is removed from mouse plasma at a similar rate to that of native human LDL.

Our attempts to incorporate drugs into LDL focused on using native human LDL as acceptor in order to reduce potential R.E. clearance of the drug complex <u>in vivo</u>. Drugs were either partitioned into the lipoprotein from concentrated drug solutions in various aqueous buffers (anthracyclines) or by exchange from HDL or micellar lipid suspensions (chlorambucil cholesteryl ester).

÷

Daunomycin was easily intercalated into LDL following the method described by Iwanik et al²³. In solutions of high daunomycin concentration, an average of 190 drug molecules were able to diffuse into each LDL particle (table 21). The resulting complexes retained their specific binding ability, 99% of the apo B being precipitated on interaction with LDL antibodies. Furthermore, daunomycin-LDL uptake into P388 leukemic cells was shown by these authors to be subject to competition from native LDL but not HDL. Interaction of the daunomycin-LDL complex with DNA resulted in the quenching of only 45% of the fluorescence of the drug 23 . The size of the DNA molecule precludes its entry into the core space of the particle and it is likely that only drug molecules sequestered on or near the surface of the LDL particle are susceptible to quencing by this probe. A further 13% of the fluorescence was quenched by potassium iodide ²³, since this agent has access to parts of LDL that DNA is unable to reach. Thus Iwanik et al²³ deduced that around 50% of the daunomycin is loosely held on the surface, with the remainder located in the hydrophobic core of LDL. This was further confirmed by the "leakage" of approximately 40% of the daunomycin from LDL during centrifugation at a density of 1.063 Kg/L.

Despite the structural similarities between daunomycin and adriamycin (fig.11), the latter is much more reluctant to partition into LDL. Adriamycin, when incubated with the lipoprotein under conditions similar to those used for preparing daunomycin-LDL, caused its immediate aggregation, a reaction which has been observed by others²³. Consequently, alternative approaches were considered in the preparation of adriamycin-LDL, since this complex has a much larger

potential spectrum of clinical use than daunomycin-LDL. LDL is known to be less prone to aggregation when stored at low concentration in solutions of high ionic strength. For this reason, many of our experiments used the pooled LDL fractions obtained directly from the. rate-zonal centrifugation . Nevertheless, aggregation still remained a problem, possibly due to the formation of an insoluble complex between the drug and cholesterol on the surface of LDL. Brown et al 202 used a sodium chloride buffer containing 10% DMSO (v/v) in a procedure for incorporating labelled cholesteryl esters into LDL. This approach in our hands generated a complex in which a small amount of adriamycin became associated with the lipoprotein. Comparison of the drug incorporation at pH7.2 and pH 10 showed neutral conditions to be better than basic by a factor of 2 (table 20). At pH.10 there was a larger degree of drug degradation (fig.51) although only native adriamycin was incorporated into LDL to any detectable level, i.e. none of the drug's metabolites were as lipophilic as or more lipophilic than the native drug. Tucker et al investigated the use of surfactants for enlarging the structure of the LDL particle. Rapid increases in the Stokes radius of the LDL particles were observed after exposure to a variety of surfactants but the incorporation of adriamycin into these surfactant-enlarged LDL particles was not significantly higher than that obtained in the absence of surfactants (table 20). It would appear therefore that steric factors are not the major cause of the low adriamycin concentration observed in adriamycin-LDL preparations. It is more likely that the hydrophobic environment provided by the LDL core is not conducive to sequestration of the drug. Since the only difference between adriamycin and daunomycin is a hydroxy substituent on Cl4 in the former, it is obvious that this highly hydrophilic group has a dominating effect on the lipid solubility of the drug. Better incorporation might be achieved if this hydroxy

group were selectively esterified by, for example, a fatty acid residue. Vitols et al²¹ have esterified adriamycin with linoleate at this position and have coupled N-retinoyl leucine to the amino group of the sugar. 100-200 molecules of this derivative became incorporated into each LDL particle although the authors do not comment on its aggregant effect on LDL in solution.

3.5. CHLORAMBUCIL

The chemical structure of chlorambucil (fig.ll) was suitable for esterification with cholesterol. This was achieved by a single reaction, using dicyclohexylcarbodiimide as a coupling agent. The object of this derivatisation was to produce a molecule which would be more amenable than the original chlorambucil molecule to incorporation into LDL. The benefits of synthesising a cholesteryl ester were three-fold:-

- The polarity of the carboxylic acid group was markedly reduced, making the drug-ester more hydrophobic than the native drug
- 2) Cholesteryl esters are the native occupants of the core of LDL and the subsequent incorporation of chlorambucil cholesteryl ester there should produce minimal disruptions of the structure of the lipoprotein
- 3) There exists a cholesteryl ester transfer protein in human plasma which mediates the

exchange of cholesteryl esters between LDL and HDL and which may be utilised for incorporation of the drug-cholesteryl ester into LDL.

Stoffel et al¹⁹⁰ have described a method for loading labelled cholesteryl esters and fatty acids into HDL by exchange from cholic acid/lipid micelles. Taking this process a stage further, iodinated cholesterol linoleoyl ether was incorporated into LDL by exchange from HDL, utilising the cholesteryl ester transfer protein from human plasma¹⁹¹. Initially this was the method used for chlorambucil ester incorporation, but the resulting LDL complex had virtually no cytotoxic activity (fig.46). This lack of cytotoxicity had two obvious possible causes: either the cells did not have adequate receptor expression; or the drug was being inactivated during the incorporation procedure. The former explanation was ruled out since the glioma cell line used, G-CCM, had already been reported²⁰ to have high receptor expression under similar conditions. On examination of the buffers used in the loading process tris(hydroxymethyl) methylamine (tris) was identified as a probable culprit for the drug inactivation. This nucleophile was likely to be attacking the carbon attached to chlorine on the chloroethyl side chains of chlorambucil, releasing chloride and rendering the drug inactive. Drug activity was conserved by replacing tris buffer with phosphate buffer, although the incorporation of chlorambucil ester remained low (table 16). Later, drug ester was exchanged from the cholic acid/lipid micelles directly to LDL, with an increase in the overall drug recovery (table 16) and no obvious effects on the integrity of the lipoprotein. By following the cholesteryl chlorambucil ester incorporation by HPLC at each stage of the cholate preparation, it was clear that a substantial amount of the
drug had been degraded over the course of the experiment. Studies on the degradation reaction surprisingly showed no free chlorambucil in the solution. Consequently the process appeared not to involve simply a hydrolysis of the ester bond between chlorambucil and cholesterol. At first the degradation was attributed to enzymes present in the crude preparation of cholesteryl ester transfer protein, but the extent of disappearance of the drug ester was higher in solutions containing no enzymes (fig.49) This is in accordance with the observation by Alberts et al¹⁶⁷ that free chlorambucil is degraded faster in water than in human plasma <u>in vitro</u>.

As an alternative to the modified Stoffel procedure (Section 2.3.2), which yielded LDL of low drug content, Craig's procedure (Section 2.3.4) for making drug/lipid microemulsions was examined. This method was originally devised for making LDL of defined cholesteryl ester composition and involves preliminary preparation of drug-lipid microemulsions. The process is much shorter than the Stoffel method, thus reducing the degree of drug degradation. Up to 30% of the drug used was recovered in LDL after exchange from these microemulsions and the drug-LDL complex was easily separated from them by microfiltration and centrifugation. It is interesting to note that, on storage at 4°C, a portion of the drug within the LDL complex was degraded rapidly over the first 24 hours and thereafter degradation was much less significant (fig.50). This biexponential drug decay points to the possibility that chlorambucil cholesteryl ester molecules are located both on the surface (the rapidly degraded component) and in the core of the LDL particle. Assuming that this deduction is correct, core-located drug molecules obtain protection from the aqueous environment when sequestered in LDL under the conditions used. Native chlorambucil decays with monoexponential kinetics 204, leaving only 10%

cf the original drug after 6 hours incubation at pH7 and 25° C, compared with recovery of 50% of the drug in LDL after 24 hours of incubation under similar conditions, a further indication of the protective environment provided by LDL.

3.6 IN VITRO TESTING OF DRUG-LDL COMPLEXES

The sensitivity of tumours to various antineoplastic agents in vitro has long been studied as a possible indicator of their in vivo response to proposed chemotherapeutic regimens. The relative in vitro colony forming ability of treated and untreated cell samples is commonly used as a method of assessing drug sensitivity 207. However, this method is not always readily applicable to primary cultures of human tumours as a result of their low cloning efficiency and slow growth rate. Furthermore, assays based on cell proliferation ability alone require extensive cell counting and are impractical for large numbers of samples. As an alternative to clone counting, cellular metabolic activity may be measured after exposure to drugs. Protein synthesis is one biosynthetic event which has been identified as a fundamental metabolic process, without which the cell cannot survive. Since leucine is an essential amino acid, the cell must acquire it from the surrounding medium and protein synthesis is most unlikely to occur without incorporation of this molecule. Thus $\begin{bmatrix} 3 \\ H \end{bmatrix}$ -leucine may be added to the culture medium and its extent of incorporation will be indicative of the metabolic activity of the cells. Consequently, the absence of intracellular $\begin{bmatrix} 3\\ H \end{bmatrix}$ -leucine several hours after drug removal will almost certainly correlate with cell death. microtitre drug sensitivity assay based on such an approach has been described for HeLa²⁰⁸ and human glioma²⁰⁷ cell lines. This procedure has several advantages over the aforementioned clonogenic assays,

123

particularly with regard to the large numbers of replicates which can easily be handled. The relatively long drug exposure time ensures that the majority of the cells in culture are able to pass through the entire cell cycle at least once during this period. Consequently, even phase-specific drugs may be adequately assessed by the procedure.

Intermediate between monolayer culture systems and <u>in vivo</u> tumour models is the spheroid model²¹⁰ of cell culture. In this system, cells are grown in multi-layer 3-dimensional arrays which are able to mimic better the arrangement of cells in the intact tumour. The ability of drugs to penetrate the microarchitecture of tumour cell layers is a variable which the monolayer system completely ignores. This distribution factor can be tested, to some degree at least, in the spheroid culture system. Drug penetration is, of course, of major importance in the treatment of tumours, since the deeper the drug is able to move into the tumour, the larger the fraction of tumour cells which will be killed.

Since relatively high concentrations of daunomycin were able to be incorporated into LDL, this complex was tested on a variety of cell culture systems. In monolayer experiments, using human glioma cells (fig.54) and HeLa cells (fig.58) in the microtitre assay and L-DAN cells (table 22) in the clonogenic assay, the targeting of daunomycin within LDL did little or nothing to increase the cytotoxicity of the drug. There are two obvious reasons for this lack of enhancement of daunomycin cytotoxicity:-

 Drug accumulation within the cell is not increased when daunomycin is carried within LDL either because the daunomycin influx/efflux mechanism is altered by the change in the initial metabolic treatment of the drug following internalisation via the LDL pathway or because the LDL pathway in these cells becomes saturated at relatively low daunomycin-LDL concentrations; alternatively

 Daunomycin may be denatured during its internalisation with LDL, before it can successfully carry out its cytotoxic action.

Iwanik et al²³ in their experiments with daunomycin-LDL on human leukemic cells found that drug accumulation within the cells was higher when the drug was administered within LDL, but, unlike the native drug, which was located, as expected, mostly in the nuclear fractions of the cells, daunomycin delivered via LDL was more abundant in the microsomalmitochondrial-lysosomal membrane fractions. This observation lends credence to the possibility that the drug had undergone a metabolic transformation before its cytotoxic action could be achieved. Of course, drug and LDL uptake and membrane fractionation experiments would have to be carried out on the cell lines used in our work to evaluate the validity of the above explanations. The results of the spheroid experiments were more favourable, daunomycin-LDL being more effective than native drug on the spheroids both when measured by spheroid growth delay and in clonogenic assay after disaggregation. Fluorescence microscopy on spheroid cross-sections after drug exposure (fig.56,57) showed that daunomycin had penetrated further into the tumour structure when carried within LDL. This improvement in drug distribution obviously had a beneficial effect on the growth delay of the spheroids treated with daunomycin-LDL.

Adriamycin was more successful as a cytotoxic agent when administered

125

within LDL to HeLa cells in the microtitre assay (fig.59). The action of this drug differs from daunomycin in that it has well-documented cytotoxic effects in the cytoplasm¹³⁰ and at the plasma membrane, as well as at the nucleus¹⁷². Consequently the change in drug distribution seen when daunomycin was carried in LDL²³, if it occurs in a similar manner with adriamycin-LDL, may not decrease the cytotoxicity of adriamycin so significantly, or may even increase it. Lysosomal degradation of adriamycin is not a problem with free or LDL-borne drug since it has already been shown that this anthracycline is resistant to lysosomal enzymes²¹⁶. There is also the possibility that adriamycin uptake into HeLa cells is low in comparison to drug uptake via LDL.

Chlorambucil cholesteryl ester-LDL was by far the most successful of the complexes tested in the glioma microtitre assay, with the $ID_{_{\rm EO}}$ of the drug lowered by a factor of 200 when it was targeted in LDL (fig.60). As already mentioned, encapsulation of the chlorambucil ester within LDL appears to reduce the hydrolysis rate of the drug and this may result in the availability of a comparatively higher concentration of active drug inside the cell over the period of drug exposure. Influx of nitrogen mustards has been shown 217 to be an active transport process in human leukemic and normal lymphoid tissue. Chlorambucilresistant cells of Yoshida ascites sarcoma have been reported 218 to accumulate 50% less drug than their sensitive counterparts, under identical conditions. Therefore the cell's ability to internalise chlorambucil may have a profound effect on cellular susceptibility to this agent, although in the case of the Yoshida ascites sarcoma 218 , other differences were also observed between the sensitive and resistant sublines. Targeting of chlorambucil within LDL in this way may then conceivably increase the drug's cytotoxic effects simply by increasing

drug influx, so exposing tumour to high drug concentrations for a longer period. It has also been postulated²¹⁹ that, once inside the cell, chlorambucil immediately associates with lipid-protein complexes whence it is released at a slow rate prior to carrying out its alkylating action. Thus, administration of the drug within LDL may effectively insert the drug into the cell at a stage further along its normal metabolic pathway. Conversely, the distribution of daunomycin within subcellular compartments is markedly different when it is administered within LDL, and this difference may well account for the observed decrease in cytotoxicity of daunomycin-LDL compared to the native drug.

3.7 IN VIVO TESTING OF DRUG-LDL COMPLEXES - THE SUBRENAL CAPSULAR ASSAY

Cell experiments are of value in predicting the likely sensitivities of tumour cells to antineoplastic drugs, but they do not take into account the effects of other organs on the distribution, metabolism and excretion of the drug. The object of using drugs within a targeting vehicle is to favourably alter the pharmacokinetic disposition of the drug so that tumour cells are harmed considerably more than normal tissues. This property can, obviously, only be tested significantly using an in vivo tumour system. Also, drugs requiring in vivo activation may only be adequately assessed using an animal model unless the active agent can be isolated. There is still considerable controversy over the best type of animal model to use for experimentation with cytotoxic drug regimens and over the relevance of the results obtained with regard to extrapolation to the human situation. In many of the models currently in use, drugs are administered very early in the growth of the tumour, often less than 24 hours after implantation, whereas in the clinical situation the neoplasm, and often metastases,

are usually long established before treatment commences. Although transplantable animal tumours are, in general, much faster growing than human tumours, the administration of drugs to the animals occurs undoubtedly at a relatively early stage in the progression of the neoplasm. Also, doses up to the LD₁₀ (the lethal dose which kills 10% of the animals which receive it) are often used against animal tumours while similar doses cannot be used in humans.

In order to test drug-LDL complexes <u>in vivo</u> the mouse model first described by Bogden et al²¹² was employed. This model, the Subrenal Capsular (SRC) assay, involves implantation of tumour fragments under the renal capsule in normal immunocompetent mice. Bogden's model has several advantages over other tumour systems. The assay is rapid, giving a result in only six days, and other workers²²⁰ have used the same procedure with a four-day timescale. Normal immunocompetent mice are used²¹⁸, so reducing the need for sterile conditions. Fragments of solid tumours are implanted thus maintaining the spatial relationship between tumour cells of the same, and different subpopulations, a relationship which may have a profound effect on drug penetration into the tumour and ultimately on overall tumour response to chemotherapy²²¹. The drug administration protocol resembles that used in the treatment of humans, unlike the single drug dosage scheme employed in many animal models.

Results obtained here using daunomycin-LDL against a human primary gastric adenocarcinoma (fig. 61) appeared at first to be very promising with tumour showing an apparently lesser size increase after treatment with the complex compared to treatment with native daunomycin. However, histological examination of tumour sections revealed a major problem with the SRC assay: Bogden's original statement that the host immune system did not invade the tumour implants to any great extent until after six days²¹² was incorrect. The large, but variable degree of lymphocyte infiltration observed in all tumour groups made the interpretation of results, by measurement of tumour dimensions, very difficult. This was further complicated by the low numbers of viable tumour cells present at day 6 (fig.62), even in the control animals, and it was concluded that the apparent differences in tumour size were in fact differences in lymphocyte infiltration. It appeared therefore that daunomycin-LDL had a greater effect in suppressing the mouse immune response (and limiting immunocyte infiltration) than that observed after treatment with native daunomycin.

At only one day after tumour implantation, Cunningham et al 214 noted a considerable lymphocytic infiltrate in tumour graft sections. Others too^{222,223} have observed immunocyte contributions to tumour size, although many have classed this contribution as insignificant, stating that it accounted for less than 20% of the total tumour size 223 . However, even at this relatively low level, the degree of lymphocyte infiltration may bias the results, as in the case of daunomycin-LDL above, since many of the common cytotoxic agents have marked immunosuppressive activity. Levy et al have composed a scoring system whereby tumour size, lymphocyte infiltration, necrosis and fibrosis are among the factors taken into consideration. This is obviously a better system of assessment but it has the disadvantage of requiring more detailed and more laborious analysis of tumour sections before conclusions can be drawn. After consideration of the low number of viable tumour cells in the control animals and the significant degree of lymphocyte infiltration in the grafts used for testing daunomycin-LDL, and numerous other grafts of this type, the SRC assay was concluded ²¹⁴ to be of no value, in assessing the chemosensitivity of

When animal tumours were used in the SRC assay a less extensive lymphocyte response was observed²¹⁴. The rat Walker-256 tumour. originally a mammary adenocarcinoma, has been used in rat models for many years. When daunomycin and daunomycin-LDL were tested against this tumour, neither agent was able to limit its growth; but after treatment with chlorambucil ester-LDL or chlorambucil, Walker-256 tumour sizes were smaller than those observed in the control or native-LDL treated groups (fig.63). Microscopic evaluation of tumour sections showed large aggregates of viable tumour cells in each graft, but also present in random fashion throughout both control and test groups were variable proportions of necrotic, fibrotic and lymphocytic tissue, although lymphocytic infiltration was at a much lower level than that observed with the primary gastric tumours. The pattern of tumour necrosis, when present, was always the same. Those tumour cells in the layers furthest from the kidney surface were the first to die (fig.64). This implies that the tumour's demand for nutrients were in excess of the capabilities of the transport system within the graft, despite the concentration gradient between kidney and graft and the presence of simple blood vessels within the tumour. Perhaps this indicates another limitation of the SRC assay, although other groups 212,220 have reported high percentage increases in tumour size with no mention of necrosis.

So, interpretation of the results obtained from the SRC assay is not straightforward since factors other than tumour dimensions must be taken into account.

The object of using a vehicle for chemotherapeutic agents is not only

to deliver more drug to its site of action, thereby inducing more tumour cell kill, but also to decrease its toxic effects on normal tissues. A common dose-limiting side effect of cytotoxic therapy is immunosuppression and since the degree of lymphocyte infiltration into tumour in experiments with gastric tumours was affected by daunomycin-LDL , white cell number per mL of blood was measured following drug exposure in later studies. On comparison of the mean white blood cell counts in each treated group with the control group on day 6 of the assay (fig.65), only native chlorambucil was found to have a significant lowering effect. Thus, since chlorambucil and chlorambucil ester-LDL treated groups exhibited similar rates of tumour growth (fig.63), this drug, when carried in LDL, may be said to exert the same apparent tumour inhibition as an equivalent dose of the native drug, with less severe immunosuppression.

3.8 CONCLUSIONS

The consistent lack of success in designing specific drugs which will effectively destroy tumours without harming other tissues has prompted the development of new methods of delivery for the existing cytotoxic agents. Several drug-vehicle combinations¹⁻⁷ have been used with varying efficacy. Liposomes⁶ fulfilled a number of the requirements of the ideal vehicle, but tumour specificity and extensive clearance of these particles by the R.E. system $\frac{\text{in vivo}^{8-11}}{\text{remain serious}}$ problems. Following the observations of Ho et al¹² and later Gal and co-workers¹³, a "natural liposome", low density lipoprotein, was proposed as a vehicle for cytotoxic drugs or imaging isotopes.

Numerous reports¹²⁻²² on LDL assimilation by cultured human and animal tumour cells have been published, and the concensus of opinion is that neoplastic cells express higher LDL receptor activity than their equivalent non-neoplastic counterparts, although there have been instances of cultured cancer cells with defects in LDL metabolism²²⁴⁻²²⁶ . Receptor-mediated uptake of LDL by solid tumours has been measured in several animal models (Section 2.2) and was found to be second only to liver and adrenal glands on a gram for gram basis. Thus LDL is not a totally specific carrier <u>in vivo</u>, but it may be sufficiently so to favourably increase the benefit:risk ratio of treatment. Specificity may be enhanced by reducing receptor-mediated uptake of LDL by liver and adrenal glands by administration of the appropriate agents (Section 2.2).

The efficacy of chemotherapy depends critically on effective penetration of the drug into tumour microarchitecture, although the investigation of this property in vivo has been largely neglected. In order to enhance drug penetration, a number of lipophilic cytotoxic agents have been synthesised, and these are currently being evaluated²²⁹. As an alternative, existing drugs may be introduced further into the tumour within a lipophilic carrier. Daunomycin, encapsulated in LDL, showed more extensive penetration into tumour spheroids than did the native drug (figs. 56 & 57) or a number of lipophilic anthracyclines²²⁸ and consequently induced a considerably longer growth delay in this model system. It would therefore be of interest to obtain a measure of comparative distributions of free and LDL-encapsulated drug in animal tumour models and correlate drug penetration with tumour regression.

A potentially useful property of this delivery system was illustrated 158 using daunomycin-LDL on anthracycline-resistant leukemic cells. Drug carried by the lipoprotein was able to penetrate the resistant cells much better than the native drug, suggesting that resistance was mediated by a change in the membrane transport of the daunomycin. Since acquired resistance is an all too common phenomenon following initial chemotherapy, the use of LDL as a vehicle for some cytotoxic agents may help to circumvent this problem. Further resistance is unlikely to be a result of blocking this route of entry since the tumour must rely heavily on LDL for the cholesterol it requires to maintain its rapid rate of membrane synthesis¹⁶, and LDL has been cited as a necessary cofactor for the initiation of DNA excision repair in cultured human lymphocytes²²⁷. Thus the cells of tumours receiving, in LDL, a chemotherapeutic agent which attacks DNA, are likely to further increase their LDL receptor expression and this positive feedback may enhance the cytotoxic effects of the drug-LDL complex.

To-date, the extent of drug incorporation into LDL is probably too low for therapeutic use in humans. Perhaps additional improvements to core replacement techniques, like Kreiger's method¹⁵⁶, may permit the synthesis of LDL with a high drug content, which behaves like native LDL <u>in vivo</u>. Alternatively the optimisation of methods of fusing drugrich microemulsions with LDL²⁰⁰, as used in producing chlorambucilester-LDL (Section 2.3.4) might produce drug-laden LDL particles which retain their native structure. However, it may be that the recently introduced "intensely potent anthracyclines"²²⁸ will enhance the efficacy of existing procedures for making drug-LDL complexes for clinical use. These new drugs²²⁸ are active at 1/10th to 1/100th of the doses commonly required for existing anthracyclines, and sufficient levels of these agents may therefore be incorporated into LDL simply by the diffusion techniques used previously (Section 2.3.5).

Another aspect of drug-LDL complexes which requires further study is the way in which the cytotoxic agents are distributed within the lipoprotein particle. For example, it appears that approximately 30% of the chlorambucil cholesteryl ester (Section 2.4.5) and 50% of the daunomycin²³ are located in the apolar core of the lipoprotein, while the remainder is held on the surface. The degree of protection afforded to core-located drug has not yet been documented, nor has the rate of leakage of drug molecules into plasma, but these properties are likely to be of importance in determining the efficacy of delivery of some cytotoxic agents.

Existing preparations of drug-LDL complexes, of course, require further testing in animal models, although unlike new cytotoxic agents, many of the toxic side effects of these drugs are already well known. The distribution of drug should, however, be dictated by the vehicle, and the major sites of toxicity may consequently be quite different from those of the free drug. Studies with radioiodinated LDL in animal tumour models (Section 2.2) have shown that the adrenal glands and the liver are likely to be sites of dose-limiting side effects of drug-LDL therapy. Comparative dose response and tissue distribution data for free and LDL-encapsulated drug should be collected prior to testing any drug-LDL preparation in humans. Among the toxic effects which may be reduced by administering drugs within LDL is immunosuppression. Preliminary work with chlorambucil ester-LDL indicates that the complex may be significantly less injurious in this regard than the free drug.

In summary, LDL is a particle which exhibits a number of characteristics which make feasible its use as a drug vehicle <u>in vivo</u>. Tumours express relatively high LDL receptor activity and it is likely that serious damage to other organs can be avoided. Furthermore, LDL enhances drug penetration into the tumour and may be of use in overcoming drug resistance. It is not unrealistic to expect that, with the development of better incorporation methods for existing drugs and the synthesis of more potent or more lipophilic drugs, LDL will be successful as a targeting vehicle for the treatment of cancer in humans in the not too distant future.

SECTION 4

REFERENCES

:

4. REFERENCES

- DEPREZ-DE CAMPENEERE, D., BAURAIN, R., HUYBRECHTS, M., TROUET, A., 1979, Comparative Study in Mice of the Toxicity, Pharmacology and Therapeutic Activity of Daunorubicin-DNA and Doxorubicin-DNA Complexes, <u>Cancer Chemother. Pharmacol.</u>, 2, 25-30.
- SZERKE, M., WADE, R., WHISSON, M.E., 1972, The use of Macromolecules as Carriers of Cytotoxic Groups (Part II).
 Nitrogen Mustard-Protein Complexes, <u>Neoplasma</u>, <u>19</u>, 211-215
- 3. TROUET, A., MASQUELIER, M., BAURAIN, R., 1982, A Covalent Linkage Between Daunorubicin and Proteins that is Stable in Serum and Reversible by Lysosomal Hydrolases, as Required for Lysosomotrophic Drug-carrier Conjugates: In Vitro and In Vivo Studies, <u>Proc. Natl</u>. <u>Acad. Sci. U.S.A.</u>, 79, 626-29.
- KANEKO, Y., 1981, Thyrotropin-Daunomycin Conjugate Shows Receptor-Mediated Cytotoxicity in Cultured Thyroid Cells, <u>Horm. Metab. Res</u>.
 <u>13</u> 110-14.
- FIENERMAN, B., PAUL, R.D., FIENERMAN, G., 1983, Treatment of a Murine Leukemia With Chlorambucil-Bound Monoclonal Antibodies, <u>Adv. Exp. Med.</u>, <u>166</u>, 59-66.
- GREGORIADIS, G., 1976, The Carrier Potential of Liposomes in Biology and Medicine, <u>N. Engl. J. Med.</u>, <u>295</u>, 704-710 and 765-70.
- 7. HEATH, T.D., MONTGOMERY, J.A., PIPER, J.R., PAPAHADJOPOULOS, D.,
 1983, Antibody-targeted Liposomes: Increase in Specific Toxicity
 of Methotrexate-Υ-aspartate, Proc. Natl. Acad. Sci. U.S.A.,
 80, 1377-81.
- POSTE, G., 1983, Liposome Targeting <u>in Vivo</u>: Problems and Opportunities, <u>Biol. Cell.</u>, <u>47</u>, 19-38.
- 9. PROFFITT, R.T., WILLIAMS, L.E., PRESANT, C.A., and 4 others, 1983,

136

Liposomal Blockade of the Reticuloendothelial System: Improved Tumour Imaging With Small Unilamellar Vesicles, Science, 220, 502-5

- SENIOR, J., GREGORIADIS, G., 1984, Role of Lipoproteins in Stability and Clearance of Liposomes Administered to mice, Biochem. Soc. Trans., 12, 339-40.
- 11. SEGAL, A.W., WILLS, E.J., RICHMOND, J.E., et al, 1974, Morphological Observations on the Cellular and Subcellular Destination of Intravenously Administered Liposomes, Br. J. Exp. Pathol., 55, 320-27.
- 12. HO, Y.K., SMITH, R.G., BROWN, M.S., GOLDSTEIN, J.L., 1978, Low Density Lipoprotein (LDL) Receptor Activity in Human Acute Myelogenous Leukemia Cells, <u>Blood</u>, <u>52</u>, 1099-1114.
- GAL, D., OHASHI, M., MACDONALD, P.C., BUCHSBAUM, H.J., SIMPSON, E.R., 1981, Low Density Lipoprotein as a Potential Vehicle for Chemotherapeutic Agents and Radionuclides in the Management of Gynaecologic Neoplasms, <u>Am. J. Obstet. Gynecol.</u>, <u>139</u>, 877-
- DASHTI, N., WOLFBAUER, G., KOREN, E., KNOWLES, B., ALAUPOVIC, P., 1984, Catabolism of Human Low Density Lipoproteins by Human Hepatoma Cell Line Hep G2., <u>Biochim. Biophys. Acta</u>, <u>794</u>, 373-84
- 15. FIRESTONE, R.A., PISANO, J.M., FALCK, J.R., MCPHAUL, M.M., KREIGER. M., 1984, Selective Delivery of Cytotoxic Compounds to Cells by the LDL Pathway, <u>J. Med. Chem.</u>, <u>27</u>, 1037.
- 16. GAL, D., MACDONALD, P.C., PORTER, J.C., SMITH, J.W., SIMPSON, E.R. 1981, Effect of Cell Density and Confluency on Cholesterol Metabolism in Cancer Cells in Monolayer Culture, <u>Cancer Res</u>. 41, 473-77.
- 17. HALBERT, G.W., STUART, J.F.B., FLORENCE, A.T., 1985, A low-density Lipoprotein- Methotrexate Covalent Complex and its Activity Against L1210 Cells in vitro, <u>Cancer Chemother. Pharmacol</u>. <u>15</u> 223-27

- MOSLEY, S.T. GOLDSTEIN, J.L. BROWN, M.S. FALCK, J.R. ANDERSON, R.G.W.
 1981, Targeted Killing of Cultured Cells by Receptor-dependent
 Photosensitisation, Proc. Natl. Acad. Sci.U.S. 78, 5717-5721.
- 19. MOSLEY, S.T. YANG, Y.L. FALCK, J.R. ANDERSON, R.G.W. 1984, Receptor-mediated Delivery of Photoprotective Agents by Low Density Lipoprotein, Exp. Cell. Res. 155, 389-96.
- 20. RUDLING, M.J. COLLINS, V.P., PETERSON, C.O. 1983, Delivery of Aclacinomycin A to Human Glioma Cells in vitro by the Low Density Lipoprotein Pathway. <u>Cancer Res</u>. <u>43</u> 4600-05
- 21. VITOLS, S.G., MASQUELIER, M., PETERSON, C., 1985, Selective Uptake of a Toxic Lipophilic Anthracycline Derivative by the Low-density Lipoprotein Receptor Pathway in Cultured Fibroblasts. <u>J. Med. Chem</u>. <u>28</u>, 451-54
- 22. VITOLS. S.G., GAHRTON, G., PETERSON, C., 1984, Significance of the Low-density Lipoprotein (LDL) Receptor Pathway for the in vitro Accumulation of AD-32 Incorporated into LDL in Normal and Leukemic White Blood Cells. Cancer Treat. Reps. 68, 515-20
- 23. IWANIK. M.J., SHAW. K.V., LEDWITH. P.O.J., YANOVICH. S., SHAW. J.M. 1984, Preparation and Interaction of a Low-Density Lipoprotein: Daunomycin Complex with P388 Leukemic Cells. <u>Cancer Res.</u>, <u>44</u>, 1206-15.
- 24. VITOLS. S., GAHRTON. G., BJORHOLME. M., PETERSON, C., 1985 Hypocholesterolemia in Malignancy Due to Elevated LDL Receptor Activity in Tumour Cells; Evidence from Studies in Leukemic Patients. Lancet, II 1150-54.
- 25. GOLDSTEIN, J.L. BROWN, M.S., 1977, The Low Density Lipoprotein Pathway and its Relation to Atherosclerosis, <u>Annu.Rev. Biochem</u>. <u>46</u>, 897-930.
- 26. GOLDSTEIN, J.L., BROWN, M.S. 1977 Atherosclerosis: The Low-Density Lipoprotein Receptor Hypothesis. <u>Metabolism</u>. <u>26</u>, 1257-75

- 27. SHEN, B.W., SCANU, A.M., KEZDY, F.J., 1977, Structure of Human Serum Lipoproteins Inferred from Compositional Analysis. <u>Proc. Natl. Acad. Sci. U.S.A.</u> 74, 837-41.
- 28. GOLDSTEIN, J.L., ANDERSON, R.G.W. BROWN, M.S., 1979, Coated Pits, Coated Vesicles and Receptor-Mediated Endocytosis. <u>Nature</u> <u>279</u>, 679-685.
- 29. WILSON, J.D., 1972, The Role of Bile Acids in the Overall Regulation of Steroid Metabolism, Arch.Intern. Med. 130, 493-505.
- 30. KOVANEN, P.T., FAUST, J.R., BROWN, M.S., GOLDSTEIN, J.L., 1979, Low Density Lipoprotein Receptor in Bovine Adrenal Cortex I. Receptor-Mediated Uptake of Low Density Lipoprotein and Utilisation of its Cholesterol for Steroid Synthesis in Cultured Adrenocortical Cells. <u>Endocrinology</u>, <u>104</u>, 599.
- 31. KOVANEN, P.T., GOLDSTEIN, J.L. CHAPPELL, D.A., BROWN, M.S., 1980 Cholesterol Uptake in Mouse and Rat Adrenals. J. Biol Chem. 255 5591-98.
- 32. MYANT, N.B., 1982 Cholesterol Transport Through the Plasma, <u>Clin. Sci</u> <u>62</u>, 261-71.
- 33. NORUM, K. BERG, T., HELGERUD, P., DREVON, C.A., 1983, Transport of Cholesterol. <u>Physiological Reviews</u>, <u>63</u>, 1343-1419.
- 34. SCANU, A.M., 1972 Structural Studies on Serum Lipoproteins. <u>Biochim. Biophys. Acta.</u> <u>265</u> 471-508.
- 35. FREDRICKSON, D.S. LEVY, R.I., LEES, R.S., 1967. Fat Transport in Lipoproteins - an Integrated Approach to Mechanism and Disorders. N.Engl.J.Med <u>276</u> 32-44, 94-103, 148-56, 215-26, 273-81.
- 36. BILHEIMER, D., EISENBERG, S., LEVY, R.I., 1972. The Metabolism of Very Low Density Lipoproteins. I. Preliminary in vitro and in vivo Observations. <u>Biochim. Biophys. Acta.</u> <u>260</u> 212-21.
- 37. EISENBERG, S. 1980. Plasma Lipoproteins and Conversions: The Origin of High and Low Density Lipoproteins. <u>Ann. N.Y. Acad. Sci</u>. <u>348</u> 30-47

- MILLER, N.E., 1979, Plasma Lipoproteins, Lipid Transport and Atherosclerosis: Recent Developments. <u>J. Clin. Pathol. 32</u> 639-50.
- 39. ALAUPOVIC, P. 1971, Apo-Lipoproteins and Lipoproteins. <u>Atherosclerosis</u> <u>13</u> 141-6.
- BREWER, H.B., Hr., SCHAEFER, E.J., GREGG, R.E., OSBORNE, J.C.Jr. ZECH, L.A., 1982, Human Plasma Lipoproteins. In: Crepaldi G., GRETEN, H., SCHETTLER, G., BEGGIO, G., ed. <u>Lipoprotein Metabolism</u> <u>and Therapy of Lipid Disorders</u>. International Symposium, Florence, Excerpta Medica: Oxford, 3-11.
- 41. HAVEL, R.J., GOLDSTEIN, J.L., BROWN, M.S., 1980, In <u>Metabolic</u> <u>Control and Disease</u>. P.K. BONDY. L.E. ROSENBERG Eds. (Saunders, Philadelphia ed.8) p.393.
- 42. SCANU, A., LANDSBERGER, F.R. 1980, Review: Key Experiments in the Elucidation of the Lipoprotein Transport System. <u>Ann N.Y.</u> <u>Acad. Sci. 348</u> 1.
- 43. EISENBERG, S., 1979. Very Low Density Lipoprotein Metabolism. <u>Prog. Biochem. Pharmacol.</u> 15 139-65.
- 44. HAVEL, R.J., 1978, Metabolism of High Density Lipoprotein. Origin of HDL. In: GOTTO, Jr. A.M., MILLER, N.E., OLIVER, M.F. ed. <u>High</u> Density Lipoproteins and Atherosclerosis. Elsevier, Amsterdam.21-28.
- 45. EISENBERG, S., LEVY, R.I., 1975, Lipoprotein Metabolism, Adv. Lipid Res. <u>13</u> 1-89.
- REICHL, D., MYANT, N.B. BROWN, M.S., GOLDSTEIN, J.L., 1978
 Biologically Active Low Density Lipoproteins in Human Peripheral
 Lymph. <u>J. Clin. Invest.</u> <u>61</u> 64-71.
- 47. LA ROSA, J.C., LEVY, R.I., HERBERT, R., LUX, S.E., FREDRICKSON, D.S.,
 1970, A Specific Apoprotein Activator for Lipoprotein Lipase.
 Biochem. Biophys. Res. Commun. 41 57-62.
- 48. SHERRILL, B., INNERARITY, T.L., MAHLEY, R.W. 1980, Rapid Hepatic

Clearance of the Canine Lipoproteins Containing Only the

E Apoprotein by a High-affinity Receptor. J. Biochem 255 1804-7

- 49. SHELBURN, F., HANK, J. MEYER, W. QUARTFORD, S., 1980, Effect of Apoproteins on Hepatic Uptake of Triglyceride Emulsions in the Rat. <u>J. Clin. Invest. 65</u> 652-8.
- 50. MOK, H.Y.I., VON BERGMANN, K., GRUNDY, S.M. 1979, Effect of Continuous and Intermittent Feeding on Biliary Lipid Secretion in Man: Application for Measurements of Intestinal Absorption of Cholesterol and Bile Acids. J. Lipid Res. 20 389-398
- 51. GRUNDY, S.M., 1979 Cholesterol Metabolism in Man, <u>West J. Med</u>. <u>128</u> 13-25.
- 52. SIPERSTEIN, M.D. 1970, Regulation of Cholesterol Synthesis in Normal and Malignant Tissues. <u>Curr. top.cel</u>1. <u>2</u> 65-100
- 53. BROWN, M.S., GOLDSTEIN, J.L. 1980 Multivalent Feedback Regulation of HMG CoA Reductase, A Control Mechanism Coordinating Isoprenoid Synthesis and Cell Growth, <u>J. Lipid Res. 21</u> 505-17
- 54. KANE, J.P., HARDMAN, D.A., PAULUS, H.E., 1980 Heterogeneity of Apolipoprotein B: Isolation of a New Species From Chylomicrons <u>Proc. Natl. Acad. Sci.U.S.A.</u> 77 2645-9.
- 55. SIGURDSON, G., NICOLL, B., LEWIS, B. 1975 Conversion of Very Low Density Lipoprotein to Low Density Lipoprotein, <u>J. Clin. Invest</u>. 56 1481-90
- 56. TURNER, P.R., COLTART, J., HAZZARD, W.R., BACCHUS, R., NICOLL, A., MILLER, N.E., LEWIS, B. 1979 Production and Conversion of Lipoproteins: Trans-Splanchnic Arteriovenous Studies in Man, Eur. J. Clin. Invest. <u>9</u> 36(abstracts).
- 57. GLOMSET, J.A. NORUM, K.R. 1973, The Metabolic Role of Lecithin: Cholesterol Acyl Transferase, Perspectives from Pathology. <u>Adv. Lipid Res</u>. <u>11</u> 1-65.

Complex in Human Plasma, Proc. Natl. Acad. Sci.U.S.A. 77 3327-30

- 59. STANBURY, J.B., WYNGAARDEN, J.B., FREDRICKSON, D.S., Eds. 1978 <u>The Metabolic Basis of Inherited Disease</u>, McGraw-Hill New York Ed. 4, 544-65.
- 60 BROWN, M.S. FAUST, J.R., GOLDSTEIN, J.L., 1975 Role of the Low Density Lipoprotein Pathway in Regulating the Content of Free and Esterified Cholesterol in Human Fibroblasts, <u>J. Clin. Invest</u>. 55 783-93.
- 61. BROWN, M.S. GOLDSTEIN, J.L., 1979, Receptor-mediated Endocytosis: Insights from the Lipoprotein Receptor System. <u>Proc. Natl. Acad.</u> <u>Sci. U.S.A.</u> <u>76</u> 3330-37
- 62. BROWN, M.S., DANA, S.E., GOLDSTEIN, J.L., 1974, Regulation of 3-Hydroxy-3-Methylglutaryl CoA Reductase Activity in Cultured Human Fibroblasts: Comparison of Cells from a Normal subject and from a Patient With Homozygous Familial Hypercholesterolemia. J. Biol. Chem: 249 789.
- GOLDSTEIN, J.L., HELGESON, J.A., BROWN, M.S., 1979, Inhibition of Cholesterol Synthesis with Compactin Renders Growth of Cultured Cells Dependent on the Low Density Lipoprotein Receptor.
 J. Biol. Chem. 254 5403-09.
- 64. GOLDSTEIN, J.L., DANA, S.E., BROWN, M.S., 1974, Esterification of Low Density Lipoprotein Cholesterol in Human Fibroblasts and its Absence in Homozygous Familial Hypercholesterolemia. <u>Proc</u>. Natl. Acad. Sci. U.S.A. <u>71</u> 4288.
- 65. BROWN, M.S. GOLDSTEIN, J.L., 1975, Regulation of the Activity of the Low Density Lipoprotein Receptor in Human Fibroblasts. <u>Cell</u>
 6 307-316.
- 66. SLATER, H.R., PACKARD, C.J., SHEPHERD, J., 1982, Measurement of Receptor-independent Lipoprotein Catabolism Using 1,2 cyclohexanedione - modified Low Density Lipoprotein. J. Lipid Res. 23, 92-96.

- 67. PACKARD, C.J., SLATER, H.R., SHEPHERD, J., 1982, The Reticuloendothelial System and Low Density Lipoprotein Metabolism in the Rabbit. Biochim. Biophys. Acta. 712 412-19.
- 68. GOLDSTEIN, J.L. HO, Y.K., BASU, S.K., BROWN, M.S., 1979 Binding Site on Macrophages That Mediates Uptake and Degradation of Acetylated Low Density Lipoprotein, Producing Massive Cholesterol Deposition. <u>Proc. Natl. Acad. Sci. U.S.A</u>. <u>76</u> 333-37.
- 69. SCHNEIDER, W.J., BEISIEGEL, U., GOLDSTEIN, J.L., BROWN, M.S., 1982 Purification of the Low Density Lipoprotein Receptor, an Acidic Glycoprotein of 164,000 Molecular Weight, <u>J. Biol. Chem</u>. <u>257</u> 2664-2673.
- 70. YAMAMOTO, T., DAVID, C.G., BROWN, M.S., SCHNEIDER, W.J., CASEY, M.L., GOLDSTEIN, J.L., RUSSELL, D.W., 1984 The Human LDL Receptor: A Cysteine Rich Protein with Multiple Alu Sequences in its mRNA Cell 39 27-38.
- 71. MAHLEY, R.W. INNERARITY, T.L., 1983, Lipoprotein Receptors and Cholesterol Homeostasis, <u>Biochim. Biophys, Acta</u>, <u>737</u> 197-22.
- 72. CUMMINGS, R.D. CORNFELD, S., SCHNEIDER, W.J. et al, 1983, Biosynthesis of the N- and O- Linked Oligosaccharides of the Low Density Lipoprotein-Receptor, <u>J.Biol. Chem</u>. 258 15261-73.
- 73. TOLLESHAUG, H. GOLDSTEIN, J.L., SCHNEIDER, W.J. BROWN, M.S. 1982, Post-translational Processing of the LDL Receptor and its Genetic Disruption in Familial Hypercholesterolemia. <u>Cell</u> <u>30</u> 715-724
- 74. ANDERSON, R.G.W., BROWN, M.S., GOLDSTEIN, J.L. 1976, Localisation of Low Density Lipoprotein Receptors on Plasma Membrane of Normal Human Fibroblasts and their absence in Cells from a Familial Hypercholesterolemia Homozygote, <u>Proc.Natl.Acad.Sci.USA</u> <u>73</u> 2434-38.
- 75. PEARSE, B.M.F., 1976 Clathrin: A Unique Protein Associated With Intracellular Transfer of Membrane by Coated Vesicles. <u>Proc</u>. Natl. Acad. Sci. U.S.A. <u>73</u> 1255-59.

- 76. ANDERSON, R.G.W., BROWN, M.S., GOLDSTEIN, J.L. 1977, The Role of the Coated Endocytic Vesicle in the Uptake of Receptor-Bound Low Density Lipoprotein in Human Fibroblasts. <u>Cell</u> 10 351-64
- 77. BASU, S.K. GOLDSTEIN, J.L. ANDERSON, R.G.W., BROWN, M.S., 1976 Degradation of Cationised Low Density Lipoprotein and Regulation of Cholesterol Metabolism in Homozygous Familial Hypercholesterolemia Fibroblasts. <u>Proc. Natl. Acad. Sci. U.S.A.</u> 73 3178-82.
- 78. INNERARITY, T.L., MAHLEY, R.W., 1978 Enhanced Binding by Cultured Human Fibroblasts of Apo-E-Containing Lipoproteins as Compared with Low Density Lipoproteins. <u>Biochemistry</u> <u>17</u> 1440-47.
- 79. BERSOT, T.P., MAHLEY, R.W. BROWN, M.S. GOLDSTEIN, J.L. 1976, Interaction of Swine Lipoproteins with the Low Density Lipoprotein Receptor in Human Fibroblasts. <u>J. Biol. Chem</u> <u>251</u> 2395-98.
- 80. PITAS, R.E., INNERARITY, T.L., MAHLEY, R.W., 1980, Cell Surface Receptor Binding of Phospholipid-Protein Complexes Containing Different Ratios of Receptor-Active and -Inactive E Apoprotein. J. Biol. Chem. <u>255</u> 5454-60.
- 81. REDGRAVE, T.G. 1970 Formation of Cholesteryl Ester-Rich Particulate Lipid During Metabolism of Chylomicrons. J. Clin. Invest. <u>49</u> 465-71.
- 82. BRESLOW, J.L. LOTHROP, D.A., CLOWES, A.W., LUX, E. 1977, Lipoprotein Regulation of 3-Hydroxy-3-Methyl Glutaryl Coenzyme A Reductase Activity in Rat Liver Cell Cultures. <u>J. Biol. Chem</u>. 252 2726-33.
- 83. MAHLEY,R.W., INNERARITY, T.L. WEISGRABER, K.H. OH, S.Y. 1979, Altered Metabolism (in vivo and in vitro) of Plasma Lipoproteins After Selective Chemical Modification of Lysine Residues of the Apoproteins. <u>J. Clin. Invest</u>. <u>64</u> 743.
- 84. GREGG, R.E. ZECH, L.A. SCHAEFER, E.T., BREWER, H.B. Jr., 1981, Type 111 Hyperlipoproteinemia: Defective Metabolism of An Abnormal

Apolipoprotein E. Science 211 584.

- 85. REICHL, D. MYANT, N.B. PFLUG, J.J. 1977 Concentrations of Lipoproteins Containing Apolipoprotein B in Human Peripheral Lymph. Biochim. Biophys, Acta. 489 98-105.
- 86. THOMAS, M.S. RUDEL, L.L., 1983, ³H -Cholesteryl Ester Labelling and Transfer Among Human and Non-human Primate Plasma Lipoproteins. <u>Analytical Biochem</u> <u>130</u> 215-22.
- 87. SHEN, M.M., KRAUSS, R.M. LINDGREN, F.T., FORTE, T.M. 1981, Heterogeneity of Serum Low Density Lipoproteins in Normal Human Subjects. <u>J. Lipid Res</u>. <u>22</u> 236-44.
- MATTHEWS, C.M.E. 1957 The Theory of Tracer Experiments with
 ¹³¹I-labelled Plasma Lipoproteins. <u>Phys. Med. Biol.</u> <u>2</u> 36-53.
- 89. PACKARD, C.J., THIRD, J.L.H.C., SHEPHERD, J., et al, 1976, Low Density Lipoprotein Metabolism in a Family of Familial Hypercholesterolemic patients. <u>Metabolism</u> <u>25</u> 995-1006.
- 90. MAHLEY, R.W., INNERARITY, T.L. PITAS, R.E., et al 1977, Inhibition of Lipoprotein Binding to Cell Surface Receptors of Fibroblasts Following Selective Modification of Arginine-rich and B Apoproteins. J. Biol. Chem. 252 7279-87.
- 91. MAHLEY, R.W. WEISGRABER, K.R. MELCHOIR, G.W., et al, 1980, Inhibition of Receptor-mediated Clearance of Lys and Arg-modified Lipoprotein from Plasma of Rats and Monkeys. <u>P.N.A.S.</u> <u>71</u> 255-9.
- 92. MAHLEY,R.W., INNERARITY, T.L., WEISGRABER, K.H., 1980, Alterations in Metabolic Activity of Plasma Lipoproteins Following Selective Chemical Modification of the Apoproteins., <u>Ann.N.Y. Acad.Sci.</u> 348 265-80.
- 93. SHEPHERD, J., BICKER, S., LORIMER, A.R., PACKARD, C.J., 1979, Receptor-mediated Low Density Lipoprotein Catabolism in Man, <u>J.Lipid Res.</u>, <u>20</u>, 999-1006.

145

the Effects of Dietary Cholesterol on Plasma Lipoproteins and Cellular Cholesterol Homeostasis in Man, <u>J. Clin. Invest.</u>, <u>67</u>, 493-502.

- 95. MAHLEY, R.W., WEISGRABER, K.H., INNERARITY, T.L., WINDMUELLER, H.G., 1979, Accelerated Clearance of Low-density and High-density Lipoproteins and Retarded Clearance of E Apoprotein-containing Lipoproteins from the Plasma of Rats After Modification of Lysine Residues, Proc. Natl. Acad. Sci. U.S.A., 76,1746-50.
- 96. KESANIEMI, Y.A., GRUNDY, S.M., 1982, Significance of Low Density Lipoprotein Production in the Regulation of Plasma Cholesterol Levels in Man, <u>J. Clin. Invest.</u>, 70, 13-21.
- 97. SLATER, H.R., PACKARD, C.J., BICKER, S., SHEPHERD, J., 1980, The Effects of Cholestyramine on Receptor Mediated Plasma Clearance and Tissue Uptake of Human Low Density Lipoproteins in the Rabbit, <u>J. Biol. Chem.</u>, <u>255</u>, 10210-13.
- 98. PATTHY, L., SMITH, E.L., 1975, Reversible Modification of Arginine Residues. Application to Sequence Studies by Restriction of Trypsin Hydrolysis to Lysine Residues, <u>J. Biol. Chem</u>. <u>250</u>, 557-64.
- 99. BROWN, M.S., KOVANEN, P.T., GOLDSTEIN, J.L., 1981, Regulation of Plasma Cholesterol by Lipoprotein Receptors, <u>Science</u>, <u>212</u>, 628-35.
- 100. WEISGRABER, K.H., INNERARITY,T.L., MAHLEY, R.W., 1978, Role of Lysine Residues of Plasma Lipoproteins in High Affinity Binding to Cell Surface Receptors on Human Fibroblasts, <u>J. Biol. Chem.</u>, <u>253</u>, 9053-62.
- 101. BILHEIMER, D.W., WATANABE, Y.,KITA, T., 1982, Impaired Receptor-Mediated Catabolism of Low Density Lipoprotein in the WHHL Rabbit, an Animal Model of Familial Hypercholesterolemia, <u>Proc. Natl. Acad.</u> Sci. U.S.A., <u>79</u>, 3305-09.
- 102. SHEPHERD, J., PACKARD, C.J., BICKER, S., 1980, Cholestyramine Promotes Receptor-mediated Low-density Lipoprotein Catabolism in

Man, N. Engl. J. Med., 302, 1219-22

- 103. THOMPSON, G.R., SOUTAR, A.R., SPENGEL, et al, 1981, Defects of Receptor-mediated Low density Lipoprotein Catabolism in Homozygous Familial Hypercholesterolemia and Hyperthyroidism in vivo, <u>Proc. Natl. Acad. Sci. U.S.A.</u>, 78, 2591-95.
- 104. PITTMAN, R.C., ATTIE, A.D., CAREW, T.E., STEINBERG, D., 1979, Tissue Sites of Degradation of Low Density Lipoprotein: Application of a Method for Determining the Fate of Plasma Proteins, <u>Proc.</u> <u>Natl. Acad. Sci. U.S.A.</u>, <u>76</u>, 5345-49.
- 105. PITTMAN, R.C., CAREW, T.E., GLASS, C.K., et al, 1983, A Radioiodinated, Intracellularly Trapped Ligand for Determining the Sites of Plasma Protein Degradation in vivo, <u>Biochem. J.</u>, <u>212</u>, 791-800.
- 106. STEIN, Y., HALPERIN, G., STEIN, O., 1981, The Fate of Cholesterol Linoleoyl Ether and Cholesterol Linoleate in the Intact Rat After Injection of Biologically Labelled Human Low Density Lipoprotein, Biochim. Biophys. Acta., 663, 569-74.
- 107. CAREW, T.E. FORAN, W.A. STEINBERG, D., 1980 Tissue Sites of Degradation of Unmodified and Reductively Modified Low Density Lipoprotein in the Rat. <u>Circulation, 62</u> (Suppl.<u>11</u>) 111-17.
- 108. ABRAMSON, Z.H. KARK, J.D. 1985, Serum Cholesterol and Primary Brain Tumours: A Case Control Study. <u>Br.J. Cancer</u> <u>52</u> 93-98.
- 109. CHAO, F.C. EFRON, P., WOLF, P., 1975, The Possible Prognostic Usefulness of Assessing Serum Proteins and Cholesterol in Malignancy, Cancer, <u>35</u>, 1223-29.
- 110. MULLER, H.G. 1930, The Cholesterol Metabolism in Health and Anemia. <u>Medicine 9</u> 119-74.
- 111. FIENLIEB, M., 1983, Review of the Edipemiological Evidence for A Possible Relationship Between Hypocholesterolemia and Cancer. <u>Cancer Res.</u> <u>43</u> 2503-07.

112. BASES, G., SHIPLEY, M.J. 1965 Enhanced Reticuloendothelial Phagocytic Activity in Myeloproliferative Diseases.

J. Reticuloendothelial Soc. 2 1-7.

- 113. ROSE, G., SHIPLEY, M.J. 1980 Plasma Lipids and Mortality: A Source of Error. Lancet 1 523-26.
- 114. INTERNATIONAL COLLABORATIVE GROUP, 1982, Circulating
 Cholesterol Levels and Risk of Death from Cancer in men aged
 40 to 69 Years. Experience of an International Collaborative
 Group. J. Am. Med. Assoc. 248 2853-59.
- 115. GINSBERG,H., GILBERT, H.S., GIBSON, J.C. LE, N.A., 1982, Increased Low-density-lipoprotein Catabolism in Myeloproliferative Disorders. Ann.Int.Med. 96 311-16.
- 116. GILBERT, H.S., GINSBERG, H., 1983 Hypocholesterolemia As a Manifestation of Disease Activity in Chronic Myelocytic Leukemia. Cancer 51 1428-33.
- 117. HAVEKES, L., VAN HINSBERG, V., KEMPEN, H.J. EMEIS, J., 1983 The Metabolism in vitro of Human Low Density Lipoprotein by the Human Hepatoma Cell Line Hep G2. <u>Biochem J</u>. 214 951-958.
- 118. FREEMAN, D.A., ASCOLI, M. 1983, The Low Density Lipoprotein Pathway of Cultured Leydig Tumour Cells. Utilisation of Low-Density Lipoprotein-derived Cholesterol for Steroidogenesis. Biochim. Biophys Acta. 754 72-81.
- 119. VITOLS,S., GAHRTON, G., OST, A. PETERSON, C., 1984, Elevated Low Density Lipoprotein Receptor Activity in Leukemic Cells with Monocytic Differentiation. <u>Blood</u> <u>63</u> 1186-93.
- 120. LINDQUIST, R., VITOLS, S., GAHRTON, G. OST, A., PETERSON, C., 1985 Low Density Lipoprotein Receptor Activity on Human Leukemic Cells-Relation to Chromosome Aberrations. <u>Acta. Med. Scand.</u> <u>217</u> 553-8.
- 121. HOTTENDORF, G.H., 1985 Toxic Factors Limiting the Use of Anticancer Agents, Chemioterapia <u>4</u> 383-7.

- 122. TANNOCK, I.F., 1986, Experimental Chemotherapy and Concepts Related to the Cell Cycle, Int. J. Radiat. Biol. <u>49</u> 335-55.
- 123. LEE, YTN, CHAN, K.K., HARRIS, P.A., COHEN, T.L., Distribution of Adriamycin in Cancer Patients, Tissue Uptakes, Plasma Concentrations After IV and Hepatic IA Administration, 1980, <u>Cancer</u> <u>45</u> 2231-39
- 124. GOLDMAN, I.D., LICHTENSTEIN, N.S., OLVEIRO, V.T., 1968, Carrier Mediated Transport of the Folic Acid Analogue, Methotrexate, in the L1210 Leukemia Cell, J. Biol. Chem. 243 5007-17.
- 125. KRISHAM, A., FREI, E., 1976, Adriamycin in the Cell Cycle, Traverse and Kinetics of Cultured Human Lymphoblasts, Cancer Res. 36 143-50.
- 126. TESTA, N.G., HENDRY, J.H., MOLINEUX, G., 1985, Long Term Bone Marrow Damage in Experimental Systems and in Patients After Radiation or Chemotherapy, <u>Anticancer Res</u>. <u>5</u> 101-10.
- 127. LOHRMANN, H.P., SCHREML, W., LANG, M., et al,1978, Changes in Granulopoiesis During and After Adjuvant Chemotherapy of Breast Cancer, <u>Br. J. Haemat</u>. <u>40</u> 369-86
- 128. PIZZO, P.A., 1984, Granulocytopenia and Cancer Therapy, Past Problems, Current Solutions, Future Challenges, <u>Cancer</u> <u>54</u> 2649-61
- 129. REILLY, J.T., NEIFELD, J.P., ROSENBERG, S.A., 1977, Clinical Course and Management of Accidental Adriamycin Extravasation, Cancer 40 2053-56.
- 130. BACHUR, N.R., GORDON, S.L., GEE, M.V., KON, H., 1979, NADPH Cytochrome P-450 Reductase Activation of Quinone Anticancer Agents to Free Radicals, Proc. Natl. Acad. Sci. U.S.A., <u>76</u> 954-57
- 131. BROCK, N., 1967, Pharmacologic Characterisation of Cyclophosphamide and Cyclophosphamide Metabolites, <u>Cancer Chemother. Rep.</u> <u>51</u> 315-25.
- 132. CHABNER, B.A., MYERS, C.E., COLEMAN, C.N., JOHNS, D.G., 1975, The Clinical Pharmacology of Antineoplastic Agents, <u>N. Engl. J. Med.</u> <u>292</u> 1107-13 & 1159-68.

149

133. WOLPERT, M.K., DAMLE, S.P., BROWN, J.E., et al, 1971,

The Role of Phosphohydrolases in the Mechanism of Resistance

of Neoplastic Cells to 6-thiopurines. Cancer Res. 31 1620-26.

134. MENARD, D.B. GISSELBRECHT, C., MARTY, M. et al, 1980.

Antineoplastic Agents and the Liver. Gastro 78 142.

- 135. CHAN, K.K. CHLEBOWSKI, R.T. TONG, M., et al, 1980, Clinical Pharmacokinetics of Adriamycin in Hepatoma Patients with Cirrhosis, Cancer Res. 40 1263-68.
- 136. RAYMOND, J.R. 1984 Nephrotoxicities of Antineoplastic and Immunosuppressive Agents, <u>Curr. Probl. Cancer</u> 8 1-32.
- 137. CONNORS, T.A. 1974 Mechanisms of Clinical Drug Resistance. <u>Biochem Pharmac</u>. <u>23</u> Suppl. 2. 89-100.
- 138. KAYE, S.B., MERRY, S., 1985 Tumour Cell Resistance to Anthracyclines - A review. <u>Cancer Chemother. Pharmacol</u>. <u>14</u> 96-103.
- 139. EHRLICH, P. 1906, <u>Collected Studies on Immunology</u>, Wiley N.Y. 11 442-447.
- 140. RUBENS, R.D., DULBECCO, R., 1974, Augmentation of Cytotoxic drug Action by Antibodies Directed at Cell Surface. <u>Nature</u> 248 81-2.
- 141. LEVY, R., HURWITZ, E., MARON, R., et al 1975, The Specific Cytotoxic Effects of Daunomycin Conjugated to Antitumour Antibodies. Cancer Res. 35 1182-86.
- 142. BERNIER, L.G. PAGE, M. GAUDREAULT, R.C. JOLY, L.P. 1984
 A Chlorambucil Anti-CEA Conjugate Cytotoxic for Human Colon
 Adenocarcinoma Cells in vitro. <u>Br.J. Cancer</u> <u>49</u> 245-6.
- 143. CORNU, G., MICHAUX, J.L., SOKAL, G., TROUET, A. 1974, Daunorubicin-DNA: Further Clinical Trials in Acute Non-lymphoblastic Leukemia, <u>Eur. J. Cancer</u> <u>10</u> 695-700.
- 144. BARBANTI-BRODANO, G., FIUME, L., 1973, Selective Killing of Macrophages By Amantin-Albumin Conjugates, <u>Nature</u> 243 281-83.

- 145. GREGORIADIS, G., DAVISSON, P., SCOTT, S., 1977, Binding of Drugs to Liposome-entrapped Macromolecules Prevents Diffusion of Drugs from Liposomes in vitro and in vivo. <u>Biochem. Soc. Trans.</u> <u>5</u> 1323-26.
- 146. LESERMAN, L.D., MACHY, P., BARBET, J., 1981 Cell Specific Drug Transfer from Liposomes Bearing Monoclonal Antibodies. <u>Nature</u> 293, 226-28.
- 147. TOMLINSON, E., BURGER, J.J. SCHOONDERWOERD, E.M.A. et al, 1982 Preparation and Characterisation of Albumin Microspheres for Intraarterial Tumour Targeting of Cytotoxic Compounds, J. Pharm. Pharmacol 34 88.
- 148. TOMLINSON, E., 1983, Microsphere Delivery Systems for Drug Targeting and Controlled Release, <u>Int. J. Pharm. Tech. & Prod.Mfr</u>. <u>4</u> 49-57.
- 149. WIDDER, K.J. MARIONO, P.A., MORRIS, R.M. et al, 1983, Selective Targeting of Magnetic Albumin Microspheres to the Yoshida Ascites Sarcoma: Ultrastructural Evaluation of Microsphere Disposition, Europ, J. Cancer Clin. Oncol. 19 141-8
- 150. SENYEI, A.E., REICH, S.D. GONCZY, E., et al 1981, In vivo Kinetics of Magnetically Targeted Low-dose Doxorubicin. <u>J. Pharm. Sci.</u> <u>70</u> 389-91.
- 151. RUDMAN, D. HOLLINS, B., BIXLER, T.J.II, MOSTELLER, R.C. 1972 Transport of Drugs, Hormones, and Fatty Acids in Lipemic Serum. J. Pharmacol. Exp. Ther. 180 797-810.
- 152. HYNDS, S.A. WELSH J. STEWART J.M., et al 1984, Low-Density Lipoprotein Metabolism in Mice with Soft Tissue Tumours, <u>Biochim</u> <u>Biophys, Acta</u>, <u>795</u> 589-95.
- 153. NORATA, G., CANTI,G. RICCI, L., et al, 1984, In vivo Assimilation of Low Density Lipoproteins by a Fibrosarcoma Tumour Line in Mice, <u>Cancer Letters</u> <u>25</u> 203-208.

154. DRESEL, H.A. FRIEDRICH, E. VIA, D.P. et al, 1985.

Characterisation of Binding Sites for Acetylated Low Density Lipoprotein in the Rat Liver in vivo and in vitro. <u>EMBO Journal</u> 4 1157-62.

- 155. COUNSELL, R.E. POHLAND, R.C., 1982, Lipoproteins as Potential Site-Specific Delivery systems for Diagnostic and Therapeutic Agents. J. Med. Chem. 25 1115-1120.
- 156. KREIGER, M., McPHAUL, M.J. GOLDSTEIN, J.L., BROWN, M.S., 1979 Replacement of Neutral Lipids of Low Density Lipoprotein with Esters of Long Chain Unsaturated Fatty Acids. J. Biol. Chem. <u>254</u> 3845-53.
- 157. REMSEN, J.F., SHIREMAN, R.B., 1981. Effect of Low Density Lipoprotein on the Incorporation of Benzo (a) Pyrene by Cultured Cells. <u>Cancer Res.</u> <u>41</u> 3179-85.
- 158. YANOVICH, S., PRESTON, L., SHAW, J.M. 1984, Characteristics of Uptake and Cytotoxicity of a Low-Density Lipoprotein-Daunomycin Complex in P388 Leukemic Cells. <u>Cancer Res</u>. <u>44</u> 3377-82.
- 159. ARNOLD, L.J. DAGEN, H., KAPLAN, N.O., 1983, Poly-(L-Lysine) as an Antineoplastic Agent and Tumour-Specific Drug Carrier. In: Goldberg E.P. (ed) <u>Targeted Drugs</u> P89. Wiley, N.Y.
- 160. KREIGER, M. BROWN, M.S. FAUST, J.R., GOLDSTEIN, J.L. 1978, Replacement of Endogenous Cholesteryl Esters of Low Density Lipoprotein with Exogenous Cholesteryl Linoleate. J. Biol. Chem. 253 4093-4101.
- 161. MASQUELIER, M. VITOLS, S., PETERSON, C., 1986, Low Density Lipoprotein (LDL) as a Carrier of Antitumoral Drugs: in vivo Fate of Drug-LDL Complexes in Mice. In Press.
- 162. GILMAN, A. 1963. The Initial Clinical Trial of Nitrogen Mustard. Am. J. Surg. 105 574-78.

the Active Intermediates Formed in the Microsomal Metabolism of Cyclophosphamide and Isophosphamide, <u>Biochem. Pharmacol</u>, <u>23</u> 115-29.

- 164. CALABRESI, P., PARKS, R.E. Jr., Alkylating Agents, Antimetabolites, Hormones and Other Antiproliferative Agents in <u>The Pharmacological</u> <u>Basis of Therapeutics</u>, Ed. L.S. Goodman and A. Gilman, McMillan, N.Y.
- 165. WHEELER, G.P., 1967, Symposium on Immunosuppressive Drugs. Some Biochemical Effects of Alkylating Agents, <u>Proc. Fedn. Am.</u> <u>Socs. Exp. Biol. 26</u> 885-92.
- 166. MARTINDALE J.R. <u>The Extra Pharmacopoeia</u>, 28th Edn. Ed. J.E.F. Reynolds, Pharmaceutical Press London, P.171-8.
- 167. ALBERTS, D.S., CHANG, S.Y. CHEN. H.S.G. et al, 1979 Pharmacokinetics and Metabolism of Chlorambucil in Man: A Preliminary Report. <u>Cancer Treat. Revs.</u> <u>6</u> Suppl. 9-17.
- 168. MCLEAN, A. WOODS, R.L. CATOVSKY, D., FARMER, P. 1979. Pharmacokinetics and Metabolism of Chlorambucil in Patients With Malignant Disease. <u>Cancer Treat. Revs</u>. <u>6</u> Suppl.33.
- 169. DUBOST, M., GANTER, P., MARAL, R., et al 1963, Un Nouvel Antibiotique a proprietes antitumorale. C.r. hedb.Seanc. Acad. Sci. Paris 257 1813-20.
- 170. DIMARCO, A., SOLDATI, M., FIORETTI, A., DASDIA, T., 1963 Richerche sull'attivita' della daunomycina su cellule normali ie neoplastiche coltivate in vitro. <u>Tumori</u> <u>49</u> 235-51.
- 171. ARCAMONE, F., CASSINELLI,.G., FANTINI, G., et al, 1969.
 Adriamycin, 14-Hydroxy-Daunomycin, a New Tumour Antibiotic from
 S. Peucetius var. Caesius. <u>Biotechnol. Bioeng</u>. 11 1101-09.
- 171a ISLAM, S.A. NEIDLE, S., GANDECHA, B.M. et al, 1985. Comparative Computer Graphics and Solution Studies of the DNA Interaction of Substituted Anthraquinones Based on Doxorubicin and Mitoxantrone.

J. Med. Chem. 28 857-64.

- 172. RUSCONI, A., CALENDI, E., 1966, Action of Daunomycin on Nucleic Acid Metabolism. <u>Biochim. Biophys. Acta</u>. <u>119</u> 413-15.
- 173. PIGRAM, W.J., FULLER, W., HAMILTON, L.D., 1972. Stereochemistry of Intercalation of Daunomycin with DNA. Nat. New Biol. 235 17-19.
- 174. BLANCHARD, J.C. SCHNEIDER, Y.T. BAURAIN, R., TROUET, A., 1981 Accumulation, Metabolism and Subcellular Localisation of Daunomycin, Doxorubicin and Their DNA Complexes in Rat Heart Ventricles. <u>Eur. J. Cancer</u> <u>17</u> 297-305.
- 175. LINDEN, W.A., BALSCH, H., VON CANSTEIN, L., et al 1974, Impulse-cytophotometric Studies on the Effects of Daunomycin, Eur. J. Cancer, 10 647-51.
- 176. HUFFMAN, D.H., BENJAMIN, R.S., BACHUR, N.R. 1972. Daunorubicin Metabolism in Acute Nonlymphocytic Leukemia. <u>Clin. Pharmac. Ther</u>. <u>13</u> 895-905.
- 177. BENDER, R.A., ZWELLING, L.A. DOROSHOW, J.H. et al 1978. Antineoplastic Drugs: Clinical Pharmacology and Therapeutic use. Drugs, <u>16</u> 46-87.
- 178. ALBERTS, D.S. BACHUR, N.R. HOLTZMAN, J.L. 1971. The Pharmacokinetics of Daunomycin in Man. Clin. Pharm. Ther. <u>12</u> 96.
- 179 BACHUR, N.R. RIGGS, C.E., GREEN, M.R. 1977, Plasma Adriamycin and Daunorubicin Levels by Fluorescence and Radioimmunoassay. Clin. Pharmac. Ther. <u>21</u> 70-77.
- 180. MERKER, P.C., LEWIS, M.R., WALKER, M.D., RICHARDSON, E.P. Jr. 1978. Neurotoxicity of Adriamycin (Doxorubicin) Perfused Through The Cerebrospinal Fluid Spaces of the Rhesus Monkey. <u>Toxic Appl.</u> Pharmacol <u>44</u> 191-205.
- 181. JOHNSON, R.K. CHITNIS, M.P. EMBREY, E.M.et al 1978. In vivo Characteristics of Resistance and Cross-resistance of an

Adriamycin-resistant subline of P388 Leukemia. <u>Cancer Treat.Rep</u>. 62 1535-47.

- 182. KAYE, S.B., BODEN, B.E., RYMAN, B.E., 1981. The Effect of Liposome (Phospholipid Vesicle) Entrapment of Actinomycin D and Methotrexate on the In vivo Treatment of Sensitive and Resistant Solid Murine Tumours. Europ. J. Cancer 17 279-89.
- 183. DANO, K, 1973 Active Outward Transport of Daunomycin in Resistant Ehrlich Ascites Tumour Cells. <u>Biochim, Biophys. Acta</u>. 323 466-83.
- 184. INABA, M., JOHNSON, R.K., 1978. Uptake and Retention of Adriamycin and Daunorubicin by Sensitive and Anthracycline-resistant sublines of P388 Leukemia. Biochem Pharmacol. 27 2123-30.
- 185. PATSCH, J.R. SAILER, E. KOSTNER, G., et al 1974. Separation of the Main Lipoprotein Classes from Human Plasma by Rate-Zonal Ultracentrifugation. J. Lipid Res. 15 356
- 186. HAVEL, R.J., EDER, H.A., BRADDON, J.H., 1955 The Distribution and Chemical Composition of Ultracentrifugally Separated Lipoproteins in Human Serum. J. Clin. Invest. <u>34</u> 1345-53.
- 187. MCFARLANE, A.S. 1958 Efficient Trace-Labelling of Proteins with Iodine. Nature (London) <u>182</u> 53.
- 188. LOWRY, O.H. ROSENBROUGH, N.J., FARR, A.L. RANDALL, R.J. 1951. Protein Measurement with the Folin Phenol Reagent. <u>J. Biol. Chem</u>. 193 265-75.
- 189. HALPERIN, G., GATT, S., 1980 The Synthesis of Cholesteryl Alkyl Ethers. Steroids 35 39-42.
- 190. STOFFEL, W., SALM, K.P., LANGER, M., 1978. A New Method for the Exchange of Lipid Classes of Human Serum HDL. <u>Hoppe-Seylers</u> Z. Physiol. Chem. 395 1385-93.
- 191. CHAJEK, T., FIELDING, C.J. 1978. Isolation and Characterisation of a Human Serum Cholesteryl Ester Transfer Protein.

Proc. Natl. Acad. Sci. U.S.A. 75 3445-9.

- 192. LAMEY, P.J., WATERHOUSE, J.P., FERGUSON, M.M., 1982. Virally Induced Pleomorphic Salivary Adenoma in the CFLP Mouse. <u>Am. J. Pathol. 109</u> 129-32.
- 193. CARACHI, R., BUSUTTIL, A., JOFFE, S.N. et al, 1980. Multiple
 Polyps and Colonic Neoplasia Induced in the Rat. J. Surgl. Res.
 29 363-370.
- 194. POUR, P., ALTHOFF, J., TAKAHASHI, M. 1977, Early Lesions of Pancreatic Ductal Carcinoma in the Hamster Model. <u>Am. J.Pathol.</u> 88 291-308.
- 195. ANGELIN, B. RAVIOLA, C.A., INNERARITY, T.L. MAHLEY, R.W., 1983
 Regulation of Hepatic Lipoprotein Receptors in the Dog.
 J. Clin. Invest. 71 816-31.
- 196. MAHLEY, R.W., HUI, D.Y., INNERARITY, T.L. et al 1981
 Two Independent Lipoprotein Receptors on Hepatic Membranes of
 Dog, Swine and Man. APO-B,E and APO-E Receptors. J. Clin.Invest.
 68 1197.
- 197. PREIS, J.M. GUSTAFSON, A,, WEIGAND, D. DUANE, W.C. 1983. Taurocholate is More Potent Than Cholate in Suppression of Bile Salt Synthesis in the Rat. J. Lipid Res. <u>24</u> 141-6.
- 198. ALBERS, J.J. TOLLEFSON, J.H., CHEN, C.H. STEINMETZ, A., 1984. Isolation and Characterisation of Human Plasma Lipid Transfer Proteins. Arteriosclerosis <u>4</u> 49-58.
- 199. ZECHNER, R., DIEPLINGER, H., ROSCHER, A., KOSTNER, G.M. 1984. The Low-Density Lipoprotein Pathway of Native and Chemically Modified Low-Density Lipoproteins in vitro. <u>Biochem J. 224</u> 569-76.
- 200. CRAIG, I.C. VIA. D.P. SHERRILL, B.C., SKLAR, L.A. et al, 1982. Incorporation of Defined Cholesteryl Esters into Lipoproteins Using Cholesteryl-Ester-rich Microemulsions. <u>J. Biol.Chem</u>. 251 330-35.
- 201. PARKS, J.S., MARTIN, J.A., JOHNSON, F.L., RUDEL, L.L. 1985. Fusion of Low Density Lipoproteins with Cholesterol-Ester-Phospholipid Microemulsions. <u>J. Biol.Chem</u>. <u>260</u> 3155-63.
- 202. BROWN, M.S., DANA, S.E., GOLDSTEIN, J.L., 1975. Receptordependent Hydrolysis of Cholesteryl Esters Contained in Plasma Low Density Lipoprotein. <u>Proc. Natl. Acad. Sci. U.S.A</u>. <u>72</u> 2925-29
- 203. TUCKER, I.G., FLORENCE, A.T. 1983. Interactions of Ionic and Non-Ionic Surfactants with Plasma Low Density Lipoprotein. J. Pharm. Pharmacol. 35 705-711.
- 204. EHRSSON, H., EKSBORG, S., WALLIN, I. et al 1980. Degradation of Chlorambucil in aqueous solution. <u>J. Pharm. Sci</u>. <u>69</u> 1091-94.
 205. WATSON, I.D., STEWART, M.J. FARID, Y.Y.Z. 1985. The Effect of Surfactants on the High-performance Liquid Chromatography of
- 206. EISENBERG, S., BILHEIMER, D.W., LEVY, R.I. et al, 1973. On the Metabolic Conversion of Human Very Low Density Lipoprotein. Biochim. Biophys, Acta. 226 361-67.

Anthracyclines. J. Pharm. Biomed. Anal. 3 555-63.

- 207. MORGAN, D., FRESHNEY, R.I. DARLING, J.L. et al, 1983. Assay of Anticancer Drugs in Tissue Culture: Cell Cultures of Biopsies from Human Astrocytoma. <u>Br. J. Cancer</u>, 47 205-214.
- 208. FRESHNEY, R.I., PAUL, J. KANE, I.M., 1975. Assay of Anti-cancer Drugs in Tissue Culture; Conditions affecting their ability to incorporate ³H-leucine after drug treatment. <u>Br. J. Cancer</u> 31 89-99.
- 209. SALMON, S.E. HAMBURGER A.W. SOEHNLEN, B.S. et al, 1978. Quantitation of Differential Sensitivity of Human-tumour Stem Cells to Anti-cancer drugs. <u>N.Eng. J. Med.</u> <u>298</u> 1321.

210. YUHAS, J.M. LI, A.P., MARTINEZ, A.O. LODMAN, A.J. 1977,

A Simplified Method for Production and Growth of Multicellular Tumour Spheroids. <u>Cancer Res. 37</u> 3634-43.

- 211. TWENTYMAN.P.R., 1982. Growth Delay in Small EMT6 Spheroids Induced by Cytotoxic Drugs and its Modification by Misonidazole Pretreatment Under Hypoxic Conditions. Br.J. Cancer 45 565-70.
- 212. BOGDEN, A.E., COBB, W.R., LEPAGE, D.G., et al. 1981. Chemotherapy Responsiveness of Human Tumours as First Transplant Generation Xenografts in the Normal Mouse: Six-day subrenal Capsule Assay. <u>Cancer</u> 48 10-20.
- 213. GOLDSTEIN, J.L. BROWN M.S., 1974. Binding and Degradation of Low Density Lipoproteins by Cultured Human Fibroblasts. Comparison of Cells from a Normal subject and from a Patient with Homozygous Familial Hypercholesterolemia. <u>J. Biol. Chem. 249</u>, 5153-62.
- 214. CUNNINGHAM, D. JACK, A., MCMURDO, D.F.S. et al, 1986. The 6-day Subrenal Capsule Assay is of no Value with Primary Surgical Explants from Gastric Cancer. <u>Br. J. Cancer</u>. In press.
- 215. TALL, A.R., SMALL, D.M., 1980. Body Cholesterol Removal: Role of High Density Lipoproteins. Adv. Lipid Res. 17 2-51.
- 216. TROUET, A. DEPREZ-DE CAMPENEERE, D. MALENGREAUX, M.D.S., et al 1974. Experimental Leukemia Chemotherapy with a "Lysosomotropic" Adriamycin-DNA Complex. <u>Europ. J. Cancer</u>. <u>10</u> 405-411.
- 217. LYONS, R.M. GOLDENBERG, G.J., 1972 Active Transport of Nitrogen Mustard and Choline by Normal and Leukemic Human Lymphoid Cells. Cancer Res. 32 1679-85.
- 218. HILL, B.T. 1975. Differential Effects of Cytotoxic Concentration of Chlorambucil on Drug-Resistant and Drug-Sensitive Cell Lines of the Yoshida Ascites Sarcoma In Culture. <u>Europ. J. Cancer</u> <u>11</u> 9-16.
- 219. HILL, B.T., JARMAN, M., HARRAP, K.R. 1971 Selectivity of Action of Alkylating Agents and Drug Resistance. 4. Synthesis of Tritium-Labelled Chlorambucil and a Study of its Cellular Uptake

158

By Drug-Sensitive and Drug-Resistant Strains of the Yoshida Ascites Sarcoma in Vitro. J. Med. Chem. 14 614-18.

- 220. LEVI, F.A., BLUM, J.P. LEMAIGRE, G. et al, 1984. A Four-day Subrenal Capsule Assay for Testing the Effectiveness of Anticancer Drugs Against Human Tumours. Cancer Res. 44 2660-67.
- 221. FIDLER, I.J. HART, I.R. 1981 Biological and Experimental Consequences of the Zonal Composition of Solid Tumours. <u>Cancer Res</u>. 41 3266-67.
- 222. EDELSTEIN, M.B., FIEBIG, H.H., SMINK, T. et al 1983. Comparison Between Macroscopic and Microscopic Evaluation of Tumour Responsiveness using the Subrenal Capsule Assay. <u>Eur.J.Cancer</u> <u>Clin. Oncol. 19 995-1009.</u>
- 223. AAMDAHL, S. FODSTAD, E., PIHL, A. 1984. Human Tumour Xenografts Transplanted under the Renal Capsule of Conventional Mice. Growth Rates and Immune Response. Int. J. Cancer 84 725-30.
- 224. ANDERSON, R.G. W. BROWN, M.S. GOLDSTEIN, J.L. 1981. Inefficient Internalisation of Receptor-Bound Low Density Lipoprotein In Human Carcinoma A-431 Cells. <u>J. Cell Biology</u> <u>88</u> 441-52.
- 225. GAL, D., SIMPSON, E.R. PORTER, J.C., et al 1982. Defective Internalisation of Low Density Lipoprotein in Epidermoid Cervical Cancer Cells. <u>J. Cell Biol</u>. <u>92</u> 597-603.
- 226. BETTERIDGE, D.J. KRONE, W. FORD, J.M. GALTON, D.J. 1979. Regulation of Sterol Synthesis in Leukaemic Blast Cells: a Defect Resembling Familial Hypercholesterolemia. <u>Eur. J. Clin. Invest</u>. 9 439-41.
- 227. JOE,C.O. RANKIN, P.W., BUSBEE, D.L. 1984. Human Lymphocytes Treated with r-7,t-8-dihydroxy-t-9,10-epoxy-7,8,9,10-tetrahydrobenzo(a)pyrene Require Low Density Lipoproteins for DNA Excision Repair. <u>Mutation Res</u>. <u>131</u> 37-43.

228. ACTON, E.M. TONG, G.L. MOSHER, C.W. et al 1984. Intensely

Potent Morpholinyl Anthracyclines. J. Med. Chem. 27 638-45.

- 229. ARCAMONE, F. 1985 Properties of Antitumour Anthracyclines and New Developments in their application: Cain Memorial Award Lecture. Cancer Res. 45 5995-99.
- 230. KERR, D.J., HYNDS, S.A., COOK, J. et al 1986. Lipophilic Anthracycline Analogues Can Partially Overcome Drug Penetration Barriers in Human Lung Tumour Spheroids. Presented at the American Association for Cancer Research Symposium, Los Angeles, USA, 1986